

Impedance Control and Stability of DC/DC Converter Systems

By:

Xin Zhang

A thesis submitted in partial fulfilment of the requirements for the degree of Doctor of Philosophy

The University of Sheffield Faculty of Engineering

Department of Automatic Control and Systems Engineering

August 2016

Abstract

Cascaded DC/DC converter systems (or 'cascaded systems') have instability problems; i.e., although the subsystems may work well individually, the whole system may be instable due to the impedance interaction among these subsystems. To solve this problem, a family of impedance-based stabilisation methods are proposed in this thesis.

First, parallel-virtual-impedance (PVI) and series-virtual-impedance (SVI) control strategies are proposed to stabilise cascaded systems via shaping the load input impedance. Theoretically, the PVI or SVI control strategy connect a virtual impedance in parallel or series with the input port of the load converter so that the magnitude or phase of the load input impedance can be modified within a very small frequency range. Therefore, with the PVI or SVI control strategy, the cascaded system can be stabilised with minimal compromise of the load performance. Then, based on the PVI and SVI control strategies, adaptive-PVI (APVI) and adaptive-SVI (ASVI) control strategies are proposed by introducing an adaptive mechanism to change the impedance. With the APVI or ASVI control strategy, the load converter can be stably connected to different source converters without changing its internal structure. It is also shown that the ASVI control strategy is better than the APVI control strategy and can make the cascaded system more stable. Moreover, a minimum-ripple-point-tracking (MRPT) controller is proposed and utilised to solve the potential problem of the ASVI control strategy. Finally, a source-side SVI (SSVI) control strategy and a VRLC damper are proposed to stabilise the cascaded system with better source performance or input filter performance, respectively.

All the proposed stabilisation and control methods are validated by extensive experimental results.

Acknowledgements

First of all, I want to express my sincere thanks to my tremendous mentor, Professor Qing-Chang Zhong, for affording me the precious opportunity to do my Ph.D. dissertation and Research Associate work at the Control and Power Systems (CAPS) group. He is an enthusiastic, energetic and patient supervisor, who stands at the forefront of both control and power electronics and sets an excellent example for me. He often shares his scientific advice on my research and every discussion with him has sparked unforeseen and fruitful directions to my research. In addition to research, he has also taught me how to overcome difficulties and be a better person in life. In my opinion, his help on my research and life are both priceless. Most importantly, I want to thank him for the understanding and trust he gave, and which I will never forget.

I also want to thank my second supervisors Dr Hua-Liang Wei, and Dr Jun Liu for sharing their ideas, and offering many helpful comments. I would also like to thank Professor Visakan Kadirkamanathan, who provided me with a lot of support. I want to thank my Ph.D. Examiners Professor Volker Pickert, and Dr Keliang Zhou for their constructive suggestions which significantly improved the quality of my Ph.D. dissertation.

I am grateful to my colleagues in the CAPS group. Their help, encouragement and friendship has made my life rich and colourful. I wish to give particular thanks to the following: Dr Wen-Long Min, Dr Yu Zeng, Dr Jing-Lin Zhou, Dr Yao Zhang, Dr George Konstantopoulos and Mr Shlomo Gadelovits.

Finally, I wish to send my affectionate thanks to my beloved parents, Su-E Ge and Kai-Pin Zhang, and to my dear wife, Xiao-Qin Ma. Their understanding, patience, encouragement and unconditional love allowed me to keep going and finish my Ph.D degree.

Contents

1	Intr	oductio	n		1
	1.1	Backg	round		1
		1.1.1	DC Dist	ributed Power Systems (DC DPS)	4
			1.1.1.1	Cascaded System	4
			1.1.1.2	Paralleled System	4
			1.1.1.3	Source Splitting	6
			1.1.1.4	Load Splitting	6
		1.1.2	The Inst	ability Problem of Cascaded DC/DC Converter Systems	6
			1.1.2.1	Reason for the Instability Problem	7
			1.1.2.2	Characteristics of Source Output Impedance and Load	
				Input Impedance	9
			1.1.2.3	Physical Explanation of the System Instability Problem .	10
			1.1.2.4	The Worst Case of Instability Problem in Cascaded Sys-	
				tems	12
	1.2	Literat	ure Revie	w	12
		1.2.1	Hardwar	e-Based Stabilisation Methods	14
		1.2.2	Software	e-Based Stabilisation Methods	15
	1.3	Motiva	ation and (Objective	16
	1.4	Outlin	e of the Tl	nesis	19
	1.5	Major	Contribut	ions	22
	1.6	List of	Publication	ons	24
2	Para	allel-Vir	tual-Imp	edance (PVI) Control for Load Converters	27
	2.1	PVI C	- ontrol Stra	ıtegy	28

		2.1.1	Basic Idea of Load Shaping Methods	28
		2.1.2	Expression of the Parallel-Virtual-Impedance $Z_{PVI}(s)$	29
		2.1.3	Derivation of the PVI Control Strategy	31
		2.1.4	Realisation of the PVI Control Strategy	33
	2.2	Design	Example	35
		2.2.1	System I — a Buck Converter with a <i>LC</i> Input Filter	35
		2.2.2	System II — a Boost Converter with a <i>LC</i> Input Filter	37
	2.3	Experi	mental Verification	39
		2.3.1	System I — a Buck Converter with a <i>LC</i> Input Filter	41
		2.3.2	System II — a Boost Converter with a <i>LC</i> Input Filter	42
	2.4	Simula	ation Verification	49
		2.4.1	Simulation Verification of System I with the Consideration of the	
			Load Input Capacitor	49
		2.4.2	Simulation Verification of System II with the Consideration of the	
			Load Input Capacitor	49
	2.5	Summ	ary	58
3	Seri	es-Virtu	al-Impedance (SVI) Control for Load Converters	59
	3.1	SVI C	ontrol Strategy	60
				<i>c</i> 0
		3.1.1	Basic Idea of Load Shaping Methods	60
		3.1.1 3.1.2	Basic Idea of Load Shaping Methods \ldots \ldots Expression of the Series-Virtual-Impedance $Z_{SVI}(s)$ \ldots \ldots	60 60
		3.1.13.1.23.1.3	Basic Idea of Load Shaping Methods \ldots \ldots Expression of the Series-Virtual-Impedance $Z_{SVI}(s)$ \ldots \ldots Derivation of the SVI Control Strategy \ldots \ldots	60 60 63
		 3.1.1 3.1.2 3.1.3 3.1.4 	Basic Idea of Load Shaping Methods \ldots \ldots Expression of the Series-Virtual-Impedance $Z_{SVI}(s)$ \ldots \ldots Derivation of the SVI Control Strategy \ldots \ldots Realisation of the SVI Control Strategy \ldots \ldots	60 60 63 66
	3.2	3.1.13.1.23.1.33.1.4Design	Basic Idea of Load Shaping Methods	60 60 63 66 67
	3.2	 3.1.1 3.1.2 3.1.3 3.1.4 Design 3.2.1 	Basic Idea of Load Shaping Methods $\dots \dots $	60 60 63 66 67 67
	3.2	 3.1.1 3.1.2 3.1.3 3.1.4 Design 3.2.1 3.2.2 	Basic Idea of Load Shaping Methods \dots Expression of the Series-Virtual-Impedance $Z_{SVI}(s)$ \dots Derivation of the SVI Control Strategy \dots Realisation of the SVI Control Strategy \dots N Example \dots System I — a Buck Converter with a <i>LC</i> Input Filter \dots System II — a Boost Converter with a <i>LC</i> Input Filter \dots	 60 60 63 66 67 67 70
	3.2 3.3	 3.1.1 3.1.2 3.1.3 3.1.4 Design 3.2.1 3.2.2 Experi 	Basic Idea of Load Shaping Methods \dots Expression of the Series-Virtual-Impedance $Z_{SVI}(s)$ \dots Derivation of the SVI Control Strategy \dots Realisation of the SVI Control Strategy \dots N Example \dots System I — a Buck Converter with a <i>LC</i> Input Filter \dots System II — a Boost Converter with a <i>LC</i> Input Filter \dots mental Verification \dots	 60 60 63 66 67 67 67 70 72
	3.23.3	 3.1.1 3.1.2 3.1.3 3.1.4 Design 3.2.1 3.2.2 Experi 3.3.1 	Basic Idea of Load Shaping Methods	 60 60 63 66 67 67 70 72 72 72
	3.2 3.3	 3.1.1 3.1.2 3.1.3 3.1.4 Design 3.2.1 3.2.2 Experi 3.3.1 3.3.2 	Basic Idea of Load Shaping Methods	60 63 66 67 67 70 72 72 72 73
	3.23.33.4	 3.1.1 3.1.2 3.1.3 3.1.4 Design 3.2.1 3.2.2 Experi 3.3.1 3.3.2 Simula 	Basic Idea of Load Shaping Methods	60 63 66 67 67 70 72 72 72 73 77
	3.23.33.4	 3.1.1 3.1.2 3.1.3 3.1.4 Design 3.2.1 3.2.2 Experi 3.3.1 3.3.2 Simula 3.4.1 	Basic Idea of Load Shaping Methods	60 63 66 67 67 70 72 72 73 77

		3.4.2	Simulation Verification of System II with the Consideration of the	
			Load Input Capacitor	31
	3.5	Summ	ary	;7
4	Ada	ptive-P	VI (APVI) Control for Load Converters 8	;9
	4.1	Stabili	ty-preferred Input Impedance of the Load Converter with the APVI	
		Contro	ol Strategy	0
	4.2	APVI	Control Strategy	2
		4.2.1	Adaptive-Parallel-Virtual-Impedance Z_{APVI} and its Adaptive Ap-	
			proach	2
		4.2.2	The APVI Control Strategy	94
		4.2.3	Realisation of the APVI Control Strategy	95
		4.2.4	Operation Flow of the APVI Control Strategy	18
	4.3	Experi	mental Verification	9
	4.4	Summ	ary)4
5	Ada	ptive-S	VI (ASVI) Control for Load Converters 10)5
5	Ada 5.1	ptive-S Stabili	VI (ASVI) Control for Load Converters10sation via Amplitude Compensation (SAC) and Phase Compensation)5
5	Ada 5.1	ptive-S Stabili (SPC)	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10)5)6
5	Ada 5.1 5.2	ptive-S Stabili (SPC) Compa	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10)5)6 17
5	Ada 5.1 5.2	ptive-S Stabili (SPC) Compa 5.2.1	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10)5)6)7)8
5	Ada 5.1 5.2	ptive-S Stabili (SPC) Compa 5.2.1 5.2.2	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10)5)6)7)8
5	Ada 5.1 5.2 5.3	ptive-S Stabili (SPC) Comp 5.2.1 5.2.2 The A	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10)6)7)8)9
5	Ada 5.1 5.2 5.3	ptive-S Stabili (SPC) Comp 5.2.1 5.2.2 The A 5.3.1	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10)5)6)7)8)9 .1
5	Ada 5.1 5.2 5.3	ptive-S Stabili (SPC) Compa 5.2.1 5.2.2 The A 5.3.1 5.3.2	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10)6)7)8)9 .1 .1 .3
5	Ada 5.1 5.2 5.3	ptive-S Stabili (SPC) Compa 5.2.1 5.2.2 The A 5.3.1 5.3.2 5.3.3	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10)6)7)8)9 .1 .1 .3 .5
5	Ada 5.1 5.2 5.3	ptive-S Stabili (SPC) Compa 5.2.1 5.2.2 The A 5.3.1 5.3.2 5.3.3 5.3.4	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10 arison of PVI and SVI Control Strategies 10 Limitation of PVI Control Strategy When Realising the SAC 10 Advantage of SVI Control Strategy When Realising the SAC 10 SVI Control Strategy 11 Adaptive-Series-Virtual-Impedance Z _{ASVI} and its Adaptive Approach 11 11 Realisation of the ASVI Control Strategy 11 Impact of the ASVI Control Strategy on the Load Converter 11)6)7)8)9 .1 .1 .3 .5 .6
5	Ada 5.1 5.2 5.3	ptive-S Stabili (SPC) Compa 5.2.1 5.2.2 The A 5.3.1 5.3.2 5.3.3 5.3.4 Experi	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10)6)7)8)9 .1 .3 .5 .6 .7
5	Ada 5.1 5.2 5.3 5.4 5.5	ptive-S Stabili (SPC) Compa 5.2.1 5.2.2 The A 5.3.1 5.3.2 5.3.3 5.3.4 Experi Summ	VI (ASVI) Control for Load Converters 10 sation via Amplitude Compensation (SAC) and Phase Compensation 10 arison of PVI and SVI Control Strategies 10 Limitation of PVI Control Strategy When Realising the SAC 10 Advantage of SVI Control Strategy When Realising the SAC 10 SVI Control Strategy 11 Adaptive-Series-Virtual-Impedance Z _{ASVI} and its Adaptive Approach 11 11 Realisation of the ASVI Control Strategy on the Load Converter 11 Impact of the ASVI Control Strategy on the Load Converter 11 ary 11)6)7)8)9 .1 .1 .3 .5 .6 .7

6	Improved ASVI Control	with Minimum	Ripple Point	Tracking for	the Load	
	Converter				1	29

	6.1	The A	SVI Control Strategy and the Motivation of the MRPT Controller .	. 130
		6.1.1	Review of the ASVI Control Strategy	. 130
		6.1.2	Potential Problem of the ASVI Control Strategy	. 134
		6.1.3	The Motivation of the MRPT Controller	. 136
	6.2	The In	nproved ASVI Control Strategy with the MRPT Controller	. 137
		6.2.1	The Relationship Between the Centre Frequency of $Z_{ASVI}(s)$ and	
			$ \triangle v_{bus} $ — the Inspiration of the MRPT Controller $\ldots \ldots \ldots$. 137
		6.2.2	Basic Idea of the MRPT Controller	. 138
		6.2.3	Improvement of the MRPT Controller with Sinusoidal-Tracking-	
			Algorithm (STA)	. 141
		6.2.4	Realisation of the MRPT Controller	. 143
	6.3	Experi	mental Verification	. 144
	6.4	Summ	ary	. 148
7	Sou	rce-side	SVI (SSVI) Control to Stabilise Cascaded Systems and Impro	ve
	the	Source 1	Performance	149
	7.1	Impact	t of the ASVI Control Strategy on the Load Converter via Two-Port	
		Netwo	rk Analysis	. 150
		7.1.1	Two-Port Network Model of the Original Load Converter	. 151
		7.1.2	Two-Port Network Model of the Load Converter With the ASVI	
			Control Strategy	. 153
		7.1.3	Impact of the ASVI Control Strategy on the Load Converter	. 154
	7.2	The SS	SVI Control Strategy	. 155
		7.2.1	The Concept of the SSVI Control Strategy	. 155
		7.2.2	Realisation of the SSVI Control Strategy	. 157
		7.2.3	Impact of the SSVI Control Strategy on the Source Converter	. 159
	7.3	Experi	mental Results	. 162
	7.4	Summ	ary	. 163
8	Virt	tual RLC	C Damper for the <i>LC</i> Input Filter to Stabilise Cascaded Systems a	nd
	Imp	orove the	e Filter Performance	169
	8.1	Existir	ng Dampers and Their Impact on the LC Input Filter	. 171

	Cha	racteris	stic	209
A	Fun	dament	tal Concepts of Constant Power Load (CPL) — Negative Resisto	r
	9.2	Future	Work	. 199
	9.1	Conclu	usions	. 197
9	Con	clusion	s and Future Work	197
	8.5	Summ	ary	. 194
	8.4	Experi	imental Verification	. 187
		8.3.3	The Impact of the <i>VRLC</i> Damper on the CPL	. 187
		8.3.2	Effectiveness Evaluation of the VRLC Damper	. 184
		8.3.1	Concept and Implementation of the VRLC Damper	. 181
	8.3	Virtual	l Implementation of the Proposed RLC Damper	. 181
		8.2.3	The Impact of the Proposed <i>RLC</i> damper on the <i>LC</i> Input Filter .	. 180
			Robustness against the Variations of L_f and C_f	. 178
		8.2.2	Parameters Selection of the Proposed RLC Damper to Achieve High	
		8.2.1	Design Principle of the Proposed <i>RLC</i> Damper	. 177
	8.2	The Pr	roposed <i>RLC</i> Damper	. 177
			Network Analysis	. 174
		8.1.3	Impact of Existing Dampers on the <i>LC</i> Input Filter via Two-port	
			Dampers	. 173
		8.1.2	Two-Port Network Model of the <i>LC</i> Input Filter with Different	• 1/2
			8.1.1.3 <i>RL</i> Series Damper	. 172
			8.1.1.2 <i>RL</i> Parallel Damper	. 172
		01111	8.1.1.1 <i>RC</i> Parallel Damper	. 171
		8.1.1	Review of the Existing Dampers	. 171

List of Figures

1.1	Two types of 'power supply' architectures: (a) CPS. (b) DPS (Luo, 2005).	2
1.2	A typical DC DPS	3
1.3	A typical AC DPS	3
1.4	Basic structures of DC DPS: (a) cascading; (b) paralleling; (c) source split-	
	ting; (d) load splitting.	5
1.5	A typical cascaded DC/DC converter system.	7
1.6	Two-port network model of the cascaded system	7
1.7	Bode plots and Nyquist plot of the typical unstable cascaded system: (a)	
	Bode plots; (b) Nyquist plot	8
1.8	The equivalent circuit model of the cascaded system — A CPL cascaded	
	with a <i>LC</i> input filter	10
1.9	Typical dampers and their application: (a) RC parallel damper; (b) RL	
	parallel damper; (c) <i>RL</i> series damper; (d) Application	13
1.10	Stabilise the cascaded system by adding bus capacitor or adaptive active	
	capacitor converter: (a) adding bus capacitor; (b) adding adaptive active	
	capactior converter.	13
1.11	Software-based stabilisation methods for source converters: (a) adding	
	virtual resistor in series with the filter inductor; (b) adding virtual resistor	
	in series with the source output; (c) adding virtual resistor in parallel with	
	the source output.	17
1.12	Software-based stabilisation methods for load converters: (a) adding vir-	
	tual resistor or capacitor in parallel with the load input; (b) sending the	
	regulation signal from source converter to load converter	17

2.1	Basic idea of load shaping methods: (a) only increasing $ Z_{iL} $ at $[f_1, f_2]$ to	
	make a total separation between $ Z_{oS} $ and $ Z_{iL} $; (b) only changing $\varphi(Z_{iL})$	
	at $[f_1, f_2]$ to ensure $ \varphi(Z_{oS}) - \varphi(Z_{iL}) < 180^0$ at $[f_1, f_2]$	28
2.2	A cascaded system with parallel virtual impedance	29
2.3	Small signal block diagram of the original load converter	31
2.4	Small signal block diagram of the load converter with Z_{PVI}	32
2.5	The PVI control strategy.	33
2.6	Characteristic of $ G_{PVI}(s) $	34
2.7	The System I	35
2.8	Bode plots of Z_{oS} and Z_{iL} of the System I	36
2.9	Software flowchart of the PVI control strategy.	37
2.10	The System II	38
2.11	Bode plots of Z_{oS} and Z_{iL} of the System II	39
2.12	Experimental waveforms of System I at full load: (a) without PVI control	
	strategy, (b) with PVI control strategy.	40
2.13	Experimental waveforms of System I at half load: (a) without PVI control	
	strategy, (b) with PVI control strategy.	40
2.14	Experimental waveforms of System I at light load: (a) without PVI control	
	strategy, (b) with PVI control strategy.	41
2.15	Simulation waveforms of System I at full load: (a) without PVI control	
	strategy, (b) with PVI control strategy.	43
2.16	Harmonic distribution histogram of System I without PVI control strategy	
	at full load: (a) i_o , (b) v_{bus} , (c) v_o	44
2.17	Harmonic distribution histogram of System I with PVI control strategy at	
	full load: (a) i_o , (b) v_{bus} , (c) v_o	45
2.18	Dynamic experimental waveforms of System I with PVI control strategy:	
	(a) load changing between 10% load and 100% load, (b) input voltage	
	changing between 80% rated voltage and 120% rated voltage	46
2.19	Experimental waveforms of System II at full load: (a) without PVI control	
	strategy, (b) with PVI control strategy.	46

2.20	Experimental waveforms of System II at half load: (a) without PVI control	
	strategy, (b) with PVI control strategy.	47
2.21	Experimental waveforms of System II at light load: (a) without PVI control	
	strategy, (b) with PVI control strategy.	48
2.22	Dynamic experimental waveforms of System II with PVI control strategy:	
	(a) load changing between 10% load and 100% load, (b) input voltage	
	changing between 80% rated voltage and 120% rated voltage	48
2.23	Simulation waveforms of System I with the added load input capacitor at	
	full load: (a) without PVI control strategy, (b) with PVI control strategy	50
2.24	Simulation waveforms of System I with the added load input capacitor at	
	half load: (a) without PVI control strategy, (b) with PVI control strategy	51
2.25	Simulation waveforms of System I with the added load input capacitor at	
	light load: (a) without PVI control strategy, (b) with PVI control strategy.	52
2.26	Dynamic simulation waveforms of System I with PVI control strategy and	
	the added load input capacitor: (a) load changing between 10% load and	
	100% load, (b) input voltage changing between 80% rated voltage and	
	120% rated voltage	53
2.27	Simulation waveforms of System II with the added load input capacitor at	
	full load: (a) without PVI control strategy, (b) with PVI control strategy	54
2.28	Simulation waveforms of System II with the added load input capacitor at	
	half load: (a) without PVI control strategy, (b) with PVI control strategy	55
2.29	Simulation waveforms of System II with the added load input capacitor at	
	light load: (a) without PVI control strategy, (b) with PVI control strategy.	56
2.30	Dynamic simulation waveforms of System II with PVI control strategy	
	and the added load input capacitor: (a) load changing between 10% load	
	and 100% load, (b) input voltage changing between 80% rated voltage and	
	120% rated voltage	57
3.1	Basic idea of load shaping methods: (a) only increasing $ Z_{iL} $ at $[f_1, f_2]$ to	
	make a total separation between $ Z_{oS} $ and $ Z_{iL} $; (b) only increasing $\varphi(Z_{iL})$	
		61
	at $[f_1, f_2]$ to ensure $ \varphi(Z_{oS}) - \varphi(Z_{iL}) < 180^{\circ}$ at $[f_1, f_2]$	01
3.2	at $[f_1, f_2]$ to ensure $ \varphi(Z_{oS}) - \varphi(Z_{iL}) < 180^\circ$ at $[f_1, f_2]$	61

ix

3.3	Small signal block diagram of the original load converter	64
3.4	Small signal block diagram of the load converter with the paralleled virtual	
	impedance.	64
3.5	Derivation of the SVI control strategy.	64
3.6	Characteristic of $ G_{SVI}(s) $	67
3.7	The System I	67
3.8	Bode plots of Z_{oS} and Z_{iL} of the System I	68
3.9	Software flowchart of the SVI control strategy.	69
3.10	The System II	70
3.11	Bode plots of Z_{oS} and Z_{iL} of the System II	71
3.12	Experimental waveforms of the System I at full load: (a) without SVI con-	
	trol strategy, (b) with SVI control strategy.	72
3.13	Experimental waveforms of the System I at half load: (a) without SVI	
	control strategy, (b) with SVI control strategy	73
3.14	Experimental waveforms of the System I at light load: (a) without SVI	
	control strategy, (b) with SVI control strategy	74
3.15	Dynamic experimental waveforms of the System I with SVI control strat-	
	egy: (a) load changing between 10% load and 100% load, (b) input voltage	
	changing between 80% rated voltage and 120% rated voltage	74
3.16	Experimental waveforms of the System II at full load: (a) without SVI	
	control strategy, (b) with SVI control strategy.	75
3.17	Experimental waveforms of the System II at half load: (a) without SVI	
	control strategy, (b) with SVI control strategy.	75
3.18	Experimental waveforms of the System II at light load: (a) without SVI	
	control strategy, (b) with SVI control strategy.	76
3.19	Dynamic experimental waveforms of the System II with SVI control strat-	
	egy: (a) load changing between 10% load and 100% load, (b) input voltage	
	changing between 80% rated voltage and 120% rated voltage	77
3.20	Simulation waveforms of the System I with the added load input capacitor	
	at full load: (a) without SVI control strategy, (b) with SVI control strategy	78

3.21	Simulation waveforms of the System I with the added load input capacitor	
	at half load: (a) without SVI control strategy, (b) with SVI control strategy.	79
3.22	Simulation waveforms of the System I with the added load input capacitor	
	at light load: (a) without SVI control strategy, (b) with SVI control strategy.	80
3.23	Dynamic simulation waveforms of the System I with SVI control strategy	
	and the added load input capacitor: (a) load changing between 10% load	
	and 100% load, (b) input voltage changing between 80% rated voltage and	
	120% rated voltage	82
3.24	Simulation waveforms of the System II with the added load input capacitor	
	at full load: (a) without SVI control strategy, (b) with SVI control strategy	83
3.25	Simulation waveforms of the System II with the added load input capacitor	
	at half load: (a) without SVI control strategy, (b) with SVI control strategy.	84
3.26	Simulation waveforms of the System II with the added load input capacitor	
	at light load: (a) without SVI control strategy, (b) with SVI control strategy.	85
3.27	Dynamic simulation waveforms of the System II with SVI control strategy	
	and the added load input capacitor: (a) load changing between 10% load	
	and 100% load, (b) input voltage changing between 80% rated voltage and	
	120% rated voltage	86
4.1	Instability reason and stability-preferred input impedance of the load con-	
	verter with the APVI control strategy: (a) instability reason; (b) stability-	
	preferred input impedance of the load converter	91
4.2	A cascaded system with adaptive parallel-virtual-impedance	91
4.3	Adaptively parallel-virtual-impedance $Z_{APVI}(f_{CS})$	93
4.4	Basic idea of the adaptive approach of $Z_{APVI}(f_{CS})$	93
4.5	Concept of the APVI control strategy.	94
4.6	Adaptive characteristic of $ G_{APVI}(s) $	95
4.7	Realisation of the $G_{APVI}(s)$: (a) control circuit of $G_{APVI}(s)$; (b) adaptive	
	mechanism of $G_{APVI}(s)$	96
4.8	Impact of K_p and K_i on $G_{APVI}(s)$: (a) large K_p and K_i ; (b) small K_p and K_i .	98
4.9	Operation flow of the APVI control method	99
4.10	Example cascaded systems	.00

 4.12 Experimental waveforms of the first cascaded system with the APVI control strategy: (a) steady state waveforms with rated input voltage and rated load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated to 100% rated load at rated input voltage. 1 4.13 Experimental waveforms of the second cascaded system with the APVI control strategy: (a) steady state waveforms with rated input voltage and 	102
 trol strategy: (a) steady state waveforms with rated input voltage and rated load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated to 100% rated load at rated input voltage. 4.13 Experimental waveforms of the second cascaded system with the APVI control strategy: (a) steady state waveforms with rated input voltage and 	102
 load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated to 100% rated load at rated input voltage. If 4.13 Experimental waveforms of the second cascaded system with the APVI control strategy: (a) steady state waveforms with rated input voltage and 	102
 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated to 100% rated load at rated input voltage. 1 4.13 Experimental waveforms of the second cascaded system with the APVI control strategy: (a) steady state waveforms with rated input voltage and 	102
the load increases from 10% rated to 100% rated load at rated input voltage. 14.13 Experimental waveforms of the second cascaded system with the APVI control strategy: (a) steady state waveforms with rated input voltage and	102
4.13 Experimental waveforms of the second cascaded system with the APVI control strategy: (a) steady state waveforms with rated input voltage and	
control strategy: (a) steady state waveforms with rated input voltage and	
rated load, (b) dynamic waveforms when the input voltage steps down from	
100% rated to 80% rated voltage at full load, (c) dynamic waveforms when	
the load increases from 10% rated to 100% rated load at rated input voltage. 1	103
5.1 Bode plots of SAC and SPC: (a) SAC: (b) SPC	106
5.2 Nyquist plots of Z_{ns}/Z_{it} with SAC and SPC: (a) with SAC: (b) with SPC.	107
5.3 The PVI control strategy and its physical concept.	109
5.4 Requirement of Z_{PVI} during $[f_1, f_2]$ when realising the SAC	109
5.5 The SVI control strategy and its physical concept.	10
5.6 Requirement of $ Z_{SVI} $ during $[f_1, f_2]$ when realising the SAC.	10
5.7 Characteristic of $ Z_{ASVI}(f_{CS}) $	12
5.8 Basic idea of the adaptive approach of $Z_{ASVI}(f_{CS})$	112
5.9 The ASVI control strategy	113
5.10 Control system of the load converter with the ASVI control strategy 1	14
5.11 Realisation of $G_{ASVI}(s)$	115
5.12 Bode plots of the load converter with/without the ASVI control strategy:	
(a) $Z_{iL}(s)$; (b) $G_{iL}(s)$; (c) $G_{vL}(s)$; (d) $Z_{oL}(s)$	16
5.13 The experimental system	18
5.14 Bode plots in Case I of the experimental system with different working	
conditions: (a) without the ASVI control strategy, v_{bus} : 38.4V \rightarrow 57.6V,	
$p_o = 100W$; (b) with the ASVI control strategy, $v_{bus} : 38.4V \rightarrow 57.6V$,	
$p_o = 100W$; (c) without the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow$	
100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, $p_o: 10W \rightarrow 100W$.	110

5.15 Experimental waveforms in Case I of the cascaded system: (a) steady state waveforms with rated input voltage and rated load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated 5.16 Simulation waveforms in Case I of the cascaded system with the wrong 5.17 Bode plots in Case II of the experimental system with different working conditions: (a) without the ASVI control strategy, v_{bus} : $38.4V \rightarrow 57.6V$, $p_o = 100W$; (b) with the ASVI control strategy, v_{bus} : 38.4V \rightarrow 57.6V, $p_o = 100W$; (c) without the ASVI control strategy, $v_{bus} = 48V$, $p_o: 10W \rightarrow$ 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, $p_o: 10W \rightarrow 100W$. 122 5.18 Experimental waveforms in Case II of the cascaded system: (a) steady state waveforms with rated input voltage and rated load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated 5.19 Simulation waveforms in Case II of the cascaded system with the wrong 5.20 Bode plots in Case III of the experimental system with different working conditions: (a) without the ASVI control strategy, v_{bus} : 38.4V \rightarrow 57.6V, $p_o = 100W$; (b) with the ASVI control strategy, v_{bus} : 38.4V \rightarrow 57.6V, $p_o = 100W$; (c) without the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow$ 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, $p_o: 10W \rightarrow 100W$. 125 5.21 Experimental waveforms in Case III of the cascaded system: (a) steady state waveforms with rated input voltage and rated load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 5.22 Simulation waveforms in Case III of the cascaded system with the wrong

6.1	Purposes and concept of the ASVI control strategy: (a) purposes of the
	ASVI control strategy — described by Bode plots; (b) purposes of the
	ASVI control strategy — described by Nyquist curve; (c) concept of the
	ASVI control strategy
6.2	Realisation of the ASVI control strategy via adding $G_{ASVI}(s)$
6.3	Realisation of the $G_{ASVI}(s)$: (a) control circuit of $G_{ASVI}(s)$; (c) adaptive
	mechanism of $G_{ASVI}(s)$
6.4	The potential problem and its existing prevention method of the ASVI
	control strategy: (a) miss f_{CS} with excessive PI control parameters; (b)
	failure mechanism; (c) existing prevention method in Chapter 5 — using
	small K_p and K_i
6.5	Experimental system and its failure Cases of the ASVI control strategy
	with the same PI controller in Chapter 5: (a) experimental system; (b) Case
	I; (c) Case II
6.6	Impact of f_o change on the ASVI control strategy: (a) Bode plots of Z_{iL} ,
	Z_{iLS} and Z_{oS} at Case I; (b) Bode plots of Z_{iL} , Z_{iLS} and Z_{oS} at Case II; (c) the
	relationship between f_o and $ \triangle v_{bus} \dots \dots$
6.7	Basic idea and algorithm flowchart of the MRPT controller: (a) basic idea;
	(b) algorithm flowchart
6.8	Improvement of the MRPT controller: (a) slow processing speed of the
	MRPT controller; (b) improvement of the MRPT controller; (c) control
	block of STA
6.9	Implementation circuit of $G_{ASVI}(s)$ with the MRPT controller
6.10	Experimental waveforms at Cases I & II with the improved ASVI control
	strategy incorporating the basic MRPT controller: (a) Case I; (b) Case II. $\ . 145$
6.11	Experimental waveforms at Cases I with the improved ASVI control strat-
	egy incorporating the improved MRPT controller: (a) steady state wave-
	forms with rated input voltage and load, (b) dynamic waveforms when the
	input voltage steps down from 100% to 80% rated voltage at full load, (c)
	dynamic waveforms when the load increases from 10% to 100% rated load
	at rated input voltage

6.12	Experimental waveforms at Cases II with the improved ASVI control strat-
	egy incorporating the improved MRPT controller: (a) steady state wave-
	forms with rated input voltage and load, (b) dynamic waveforms when the
	input voltage steps down from 100% to 80% rated voltage at full load, (c)
	dynamic waveforms when the load increases from 10% to 100% rated load
	at rated input voltage
7.1	Distributed power system with multiple upstream converters and one down-
	stream converter
7.2	Two-port network model of the original load converter: (a) block diagram;
	(b) two-port network model
7.3	Two port network model of the load converter with the ASVI control strat-
	egy: (a) block diagram; (b) two-port network model
7.4	Example system with the ASVI control strategy
7.5	Bode plots of the load converter with or without the ASVI control strategy:
	(a) $Z_{iL}(s)$ and $Z_{iLS}(s)$; (b) $G_{iL}(s)$ and $G_{iLS}(s)$; (c) $G_{\nu L}(s)$ and $G_{\nu LS}(s)$; (d)
	$Z_{oL}(s)$ and $Z_{oLS}(s)$
7.6	Concept of the SSVI control strategy
7.7	The SSVI control strategy: (a) control block; (b) algorithm
7.8	Two-port network model of the source converter with the SSVI control
	strategy
7.9	Example system with SSVI control strategy
7.10	Bode plots of the source converter without or with the SSVI control strat-
	egy: (a) $Z_{iS}(s)$ and $Z_{iSS}(s)$; (b) $G_{iS}(s)$ and $G_{iSS}(s)$; (c) $G_{vS}(s)$ and $G_{vSS}(s)$;
	(d) $Z_{oS}(s)$ and $Z_{oSS}(s)$
7.11	Experimental system
7.12	Experimental waveforms of the cascaded system with the SSVI control
	strategy: (a) steady state waveforms when $V_{in} = 48V$, $P_o = 100W$, (b) dy-
	namic waveforms when V_{in} steps up from 38.4 V to 48 V, (c) dynamic
	waveforms when V_{in} steps down from 48 V to 38.4 V, (e) dynamic wave-
	forms when P_o increases from 10 W to 100 W, (f) dynamic waveforms
	when P_o decreases from 100 W to 10 W

7.13	Dynamic waveforms of the original source converter without the SSVI con-
	trol strategy: (a) V_{in} steps up from 38.4 V to 48 V, (b) V_{in} steps down from
	48 V to 38.4 V, (c) P_o increases from 10 W to 100 W, (d) P_o decreases from
	100 W to 10 W
7.14	Dynamic waveforms of the source converter with the SSVI control strategy:
	(a) V_{in} steps up from 38.4 V to 48 V, (b) V_{in} steps down from 48 V to 38.4
	V, (c) P_o increases from 10 W to 100 W, (d) P_o decreases from 100 W to 10
	W
8.1	Typical dampers and their application: (a) RC parallel damper; (b) RL
	parallel damper; (c) <i>RL</i> series damper; (d) application
8.2	Generic two-port network model of the LC input filter with different dampers:
	(a) circuit model; (b) block diagram
8.3	Impact of different dampers on the <i>LC</i> input filter: (a) $Z_{ox}(s)$; (b) $1/Z_{ix}(s)$;
	(c) $G_{\nu x}(s) \& G_{ix}(s)$
8.4	The RLC damper: (a) structure; (b) characteristics; (c) design principle 178
8.5	The relationship between f_L , f_H and L_f , C_f : (a) $\frac{f_L}{f_{LR}}$ vs $\frac{L_f}{L_{fR}}$; (b) $\frac{f_L}{f_{LR}}$ vs $\frac{C_f}{C_{fR}}$;
	(c) $\frac{f_H}{f_{HR}}$ vs $\frac{L_f}{L_{fR}}$; (d) $\frac{f_H}{f_{HR}}$ vs $\frac{C_f}{C_{fR}}$
8.6	Bode plots of $Z_{o4}(s)$ when L_f or C_f changes: (a) $L_f : 0.9mH \rightarrow 1.1mH$,
	$C_f = 50\mu F$; (b) $L_f = 1mH, C_f : 45\mu F \to 55\mu F.$
8.7	Control strategy of the VRLC damper: (a) concept; (b) digital implementa-
	tion; (c) different discretization effect of $G_{RLC}(s)$
8.8	Comparison of the <i>RLC</i> and <i>VRLC</i> dampers: (a) function comparison; (b)
	Bode plots of $Z_{RLC}(s)$ and $Z_{VRLC}(s)$
8.9	Impact of <i>RLC</i> and <i>VRLC</i> dampers on the <i>LC</i> input filter: (a) $Z_{ox}(s)$; (b)
	$1/Z_{ix}(s)$; (c) $G_{vx}(s)$ & $G_{ix}(s)$
8.10	Impact of the <i>VRLC</i> damper on the CPL: (a) $Z_{iL}(s)$; (b) $Z_{oL}(s)$; (c) $G_{iL}(s)$;
	(d) $G_{\nu L}(s)$
8.11	The experimental system: (a) main circuit; (b) steady state experimental
	waveforms at rated conditions without any dampers

8.12	Experimental waveforms with different dampers when system input volt-	
	age changed from 80% to 100% rated voltage at full load: (a) with VRLC	
	damper; (b) with <i>RC</i> parallel damper; (c) with <i>RL</i> parallel damper; (d) with	
	<i>RL</i> series damper	. 190
8.13	Comparison of $\sigma_{I_{in}}$, $t_{I_{in}}$, $\sigma_{V_{bus}}$ and $t_{V_{bus}}$ with different dampers when system	
	input voltage changed from 80% to 100% rated voltage at full load: (a) $\sigma_{I_{in}}$;	
	(b) $t_{I_{in}}$; (c) $\sigma_{V_{bus}}$; (d) $t_{V_{bus}}$.	. 191
8.14	Experimental waveforms with different dampers when system output power	
	changed from 10 W to 100 W at rated input voltage: (a) with VRLC damper;	
	(b) with RC parallel damper; (c) with RL parallel damper; (d) with RL series	
	damper	. 192
8.15	Comparison of $\sigma_{I_{in}}$, $t_{I_{in}}$, $\sigma_{V_{bus}}$ and $t_{V_{bus}}$ with different dampers when system	
	output power changed from 10 W to 100 W at rated input voltage: (a) $\sigma_{I_{in}}$;	
	(b) $t_{I_{in}}$; (c) $\sigma_{V_{bus}}$; (d) $t_{V_{bus}}$.	. 193
8.16	System efficiency curves with different dampers: (a) efficiency vs p_o at	
	rated v_{in} ; (b) efficiency vs v_{in} at rated p_0	. 194

List of Tables

2.1	Definitions of variables and transfer functions in load converter
2.2	Parameters of the System I
2.3	Parameters of the System II
3.1	Definitions of variables and transfer functions in load converter
3.2	Parameters of the System I
3.3	Parameters of the System II
4.1	Variables and transfer functions of load converter
5.1	Variables and transfer functions of load converter
7.1	Parameters of the example system with the ASVI control strategy 154
7.2	Comparison for ASVI and SSVI control strategies
8.1	Definitions of $A_x(s)$ and $B_x(s)$ in different dampers
8.2	Parameters of the example LC input filter with different dampers
8.3	Impact on LC input filter with different dampers
8.4	Variables and transfer functions of the CPL

Nomenclature

- AACC Adaptive active capacitor converter
- APVI Adaptive parallel virtual impedance
- ASVI Adaptive series virtual impedance
- CCM Continuous current mode
- CPL Constant power load
- CPS Centralized power system
- DCM Dis-continuous current mode
- DPS Distributed power system
- DSP Digital signal processor
- EMI Electro magnetic interference
- FPGA Field programmable gate array
- GaN Gallium Nitride
- GM Gain margin
- MCU Micro-controller unit
- MPPT Maximum power point tracking
- MRP Minimum ripple point
- MRPT Minimum Ripple point tracking

- PI Proportional integral
- PLL Phase-locked-loop
- PM Phase margin
- PO Perturb and observe
- PV Photovoltaic
- PVI Parallel virtual impedance
- PWM Pulse width modulation
- Q Quality factor
- SAC Stabilized via amplitude compensation
- SiC Silicon Carbide
- SPC Stabilized via phase compensation
- SSVI Source-side series virtual impedance
- STA Sinusoidal tracking algorithum
- SVI Series virtual impedance
- THD Total harmonic distortion
- UPS Uninterrupted power supply
- VRLC Virtual RLC
- ZOH Zero-order hold

Chapter 1

Introduction

1.1 Background

Nowadays, there are two types of 'power supply' architectures (Luo, 2005), centralised power system (CPS) and distributed power system (DPS). Fig. 1.1 illustrates the structures of CPS and DPS.

Fig. 1.1(a) shows the two typical features of CPS (Luo, 2001): a) there is only one power conversion stage in CPS, and b) there are many output voltages generated by CPS. The advantages of CPS include its concentrating of all power processing and thermal management technologies into a single converter that can be designed, subcontracted, or purchased as a stand-alone item (Bo, 2003). As a result, CPS is very popular in traditional electronic equipment, i.e., old computer or server power supply systems (Bodi, 1988). However, the CPS has two unavoidable disadvantages (Ye et al., 2002; Ren et al., 2005): a) It is not easy to achieve redundant design; b) Partial failure may destroy the whole system. These drawbacks mean that the CPS often fails to provide adequate performance for new generations of electronic equipment, especially for some large power applications such as space stations (Thomas and Hallinan, 1990; Lee et al., 1988), aircraft (Sarlioglu and Morris, 2015), new communication systems (Liu et al., 2007b), electrical vehicles (Emadi et al., 2006) and renewable energy systems (Arai et al., 2008).

DPS was developed in order to overcome the drawbacks of CPS (Thorsell, 1990; Schulz et al., 1990; Karlsson and Svensson, 2003). As shown in Fig. 1.1(b), DPS has the following characteristics (Luo, 2001): a) there are multiple power conversion stages in DPS; b)



Figure 1.1: Two types of 'power supply' architectures: (a) CPS. (b) DPS (Luo, 2005).

intermediate voltage bus is around the whole system; c) multiple upstream or downstream power converters are connected to the same voltage bus; d) each downstream power converter is located near its own load to provide the load voltage. It is worthy of mention that, in Fig. 1.1(b), the upstream conversion stage could be either a power converter or an electro-magnetic interference (EMI) filter. With these characteristics, the major disadvantages of CPS are entirely eliminated by DPS, and a consequence DPS is becoming increasingly popular (Boroyevich et al., 2010).

However, DPS also has two disadvantages: a) as DPS is formed by lots of power converters, its cost is higher than that of CPS; and b) the interaction between the power converters may cause instability problems for the entire DPS. The former shortage is unavoidable in practice, i.e., if the DPS is utilised, its relatively high cost should be accepted. However for the latter shortage, the instability problem can be solved by effective stabilisation methods. That is also the main research topic in this thesis.

DPS can be divided into DC DPS (Hua et al., 1994) and AC DPS (Shi et al., 2014), according to types of intermediate voltage bus. A typical DC DPS is presented in Fig.







Figure 1.3: A typical AC DPS.

1.2. As shown, it is composed of a number of power converters, such as DC/DC converter, DC/AC inverter and AC/DC rectifier; and all the power converters are connected to the same DC voltage bus. Fig. 1.3 shows a typical AC DPS. In contrast to the DC DPS, the above power converters in AC DPS are all connected to the same AC voltage bus, instead of the DC voltage bus. It is worth pointing out that both DC DPS and AC DPS are two different structures of DPS, and it is hard to say which is better or worse (Xu and Cheng, 2011). However, the DC DPS is the only one studied in this thesis.

1.1.1 DC Distributed Power Systems (DC DPS)

As shown in Fig. 1.4, most DC DPS can be derived from combinations of four basic configurations (Luo, 2001), which are cascading, paralleling, split source and split load. A description of these architectures and their characteristics is given below.

1.1.1.1 Cascaded System

As shown in Fig. 1.4(a), the cascaded system includes at least two power conversion stages, and these power conversion stages are all interconnected by DC voltage bus. There are three advantages to the cascaded system. First, due to the application of the DC voltage bus, each power converter can be placed near its corresponding load to improve the regulation precision and dynamic performance of the load voltage. Second, the DC bus voltage can appropriately be raised to decrease the power loss caused by the line impedance. Finally, multiple power conversion stages are suitable for the wide input voltage application.

1.1.1.2 Paralleled System

As shown in Fig. 1.4(b), the paralleled system shares a common source and load of its subconverters. It usually has three advantages. First, since the output power of each power converter is only one part of the system power, the thermal design of each power converter becomes easier. Second, although the paralleling structure increases the number of the power semiconductor devices and passive components, the electric and thermal stress of each power converter is greatly reduced. Finally, the redundant and modular design is very easy to be achieved by the paralleled system, which is also convenient for increasing the capacity of the system.



Figure 1.4: Basic structures of DC DPS: (a) cascading; (b) paralleling; (c) source splitting; (d) load splitting.

1.1.1.3 Source Splitting

Source splitting structure is shown in Fig. 1.4(c). It allows the use of separate power sources to supply a common load. This structure is very suitable for uninterrupted power supply (UPS), i.e., if the main power source is not working, backup energy can be connected to the DC voltage bus, supporting the system via the source splitting structure.

1.1.1.4 Load Splitting

As shown in Fig. 1.4(d), the load splitting structure is a configuration where separate power converters are utilised to feed different loads. This structure is suitable for a large-scale power supply system, such as a spacecraft (Lingjie et al., 2013), shipboard (McCoy and Amy, 2009) and DC micro-grid (Kwasinski and Onwuchekwa, 2006). These systems usually have many loads physically distributed over substantial distances. However, with the load splitting structure, separate power converters can be located in close proximity to each load to provide adequate regulation.

With the above four typical structures, DC DPS has become increasing popular, and its applications range from small, several hundred watt personal computers (Heath, 1991), through to kilo watt electronic vehicle systems (Kassakian, 2000), to mega watt micro-grid systems as well as other high-power applications found in smart grid systems (Lee et al., 2012).

1.1.2 The Instability Problem of Cascaded DC/DC Converter Systems

As discussed before, cascaded structure is a typical connection structure of DC DPS. Thanks to the advantages of the cascaded structure, the cascaded DC/DC converter system (also called as cascaded system) is becoming increasingly popular in various applications, such as space stations, shipboards, battery charging systems, and some other electrical devices comprising of two or more converters (Blaabjerg et al., 2005; Shamsi and Fahimi, 2013; Ovalle et al., 2015). A typical cascaded system is depicted in Fig. 1.5. Its upstream and downstream converters are called source and load converters, respectively. However, the closed-loop controlled load converter acts as a constant power load (CPL) and behaves as negative impedance in a small-signal analysis (Kwasinski and Onwuchekwa, 2011). As



Figure 1.5: A typical cascaded DC/DC converter system.



Figure 1.6: Two-port network model of the cascaded system.

a result, the stability of the cascaded system may be deteriorated by CPL, with the oscillating phenomenon then appearing in the cascaded system as shown in Fig. 1.5. This is the instability problem of the cascaded system (Zhang et al., 2013; Konig et al., 2014; Du et al., 2013). The fundamental concept of CPL and its negative resistor characteristics are explained in Appendix A.

1.1.2.1 Reason for the Instability Problem

In order to further reveal the instability reason of the cascaded system, the two-port network model of the cascaded system of Fig. 1.5 is depicted in Fig. 1.6. Here, v_{in} and i_{in} are the input voltage and current of the cascaded system, respectively. v_{bus} and i_{bus} are the bus voltage and current of the cascaded system, respectively, and v_o and i_o the output voltage and current of the cascaded system, respectively. In Fig. 1.6, $G_{vX}(s)$ is the closed loop input to output voltage transfer function of the source (X = S) or load converter (X = L); $Z_{iX}(s)$ is the closed loop input impedance of the source or load converter; $G_{iX}(s)$ is the closed loop load to input current transfer function of the source or load converter; and $Z_{oX}(s)$ is the closed loop output impedance of the source or load converter. According to Fig. 1.6, the four basic input-to-output transfer functions of the cascaded system can be derived as



Figure 1.7: Bode plots and Nyquist plot of the typical unstable cascaded system: (a) Bode plots; (b) Nyquist plot.

follows:

$$\frac{\hat{i}_{in}(s)}{\hat{v}_{in}(s)}\Big|_{\hat{i}_o=0} = \frac{G_{vS}(s)G_{iS}(s)/Z_{iL}(s)}{1+T_m(s)} + \frac{1}{Z_{iS}(s)}$$
(1.1)

$$\frac{\hat{v}_{o}(s)}{\hat{v}_{in}(s)}\Big|_{\hat{i}_{c}=0} = \frac{G_{vS}(s)G_{vL}(s)}{1+T_{m}(s)}$$
(1.2)

$$\frac{\hat{i}_{in}(s)}{\hat{i}_{o}(s)}\Big|_{\hat{v}_{m}=0} = \frac{G_{iS}(s)G_{iL}(s)}{1+T_{m}(s)}$$
(1.3)

$$\frac{\hat{v}_o(s)}{\hat{i}_o(s)}\Big|_{\hat{v}_{in}=0} = -Z_{oL}(s) - \frac{G_{iL}(s)G_{vL}(s)Z_{oS}(s)}{1+T_m(s)}$$
(1.4)

where

$$T_m(s) = \frac{Z_{oS}(s)}{Z_{iL}(s)}$$
(1.5)

It is known that, if the source and load converters are stable individually, there will be no right half plane (RHP) poles in the numerator expressions of (1.1) ~ (1.4). Therefore, at this moment, the stability of the cascaded system will be determined by $T_m(s)$, i.e., if $T_m(s)$ meets the Nyquist criteria, the cascaded system is stable, otherwise, the system is unstable. As a result, $T_m(s)$ is also called the minor loop gain of the cascaded system (Zhang et al., 2015).

As discussed before, the mismatch between $Z_{oS}(s)$ and $Z_{iL}(s)$ is the basic instability
reason of the cascaded system. Actually, this instability phenomenon can be predicted by Middlebrook criterion (Middlebrook, 1979; Turner et al., 2013; Feng et al., 2002), i.e., as shown the Bode plots in Fig. 1.7(a), if the amplitude of the source output impedance $|Z_{oS}|$ is intersected with the amplitude of the load input impedance $|Z_{iL}|$, and if the cut-off frequency f_{CS} of the source converter is lower than the cut-off frequency f_{CL} of the load converter, the cascaded system is unstable even if the converters can work well individually. In Fig. 1.7(b), the Nyquist plot of Fig. 1.7(a) is also depicted, as seen, the system loop gain $Z_{oS}(s)/Z_{iL}(s)$ indeed encircles (-1, j0), which further proves that the cascaded system is unstable in this case.

Therefore, even though the subsystems can work well individually, the cascaded system may suffer instability problem due to the interaction between $Z_{oS}(s)$ and $Z_{iL}(s)$. It should be stressed that the above instability problem refers to the slow scale instability problem, and not the fast scale instability problem. This is because the analysis of the above instability problem is based on state-space average model which ignores the information around the switching frequency of power converters. It is also worth pointing out that all the instability problems and analysis refer only to slow scale instability cases in this thesis.

1.1.2.2 Characteristics of Source Output Impedance and Load Input Impedance

As discussed above, the instability problem of the cascaded system is caused by the intersection of the source output impedance and load input impedance. Therefore, the characteristics of the source output impedance and the load input impedance are reviewed in this sub-section. Some important conclusions have been drawn from the existing literature and summarised here (S. Abe and Ninomiya, 2006; Hankaniemi et al., 2005; Middlebrook and Cuk, 1976; Sable et al., 1991; Kazimierczuk and Starman, 1999; Abe et al., 2008; Thandi et al., 1999).

a) Output impedance of the source converter Z_{oS} : Z_{oS} is the output impedance of the source converter independent on its load. The characteristic of Z_{oS} is depicted by dotted lines, as shown in Fig. 1.7(a). Here, if the source converter is a switching-mode power supply, f_{CS} is the cut-off frequency of its voltage loop. If the source converter is a *LC* input filter, f_{CS} is the filter resonant frequency. It is obvious that, Z_{oS} is similar to the output impedance of a *LC* filter. If $f < f_{CS}$, Z_{oS} presents the characteristics of an inductance; and



Figure 1.8: The equivalent circuit model of the cascaded system — A CPL cascaded with a *LC* input filter.

if $f > f_{CS}$, Z_{oS} presents the characteristic of the source output filter capacitor. It is worth pointing out that, for the source converter, its open-loop output impedance is higher than its closed-loop output impedance, and if the source converter is a *LC* input filter, its output impedance is higher than other type source output impedance (S. Abe and Ninomiya, 2006).

b) Input impedance of the load converter Z_{iL} : Generally, Z_{iL} refers specifically to the input impedance of the load converter operating in continuous current mode (CCM). The characteristic of Z_{iL} is depicted by solid lines, as shown in Fig. 1.7(a), where f_{CL} is the cut-off frequency of the voltage loop of the load converter. If $f < f_{CL}$, Z_{iL} behaves as a negative resistor with the value of $-\frac{V_{bus}^2}{P_o}$ (Hankaniemi et al., 2005), where V_{bus} is the intermediate bus voltage and P_o is the output power of the load converter; and if $f > f_{CL}$, Z_{iL} behaves as an inductor. It is worth mentioning that when $f < f_{CL}$, $|Z_{iL}|$ is inversely proportional to P_o (Middlebrook and Cuk, 1976; Sable et al., 1991; Kazimierczuk and Starman, 1999).

1.1.2.3 Physical Explanation of the System Instability Problem

According to sub-section 1.1.2.2, Z_{oS} is similar to the output impedance of a *LC* filter, and Z_{iL} behaves as a negative resistor if the load converter is controlled as a CPL. Therefore, from the impedance point of view, the cascaded system can be simplified into an equivalent *LC* input filter cascaded with a negative resistor. As shown in Fig. 1.8, L_f and C_f are the inductance and capacitor of the equivalent *LC* input filter, respectively. R_{inc} is the equivalent negative resistor.

As shown in Fig. 1.8, the transfer function of the system can be described by the wellknown second-order function:

$$T(s) = \frac{v_{bus}}{v_{in}} = \frac{1}{s^2 L_f C_f + s \frac{L_f}{R_{inc}} + 1}$$
(1.6)

By introducing the resonant frequency ω_o and the damping factor ζ , (1.6) can be re-

arranged in a more familiar format:

$$T(s) = \frac{1}{\frac{s^2}{\omega_o^2} + 2\zeta \frac{s}{\omega_o} + 1}$$
(1.7)

where

$$\omega = \frac{1}{\sqrt{L_f C_f}} \tag{1.8}$$

$$\zeta = \frac{L_f \omega_o}{2R_{inc}} \tag{1.9}$$

If the input is excited with a unit-step function, the following Laplace equation can be obtained:

$$Y(s) = \frac{1}{\frac{s^2}{\omega_o^2} + 2\zeta \frac{s}{\omega_o} + 1} \frac{1}{s}$$
(1.10)

To obtain the response in the time domain, the inverse Laplace transform is utilised in (1.10). The result can be derived as:

$$Y(t) = 1 - \frac{e^{-\zeta \omega_0 t}}{\sqrt{1 - \zeta^2}} \sin(\omega_d t + \theta) \qquad for \, \zeta < 1 \tag{1.11}$$

where

$$\omega_d = \omega_o \sqrt{1 - \zeta^2} \tag{1.12}$$

$$\theta = \cos^{-1}(\zeta) \tag{1.13}$$

It is known that, the poles affecting (1.7), denoted ρ_1 and ρ_2 , represent the denominator roots for which $s^2 + 2\zeta \omega_o s + \omega_o^2 = 0$. Depending on ζ value, these poles affect the stability of the system described by (1.10) (Ozbay, 1999):

- $\zeta < 0$ In that case, poles affecting (1.10) feature a positive real portion. Whatever the excitation level, the transient response diverges.
- $\zeta = 0$ This particular case implies two pure imaginary poles $\rho_{1,2} = \pm j\omega_o$, making the system output permanently oscillating (no decay).
- ζ > 0 The two poles now have a real portion (ohmic losses), and the system exhibits different responses depending on whether ζ > 1 (over-damping), ζ = 1 (critical damping), or 0 < ζ < 1 for which we obtain a decaying oscillating response.

According to (1.9), if $R_{inc} < 0$, then $\zeta < 0$, and therefore the whole system will be unstable and its transient response diverge at any level of excitation. That is the physical explanation of why a CPL may become unstable with its *LC* input filter. In addition, it can also be seen that the -180° phase resulted by the load negative resistor characteristic is the main reason for the instability of the cascaded system (Abe et al., 2008; Thandi et al., 1999).

1.1.2.4 The Worst Case of Instability Problem in Cascaded Systems

Due to the characteristics of Z_{oS} , when the source converter is a *LC* filter, its output impedance is higher than other types of source converters (S. Abe and Ninomiya, 2006). Due to the characteristics of Z_{iL} , when the load converter operates at a full load, it has the lowest negative resistor. According to (1.9) and (1.11), the lower negative resistor, the larger oscillation of the unstable cascaded system. Therefore, the worst case of instability problem of a cascaded system occurs at full load and when the source converter is a *LC* input filter (Zhang et al., 2013).

1.2 Literature Review

Since the instability problem is very likely to occur in a practical cascaded system, the question of how to solve the instability problem of the cascaded system has been receiving increasing attention. At present, many impedance-based stabilisation methods have been proposed for the cascaded system, and they can be divided into two types: hardware-based (Erickson and Maksimov, 2001; Cespedes et al., 2011; Marx et al., 2012; Abe et al., 2008; Zhang et al., 2013) and software-based (Rahimi and Emadi, 2009; Karppanen et al., 2007; Wu and Lu, 2015; Li et al., 2015; Liu et al., 2007a; Wang et al., 2005). For hardware-based methods, the passive components, such as resistors, capacitors, inductors, and converters are utilised to change the source output impedance or load input impedance to stabilise the cascaded system. For software-based methods, advanced control methods are utilised to modify the source output impedance or load input impedance to ensure the stability of the cascaded system.



Figure 1.9: Typical dampers and their application: (a) *RC* parallel damper; (b) *RL* parallel damper; (c) *RL* series damper; (d) Application.



Figure 1.10: Stabilise the cascaded system by adding bus capacitor or adaptive active capacitor converter: (a) adding bus capacitor; (b) adding adaptive active capacitor converter.

1.2.1 Hardware-Based Stabilisation Methods

Since the reduced system damping caused by the CPL is the root cause of the instability problem in cascaded systems (Middlebrook, 1979), the most intuitive hardware-based method is to increase the system damping by using dampers (Erickson and Maksimov, 2001; Cespedes et al., 2011; Marx et al., 2012). Three typical dampers are depicted in Fig. 1.9, where the elements in the dashed line are used for damping. Due to the structure and position of the damper, the three dampers are called the *RC* parallel damper (Fig. 1.9(a)), the *RL* parallel damper (Fig. 1.9(b)) and the *RL* series damper (Fig. 1.9(c)), respectively. In the system shown in Fig. 1.9(d) with these dampers, the peak of the output impedance of the *LC* input filter can be damped, and a total separation between $|Z_o|$ and $|Z_{iL}|$ ensured to stabilise the whole system.

It is worth pointing out that if the input filter is a π , *LCL* or other high order filters, the above dampers may be not useful. This is because the characteristic of the output impedance of these high order filters is more complex than the *LC* filter. For instance, the output impedance of the *LC* filter only contains one resonant peak, but the output impedance of both π and *LCL* filters contain two resonant peaks. However, the above dampers cannot simultaneously damp two resonant peaks. It is also worth pointing out that, the input filter only refers to the *LC* filter and the high order filters will not be discussed in this thesis. In addition, if the source stage is a closed-loop converter with *LC* output filter, the above dampers are not suitable either. Though the above dampers can dampen the output impedance of the closed-loop converter, they also increase the order of the converter and may make the converter itself unstable. As a result, the above dampers are specifically proposed for the application whose source stage is a *LC* input filter.

However, in some cascaded systems, the source converter is a switching power converter, and not a *LC* input filter. For instance, in electric vehicle on-board DC power systems, the feeder of the CPL is usually another power converter. It is clear that the passive dampers in Fig. 1.9 may not be suitable for such cascaded systems. Therefore, another two hardware-based stabilisation methods are proposed as shown in Fig. 1.10 (Abe et al., 2008; Zhang et al., 2013).

In Fig. 1.10(a), the capacitance of the bus capacitor C_{bus} (Abe et al., 2008; 2006) or the original load input capacitor C_{in} (An and Lu, 2016) is increased. As a result, both the source converter and the load converter are entirely decoupled by the C_{bus} or C_{in} . Thus, system stability is also ensured. However, this stabilisation method has some unavoidable drawbacks: 1) in order to stabilise the cascaded system, a large capacitance value of the bus capacitor is required and the electrolytic capacitor has to be utilised. However, the electrolytic capacitor also means worse reliability for the cascaded system (Ma et al., 2013); 2) a larger C_{bus} or C_{in} results in poor dynamic performance of the cascaded system (Thandi et al., 1999); and 3) a larger C_{bus} or C_{in} reduces the system power density (Cao et al., 2015).

In order to overcome the drawbacks of the stabilisation method in Fig. 1.10(a), an auxiliary DC/DC converter, known as an adaptive active capacitor converter (AACC), is proposed to mimic the function of C_{bus} (Zhang et al., 2013). Fig. 1.10(b) presents a typical structure of an AACC, which contains a capacitor, an inductor and two switchers. It is worth mentioning that the capacitor in the AACC is very small and can be realised by a film capacitor. Therefore, this AACC not only has the same stabilisation function as C_{bus} , but also does not deteriorate the reliability of the cascaded system. In addition, the AACC ensures the dynamic performance of the cascaded system via an advanced control method. However, the weak side of the AACC stabilisation method is that an extra DC/DC converter is required resulting in additional cost and power losses. Therefore, as with the larger C_{bus} or C_{in} , the AACC also reduces the power density of the system.

1.2.2 Software-Based Stabilisation Methods

As hardware-based stabilisation methods can lead to significant power loss, software-based methods based on advanced control of the source converter (Rahimi and Emadi, 2009; Karppanen et al., 2007; Wu and Lu, 2015; Li et al., 2015) or the load converter (Liu et al., 2007a; Wang et al., 2005) are proposed.

Fig. 1.11 shows the existing software-based stabilization methods for the source converter (Rahimi and Emadi, 2009; Karppanen et al., 2007; Wu and Lu, 2015; Li et al., 2015). In Fig. 1.11(a), an inductor current inner loop is introduced to the source converter to add a virtual resistor in a series with the filter inductor (Rahimi and Emadi, 2009; Karppanen et al., 2007). With this virtual resistor, the resonant peak of the source output impedance

can be damped, and hence the stability of the cascaded system is ensured. In Fig. 1.11(b), the droop control is utilised by the source converter (Wu and Lu, 2015). As a result, a large virtual resistor is added in the series, with the output of the source converter to damp the oscillation of the cascaded system. In Fig. 1.11(c), the output voltage differential inner loop is used by the source converter to add a virtual resistor in parallel with its output (Li et al., 2015). With this output voltage differential inner loop, the output impedance of the source converter can be greatly reduced so that the system instability problem can be solved.

Fig. 1.12 presents the existing software-based stabilisation methods for the load converter (Liu et al., 2007a; Wang et al., 2005). In Fig. 1.12(a), a stabilising power is injected into the load converter (Liu et al., 2007a). With this injected power, a virtual capacitor or resistor is equivalently added in parallel with the input port of the load converter to remove its CPL characteristic. In Fig. 1.12(b), the regulation signal maps of the source converter is developed and sent to the control block of the load converter to shape the load input impedance (Wang et al., 2005).

1.3 Motivation and Objective

As discussed above, prior to this research, many impedance-based solutions have been proposed to stabilise the cascaded system, and they can be classified into two types: hardware-based (Erickson and Maksimov, 2001; Cespedes et al., 2011; Zhang et al., 2013) and software-based (Rahimi and Emadi, 2009; Karppanen et al., 2007; Wu and Lu, 2015; Li et al., 2015; Liu et al., 2007a; Wang et al., 2005). However, though these stabilisation methods are able to stabilise the cascaded system, they share three drawbacks.

Firstly, most software-based stabilisation methods focus on changing the source output impedance, whilst the software-based stabilisation methods of regulating the load input impedance are rarely reported. However, load software-based stabilisation methods are necessary in practice. For instance, if the source converter is a *LC* input filter, existing source software-based stabilisation methods cannot control the *LC* input filter and cannot stabilise the cascaded system. Though passive dampers can be utilised at this moment, doing so brings significant power loss to the cascaded system. Therefore, of the existing



Figure 1.11: Software-based stabilisation methods for source converters: (a) adding virtual resistor in series with the filter inductor; (b) adding virtual resistor in series with the source output; (c) adding virtual resistor in parallel with the source output.



Figure 1.12: Software-based stabilisation methods for load converters: (a) adding virtual resistor or capacitor in parallel with the load input; (b) sending the regulation signal from source converter to load converter.

stabilisation methods, only the method in (Liu et al., 2007a) is suitable in this case. It is worth pointing out that though the method in (Wang et al., 2005) is also a load stabilisation software-based method, it requires the regulation signal from the source control block. Thus, this method is also unachievable when the source converter is a *LC* input filter.

Secondly, most of the existing stabilisation methods need the information of both the regulated converter and the other converters. For instance, the existing load stabilisation methods need to shape the load input impedance according to the characteristics of the source output impedance. Similarly, the existing source stabilisation methods need to shape the source output impedance according to the characteristics of the load input impedance. In other words, when the source or load converter changes, the existing load or source stabilisation method must be re-designed, which increases the development cycles and cost of the whole cascaded system. In (Zhang et al., 2013), the authors studied this problem and proposed that AACC be used to stabilise the cascaded system without requiring any information of the source or load converter. However, AACC reduces the power density of the cascaded system and is unacceptable for weight-limited or/and volume-limited systems, such as aircraft and hybrid electrical vehicles power supply systems (Wen et al., 2012; Wang et al., 2011).

Thirdly, all the existing stabilisation methods are focused on how to shape the source output impedance or load input impedance, but ignore their impact on the original performance of the cascaded system. However, stabilising the cascaded system can impede keeping the original performance of the cascaded system. For instance, if the load converter wants to achieve a good dynamic performance, its input impedance should be controlled as a negative resistor from small signal analysis, but this negative resistor characteristic is bad news for the stability of a cascaded system. Another example is that of the passive dampers: though the existing *RC* or *RL* dampers can damp the output impedance of the source converter and stabilise the cascaded system, they also bring slower dynamic performance and reduce EMI ability of the source converter.

Therefore, the objective of this thesis is to propose a family of software-based stabilisation methods to overcome the above drawbacks. It will do so in the following ways:

a) To overcome the first drawback, this thesis proposes a family of load software-based stabilisation methods to shape the load input impedance only in a specific frequency range.

As a result, the proposed load software-based stabilisation methods would not only stabilise the cascaded system, but also keep most of the performance of the original load converter.

b) To overcome the second drawback, this thesis introduces the adaptive function to the proposed load software-based stabilisation methods. With the adaptive stabilisation methods, the load converter would stably connect to different source converters without changing its internal structure.

c) To overcome the third drawback, this thesis investigates the impact of the existing stabilisation methods via two-port network analysis, and proposes two software-based methods to stabilise the cascaded system while improving the system performance.

However, it should be stressed that all the proposed methods are devoted to solving the instability problem of the cascaded system formed by just one source converter and one load converter.

1.4 Outline of the Thesis

There are nine chapters in this thesis. Chapter 1 is the introduction, and Chapters 2 to 8 present the proposed stabilisation methods. Chapter 9 concludes the thesis and makes suggestions for further work. The thesis is organised as follows.

In Chapter 2, a parallel-virtual-impedance (PVI) control strategy is proposed to solve the instability problem of the cascaded system. It connects a virtual impedance in parallel with the input port of the load converter so that the magnitude or phase of the load input impedance is modified in a very small range of frequency. As a result, with the PVI control strategy, the cascaded system can be stabilised with minimised load compromise. The requirements, design considerations and experimental verification of the PVI control strategy are presented in this chapter.

Since the PVI control strategy can add a virtual impedance in parallel with the load converter to stabilise the cascaded system with minimised load compromise, it seems natural that a corresponding virtual impedance can also be added in series with the load converter to realise the same function. Therefore, a series-virtual-impedance (SVI) control strategy is proposed in Chapter 3, which can also be treated as the development and complement of the PVI control strategy. The concept, derivation, control realisation and experimental verification of the SVI control strategy are presented in Chapter 3.

Though a family of load stabilisation methods, such as PVI and SVI control strategies, are proposed in Chapters 2 and 3 respectively, they require detailed information on the source converters. As a result, if the source converter is changed, the PVI or SVI control strategy must be re-designed for the load converter. It is obvious that the characteristics of the PVI and SVI control strategies increase the development cycles and cost of the cascaded system. In order to solve this problem, the adaptive function is introduced to the PVI control strategy to generate another new control strategy — the adaptive-PVI (APVI) control strategy in Chapter 4. With the APVI control strategy, the load converter can be stably connected to different source converters without changing its internal structure. The concept, realisation and experimental verification of the APVI control strategy are presented in Chapter 4.

In Chapter 5, it is shown that the cascaded system can be stabilised via amplitude compensation (SAC) or phase compensation (SPC) for the input impedance of the load converter. Both PVI and SVI control strategies are essentially two realisation methods for SAC and SPC. Therefore, the SAC and SPC are firstly analysed and compared in Chapter 5. It is shown here that, though both SAC and SPC can stabilise the cascaded system, the system utilising the SAC is unconditionally stable, but conditionally stable when utilizing the SPC, i.e., the cascaded system with SAC is more stable than that with SPC. After this, the comparison is also carried out for the PVI and SVI control strategies. The results shows that the PVI control strategy has inevitable limitations when realising the SAC during the whole load and input voltage range of the load converter, though the SVI control strategy does not have limitations. Therefore, compared to the PVI control strategy, the SVI control strategy is more suitable for the SAC. Furthermore, the adaptive concept proposed in Chapter 4 is also introduced here to the SVI control strategy to generate the adaptive-SVI (ASVI) control strategy. Similar to the APVI control strategy, the ASVI control strategy stably connects the load converter to different source converters without changing its internal structure. Moreover, the ASVI control strategy can make the cascaded system more stable than the APVI control strategy. The concept, control block, realisation and experimental verification of the ASVI control strategy are also presented in Chapter 5.

The ASVI control strategy utilises a proportional-integral (PI) controller to find the

centre frequency of the ASVI. However, this PI controller may miss the right frequency with excessive proportional or integral coefficients and lead to the failure of the ASVI control strategy. Although this potential problem may not arise with small proportional and integral coefficients, it is a potential problem for the ASVI control strategy. To make the ASVI control strategy more reliable, this problem is studied in Chapter 6. The relationship between the bus voltage ripple and the centre frequency of the ASVI is firstly analysed. It is found that the relationship curve behaves like an inverted power-voltage curve of a PV cell. Therefore, a perturb and observe (P&O) algorithm based minimum-ripple-pointtracking (MRPT) controller is proposed to find the centre frequency of ASVI instead of the original PI controller. With the MRPT controller, the ASVI control strategy can find the centre frequency of the ASVI without the above problem. Furthermore, in order to find the centre frequency of ASVI quickly, the sinusoidal-tracking-algorithm (STA) is further introduced into the MRPT controller to improve its processing speed. Both the potential problem of the original ASVI control strategy with the PI controller and the effectiveness of the improved ASVI control strategy with the proposed MRPT controller are discussed and experimentally verified in Chapter 6.

The analysis in Chapters 2 to 6 indicates that the ASVI control strategy is the best load stabilisation method for the cascaded system. The ASVI control strategy can not only make the cascaded system become an unconditional stable system via shaping the load input impedance, but also changes the load input impedance only in a very small frequency range to minimise its impact on the original load converter.

However, though the ASVI control strategy has already greatly reduced its impact on the load converter, its rest impact is negative. In order to solve this problem, Chapter 7 moves the ASVI from the load side to the source side via a proposed source-side SVI (SSVI) control strategy for the source converter. The proposed SSVI control strategy not only has the same stabilisation function with the ASVI control strategy, but also improves the performance of the source converter. In addition, since the SSVI control strategy is realised by changing the control block of the source converter, the performance of the load converter is not affected. Therefore, the SSVI control strategy can be treated as a supplement to, and improvement on, the ASVI control strategy. Additionally, due to the method of realisation, the SSVI control strategy also can be divided into the source stabilisation methods of the cascaded system. It is also worth pointing out that since the SSVI control strategy stabilises the system by reducing the source output impedance, it is also suitable for the DPS with multiple upstream converters and one downstream converter.

As discussed above, the *LC* filter at the input of a DC/DC converter may cause instability when the converter is controlled as a CPL and one of the effective solutions is to reduce the output impedance of the *LC* input filter by different stabilisation dampers. In Chapter 8, the impact of these dampers on the *LC* filter is analysed via two-port network analysis, and it is found that the existing dampers all degrade the performance of the original *LC* input filter to some extent. In order to overcome this issue, an *RLC* damper is proposed to stabilise the whole system while improving the performance of the *LC* input filter. In addition, this *RLC* damper is also designed to achieve real robustness against the parameter variations of the *LC* input filter. Furthermore, in order to avoid the power loss when physically implementing the damper, a control strategy for the CPL is proposed to implement the *RLC* damper as a virtual *RLC* (*VRLC*) damper. The actual effectiveness of the *VRLC* damper and its impact on the CPL are fully evaluated via two-port network analysis, and experimentally verified in Chapter 8.

Finally, Chapter 9 concludes the thesis and makes recommendations for future work.

1.5 Major Contributions

The major contribution of this research is to propose a family of impedance-control-based methods to stabilise the cascaded system without the drawbacks of the existing solutions. It achieves this primarily in six ways:

 A set of virtual-impedance-based control strategies are proposed in this thesis – PVI (Chapter 2) and SVI (Chapter 3). The PVI and SVI control strategies connect a virtual impedance in parallel or series with the input impedance of the load converter so that the magnitude or phase of the load input impedance is modified in a small frequency range to ensure the stability of the whole system. Since PVI and SVI control strategies only modify the load input impedance in a very small frequency range, they can stabilise the cascaded system with minimised load compromise. Therefore, the PVI and SVI control strategies can not only be treated as the development and complement of the load stabilisation methods, but also as good solutions for overcoming the contradiction between the stability and performance of the cascaded system. In addition, it is worthy of mention that, since both PVI and SVI control strategies realise the impedance decoupling of the source and load converters, it could not only solve the instability problem of the cascaded system but also have the benefit of removing the input capacitor of the load converter whose original main function is impedance decoupling.

- 2. Adaptive concept is introduced to stabilisation methods to generate two novel software-based stabilisation methods: APVI (Chapter 4) and ASVI (Chapter 5) control strategies. With the APVI and ASVI control strategies, the load converter can be stably connected to different source converters without changing its internal structure. In contrast to the existing stabilisation methods, the APVI and ASVI control strategies avoid the redesign of the load converter, and hence save the cost for the cascaded system.
- 3. The SAC and SPC methods are compared in Chapter 5. It is known that both SAC and SPC are two popular load stabilisation methods for the cascaded system. However, prior to this thesis, there was no clear conclusion on which was better or worse. This thesis, however, finds that though both SAC and SPC can stabilise the cascaded system, the system utilising the SAC is unconditionally stable but conditionally stable when utilising the SPC, i.e., the cascaded system with SAC is more stable than that with SPC.
- 4. The PVI and SVI control strategies are compared in Chapter 5. The results indicate that the PVI control strategy has inevitable limitations when realising the SAC during the whole load and input voltage range of the load converter, whilst the SVI control strategy does not have limitations. Therefore, the SVI control strategy is more suitable for the SAC than the PVI control strategy.
- 5. The potential problem of the ASVI control strategy is solved in Chapter 6. Though the ASVI control strategy is so far the best load stabilisation method, it is not perfect. Since the ASVI control strategy utilises a PI controller to find the centre frequency of the ASVI, the PI controller may miss the right frequency with excessive proportional or integral coefficients and cause the failure of the ASVI control strategy. Although

this problem will not necessarily arise with small proportional and integral coefficients, it is potentially an issue for the ASVI control strategy. To make the ASVI control strategy more reliable, a MRPT controller is proposed to replace the original PI controller. With the MRPT controller, the ASVI control strategy could find the centre frequency of the ASVI without the aforementioned problem arising. In addition, the MRPT controller is further improved by STA to help it quickly find the centre frequency of ASVI.

- 6. SSVI control strategy is proposed to stabilise the cascaded system while improving the performance of the source converter (Chapter 7). Though the existing stabilisation methods can stabilise the cascaded system, they all degrade the performance of the original cascaded system to some extent. Even the proposed PVI and SVI control strategies only keep most of the performance of the load converter: they cannot improve it. However, the SSVI control strategy solves this problem by not only making the cascaded system stable, but also making the source converter work better. In addition, it is worth pointing out that since the SSVI control strategy stabilises the system by reducing the source output impedance, it is not only suitable for the cascaded system with one upstream converter and one downstream converter, but also for the distributed power system with multiple upstream converters and one downstream converter.
- 7. *VRLC* damper is proposed to stabilise the cascaded system while improving the performance of the *LC* input filter (Chapter 8). Though the existing dampers can solve the instability problem between the CPL and its *LC* input filter, they degrade the performance of the original *LC* input filter to some extent. In order to overcome this drawback, a *VRLC* damper is proposed. This can not only stabilise the cascaded system, but also improve the performance of the *LC* input filter and improve the system efficiency compared to the existing dampers.

1.6 List of Publications

1. Xin Zhang, Qing-Chang Zhong, "Improved Adaptive-Series-Virtual-Impedance Control Incorporating Minimum Ripple Point Tracking for Load Converters in DC Systems", IEEE Transactions on Power Electronics, Volume: 31, Issue: 12, 8088-8095, 2016

- Xin Zhang, Qing-Chang Zhong, Wen-Long Ming, "A Virtual *RLC* Damper to Stabilize DC/DC Converters Having an *LC* Input Filter while Improving the Filter Performance", IEEE Transactions on Power Electronics, Volume: 31, Issue: 12, 8017-8023, 2016.
- Xin Zhang , Qing-Chang Zhong, Wen-Long Ming, "Stabilization of a Cascaded DC Converter System via Adding a Virtual Adaptive Parallel Impedance to the Input of the Load Converter", IEEE Transactions on Power Electronics, Volume: 31, Issue: 3, 1826 – 1832, 2016.
- Xin Zhang, Qing-Chang Zhong, Wen-Long Ming, "Stabilization of Cascaded DC/DC Converters via Adaptive Series-Virtual-Impedance Control of the Load Converter", IEEE Transactions on Power Electronics, Volume: 31, Issue: 9, 6057 - 6063, 2016.
- Wen-Long Ming, Qing-Chang Zhong and Xin Zhang, "A single-phase four-switch rectifier with significantly reduced capacitance," IEEE Transactions on Power Electronics, Volume: 31, Issue: 2, 1618-1632, 2016.
- Xin Zhang ,Qing-Chang Zhong, Wen-Long Ming, "Adaptive series-virtual-impedance control strategy for the load converter to improve the stability of the cascaded system", IEEE PEDG, Vancouver, Canada, 27-30 June 2016.
- Xin Zhang ,Qing-Chang Zhong, Wen-Long Ming, "Source-side Series-virtual-impedance Control Strategy to Stabilize the Cascaded System with Improved Performance", IEEE ECCE, Milwaukee, Wisconsin, USA, 18-22 September 2016.
- Xin Zhang ,Qing-Chang Zhong, Wen-Long Ming, "A Novel Stabilization Method of DC/DC Converters Having an *LC* Input Filter via Utilizating a Virtual *RLC* Damper", IEEE ECCE, Milwaukee, Wisconsin, USA, 18-22 September 2016.
- Xin Zhang, Xin-Bo Ruan, Qing-Chang Zhong, "Improving the Stability of Cascaded DC/DC Converter Systems via Shaping the Input Impedance of the Load Converter

With a Parallel or Series Virtual Impedance", IEEE Transactions on Industrial Electronics, Volume:62 Issue:12, Pages: 7499 – 7512, 2015.

- Xin Zhang ,Qing-Chang Zhong, Wen-Long Ming, Xin-Bo Ruan, "Stabilization of a cascaded DC system via adding a virtual impedance in series with the load converter", IEEE PEDG 2015, Pages: 677-683.
- Xin Zhang ,Qing-Chang Zhong, Wen-Long Ming, Xin-Bo Ruan, "Stabilization of a cascaded DC system via adding a virtual impedance in parallel with the load converter", IEEE PEDG 2015, Pages: 335-341.
- Xin Zhang ,Qing-Chang Zhong, Wen-Long Ming, S. Gadelovits, "Impedance-based local stability criterion for standalone photovoltaic-battery hybrid system", IEEE PEDG 2015, Pages: 717-723.
- Xin Zhang ,Qing-Chang Zhong, Wen-Long Ming, "Fast scale instability problem of cascaded buck conversion system and its phase-shifted-carrier solution", IEEE ECCE 2015, Pages: 2684 - 2689.

Chapter 2

Parallel-Virtual-Impedance (PVI) Control for Load Converters

As discussed in Chapter 1, a load stabilisation method is necessary for the cascaded system because the source output impedance cannot be changed by control method in some special applications. For instance, if the source converter is a *LC* input filter, the existing software-based source stabilisation method cannot modify its output impedance and cannot stabilise the cascaded system either. Therefore, this chapter proposes a parallel-virtual-impedance (PVI) control strategy, which connects a virtual impedance in parallel with the input port of the load converter so that the magnitude or phase of the load input impedance is modified in a small frequency range, thus solving the instability problem of the cascaded system. As a result, even if the source converter is a *LC* input filter, the cascaded system still can be stabilised with the minimised load compromise via the proposed PVI control strategy. The requirements and design consideration of the PVI control strategy are discussed in this chapter.

The rest of this chapter is organised as follows. In Section 2.1, the basic idea of the load shaping method is discussed, and the concept, derivation and realisation of the PVI control strategy are presented. Two design examples of the PVI control strategy are then given in Section 2.2. Based on the design examples, two experimental/simulation cascaded systems are fabricated to verify the effectiveness of the PVI control strategy in Sections 2.3 and 2.4, respectively; and in Section 2.5, this Chapter is summarised.



Figure 2.1: Basic idea of load shaping methods: (a) only increasing $|Z_{iL}|$ at $[f_1, f_2]$ to make a total separation between $|Z_{oS}|$ and $|Z_{iL}|$; (b) only changing $\varphi(Z_{iL})$ at $[f_1, f_2]$ to ensure $|\varphi(Z_{oS}) - \varphi(Z_{iL})| < 180^0$ at $[f_1, f_2]$.

2.1 PVI Control Strategy

2.1.1 Basic Idea of Load Shaping Methods

It is understood that if the designer wants to make the cascaded system stable while keeping most of the dynamic performance of the original load converter, the best method is to change Z_{iL} only in the vicinity of the intersection frequencies of $|Z_{oS}|$ and $|Z_{iL}|$. Here, Z_{oS} is the source output impedance, and Z_{iL} is the load input impedance. As shown in Fig. 2.1, f_1 and f_2 are the intersection frequencies of $|Z_{oS}|$ and $|Z_{iL}|$.

Fig. 2.1 illustrates that there are two better ways to regulate Z_{iL} : (a) only increasing $|Z_{iL}|$ between f_1 and f_2 to make a total separation of $|Z_{oS}|$ and $|Z_{iL}|$ (See Fig. 2.1(a));



Figure 2.2: A cascaded system with parallel virtual impedance.

and (b) only increasing $\varphi(Z_{iL})$ at $[f_1, f_2]$ to ensure $|\varphi(Z_{oS}) - \varphi(Z_{iL})| < 180^0$ during $[f_1, f_2]$ ((See Fig. 2.1(b))). In Fig. 2.1, Z_{iLP} is the input impedance of the load converter with the latter proposed PVI control strategy.

2.1.2 Expression of the Parallel-Virtual-Impedance $Z_{PVI}(s)$

In fact, the load shaping idea in Section 2.1.1 can be realised by adding a virtual impedance $Z_{PVI}(s)$ in parallel with the load input port. As shown in Fig. 2.2, after adding $Z_{PVI}(s)$, the shaped load input impedance $Z_{iLP}(s)$ can be expressed as

$$Z_{iLP}(s) = Z_{iL}(s) \parallel Z_{PVI}(s) = \frac{Z_{iL}(s) \cdot Z_{PVI}(s)}{Z_{iL}(s) + Z_{PVI}(s)}$$
(2.1)

where $Z_{iL}(s)$ is the original input impedance of the load converter and if $f < f_{CL}$ its expression is $-V_{bus}^2/P_o$. Here, V_{bus} and P_o are the input voltage and output power of the load converter, respectively.

a) According to Fig. 2.1(a), if a total separation between $|Z_{oS}|$ and $|Z_{iLP}|$ is needed and the minimised change of Z_{iL} is also required, Z_{iLP} should satisfy:

$$If \{ f \in [f_1, f_2] \} \rightarrow |Z_{iLP}| > Z_{oSP}$$

$$(2.2)$$

$$If \{ f \notin [f_1, f_2] \} \rightarrow Z_{iLP} = Z_{iL}$$

$$(2.3)$$

where Z_{oSP} is the peak value of $|Z_{oS}|$.

According to (2.2) and (2.3), Z_{iLP} can be further expressed as

$$Z_{iLP} = \begin{cases} -|Z_{iL1}| & f \in [f_1, f_2] \\ Z_{iL} & f \notin [f_1, f_2] \end{cases}$$
(2.4)

where $|Z_{iL1}| > Z_{oSP}$ is the amplitude of Z_{iLP} during $[f_1, f_2]$.

According to (2.1) and (2.4), in order to ensure $|Z_{iL}|$ become $|Z_{iLP}|$ as shown in Fig. 2.1(a) successfully, Z_{PVI} can be selected as

$$Z_{PVI} = \frac{Z_{iL}Z_{iLP}}{Z_{iL} - Z_{iLP}} = \begin{cases} \frac{(V_{bus}^2/P_o)|Z_{iL1}|}{(-V_{bus}^2/P_o) - (-|Z_{iL1}|)} & f \in [f_1, f_2] \\ +\infty & f \notin [f_1, f_2] \end{cases}$$
$$= \begin{cases} -\frac{V_{bus}^2|Z_{iL1}|}{V_{bus}^2 - |Z_{iL1}|P_o} & f \in [f_1, f_2] \\ +\infty & f \notin [f_1, f_2] \end{cases}$$
(2.5)

b) Similarly, according to Fig. 2.1(b), if $|\varphi(Z_{oS}) - \varphi(Z_{iLP})| < 180^{0}$ is needed to be ensured within $[f_1, f_2]$ and the minimised change of Z_{iL} is also required, Z_{iLP} should satisfy:

$$If \{ f \in [f_1, f_2] \} \rightarrow |\theta| < 90^0$$
 (2.6)

$$If \{ f \notin [f_1, f_2] \} \rightarrow \varphi(Z_{iLP}) = \varphi(Z_{iL})$$

$$(2.7)$$

where θ is the phase of Z_{iLP} during $[f_1, f_2]$.

According to (2.6) and (2.7), Z_{iLP} can be further expressed as

$$Z_{iLP} = \begin{cases} \left(V_{bus}^2/P_o\right)e^{j\theta} & f \in [f_1, f_2] \\ Z_{iL} & f \notin [f_1, f_2] \end{cases}$$
(2.8)

According to (2.1) and (2.8), in order to ensure $|Z_{iL}|$ become $|Z_{iLP}|$ as shown in Fig. 2.1(b) successfully, Z_{PVI} can be selected as

$$Z_{PVI} = \frac{Z_{iL}Z_{iLP}}{Z_{iL} - Z_{iLP}} = \begin{cases} \frac{-(V_{bus}^2/P_o)[(V_{bus}^2/P_o)e^{j\theta}]}{(-V_{bus}^2/P_o) - [(V_{bus}^2/P_o)e^{j\theta}]} & f \in [f_1, f_2] \\ +\infty & f \notin [f_1, f_2] \end{cases}$$
$$= \begin{cases} \frac{V_{bus}^2}{P_o} \cdot \frac{e^{j\theta}}{1 + e^{j\theta}} & f \in [f_1, f_2] \\ +\infty & f \notin [f_1, f_2] \end{cases}$$
(2.9)

In summary, if the Z_{PVI} , which is expressed by (2.5), is added in parallel with the load input impedance, a total separation between $|Z_{oS}|$ and $|Z_{iLP}|$, as shown in Fig. 2.1(a), can be realised. If the Z_{PVI} , as expressed at (2.9), is added in parallel with the load input impedance, $|\varphi(Z_{oS}) - \varphi(Z_{iLP})| < 180^{\circ}$ within $[f_1, f_2]$, as shown in Fig. 2.1(b), can be



Figure 2.3: Small signal block diagram of the original load converter.

\hat{v}_{bus}	Perturbation of the bus voltage
î	Perturbation of the load current
î _{bus}	Perturbation of the bus current
\hat{d}	Perturbation of the duty cycle
\hat{v}_o	Perturbation of the output voltage
\hat{v}_{or}	Perturbation of the output voltage reference
$G_{v}(s)$	Transfer function of the voltage regulator
$G_M(s)$	Transfer function of the modulator
$Z_{iOL}(s)$	Open-loop input impedance
$G_{vd}(s)$	Control to output voltage transfer function
$G_{id}(s)$	Control to input current transfer function
$Z_{oOL}(s)$	Open-loop output impedance
$H_{s}(s)$	Sampling coefficient of the output voltage
$\overline{G_{iOL}(s)}$	Open-loop load to input current transfer function
$\overline{G_{vOL}(s)}$	Open-loop input to output voltage transfer function

Table 2.1: Definitions of variables and transfer functions in load converter

ensured.

2.1.3 Derivation of the PVI Control Strategy

As shown in Section 2.1.2, with Z_{PVI} , the cascaded system can be stabilised by changing $|Z_{iL}|$ or $\varphi(Z_{iL})$ in a very small frequency range to minimise the load compromise. Hence, in this section, a PVI control strategy for the load converter is proposed to realise Z_{PVI} .

First of all, the small signal block diagram of the original load converter is presented in Fig. 2.3. As seen, v_{bus} and i_{bus} are the input voltage and current of the load converter, respectively; and v_o and i_o are the output voltage and current of the load converter, respect-



Figure 2.4: Small signal block diagram of the load converter with Z_{PVI} .

ively. *d* is the duty cycle of the load converter. $Z_{iOL}(s)$, $G_{iOL}(s)$, $G_{id}(s)$, $G_{vOL}(s)$, $Z_{oOL}(s)$ and $G_{vd}(s)$ are the six basic open-loop transfer functions of the load converter. In addition, for the original load converter, a voltage closed-loop is utilised, where $H_s(s)$ and $G_v(s)$ are the sampling coefficient and voltage regulator transfer function of the output voltage, respectively, and $G_M(s)$ is the transfer function of the modulator. All the variables and transfer functions of Fig. 2.3 are described in Table 2.1.

 $Z_{iL}(s)$ is the input impedance of the original load converter. Therefore, for the original load converter, its i_{bus} and v_{bus} have the following relationship:

$$\frac{i_{bus}}{v_{bus}} = \frac{1}{Z_{iL}(s)} \tag{2.10}$$

If Z_{PVI} is required to be added in parallel with the input port of the load converter, the relationship between i_{bus} and v_{bus} is changed to:

$$\frac{i_{bus}}{v_{bus}} = \frac{1}{Z_{iL}(s)} + \frac{1}{Z_{PVI}(s)}$$
(2.11)

According to (2.11), the small signal block diagram of the load converter can be changed to Fig. 2.4.

As shown in Fig. 2.4, $1/Z_{PVI}$ is introduced to the control block between the v_{bus} and i_{bus} directly. From the small signal block diagram point of view, Fig. 2.4 achieves the purpose of adding Z_{PVI} in parallel with the input port of the load converter. However, from the control point of view, this method cannot be realised directly. It is because any converter



Figure 2.5: The PVI control strategy.

is only controlled by its duty cycle. In order to address this issue, the output of $1/Z_{PVI}$ is moved to the output voltage reference and equivalent adjustments are made to the transfer function to $G_{PVI}(s)$, as shown in Fig. 2.5.

According to Fig. 2.5, if $G_{PVI}(s)$ is required to realise the parallel virtual impedance $1/Z_{PVI}$, the following relationship can be obtained:

$$\hat{v}_{bus} \cdot \frac{1}{Z_{PVI}(s)} = \hat{v}_{bus} \cdot G_{PVI}(s) \cdot G_v(s) \cdot G_M(s) \cdot G_{id}(s) \cdot \frac{1}{1 + T_v(s)}$$
(2.12)

where $T_{\nu}(s) = H_s(s)G_{\nu}(s)G_{\nu d}(s)$ is the voltage loop gain of the load converter.

According to (2.12), the expression of $G_{PVI}(s)$ can be derived as:

$$G_{PVI}(s) = \frac{1}{Z_{PVI}(s)} \cdot \frac{1 + T_{\nu}(s)}{G_{\nu}(s) \cdot G_M(s) \cdot G_{id}(s)}$$
(2.13)

Up to this point, Fig. 2.3 ~ Fig. 2.5 present the whole derivation process of the PVI control strategy; and $G_{PVI}(s)$ is the input-impedance-regulator of the PVI control strategy.

2.1.4 Realisation of the PVI Control Strategy

As discussed above, $G_{PVI}(s)$ is the key point of the PVI control strategy. Therefore, methods of realising $G_{PVI}(s)$ are discussed carefully in this section. According to (2.5), (2.9) and Fig. 2.5, the expression of $G_{PVI}(s)$ can be derived as:



Figure 2.6: Characteristic of $|G_{PVI}(s)|$.

a) To achieve a total separation between $|Z_{oS}|$ and $|Z_{iLP}|$, $G_{PVI1}(s)$ is

$$G_{PVI1}(s) = \begin{cases} -\frac{V_{bus}^2 - |Z_{iL1}|P_o}{V_{bus}^2 |Z_{iL1}|} \cdot \frac{1 + T_v(s)}{G_v(s) \cdot G_M(s) \cdot G_{id}(s)} & f \in [f_1, f_2] \\ 0 & f \notin [f_1, f_2] \end{cases}$$
(2.14)

b) To ensure $|\varphi(Z_{oS}) - \varphi(Z_{iLP})| < 180^{\circ}$ at f_1 and f_2 , $G_{PVI2}(s)$ is

$$G_{PVI2}(s) = \begin{cases} \frac{P_o}{V_{bus}^2} \cdot \frac{1+e^{j\theta}}{e^{j\theta}} \cdot \frac{1+T_v(s)}{G_v(s) \cdot G_M(s) \cdot G_{id}(s)} & f \in [f_1, f_2] \\ 0 & f \notin [f_1, f_2] \end{cases}$$
(2.15)

According to (2.14) and (2.15), $G_{PVI}(s)$ is a frequency-based piecewise transfer function, where $|G_{PVI}(s)| > 0$ within $[f_1, f_2]$ and $|G_{PVI}(s)| = 0$ outside $[f_1, f_2]$ as shown in Fig. 2.6. Hence, a band-pass filter $G_{BPF}(s)$ can be utilised to help achieve $G_{PVI}(s)$. In other words, (2.14) and (2.15) can be re-expressed as (2.16) and (2.17), respectively.

$$G_{PVI1}(s) = -\frac{V_{bus}^2 - |Z_{iL1}| P_o}{V_{bus}^2 |Z_{iL1}|} \cdot \frac{1 + T_v(s)}{G_v(s) \cdot G_M(s) \cdot G_{id}(s)} \cdot G_{BPF}(s)$$
(2.16)

$$G_{PVI2}(s) = \frac{P_o}{V_{bus}^2} \cdot \frac{1 + e^{j\theta}}{e^{j\theta}} \cdot \frac{1 + T_v(s)}{G_v(s) \cdot G_M(s) \cdot G_{id}(s)} \cdot G_{BPF}(s)$$
(2.17)

where $G_{BPF}(s)$ adopts a typical second order band-pass filter, which is composed of a second order high-pass filter and a second order low-pass filter, given as

$$G_{BPF}(s) = \frac{s^2}{s^2 + (2\pi f_1/Q_H)s + (2\pi f_1)^2} \cdot \frac{(2\pi f_2)^2}{s^2 + (2\pi f_2/Q_L)s + (2\pi f_2)^2} \quad (2.18)$$

where Q_H and Q_L are quality factors of the high-pass and low-pass filter, respectively.

According to (2.16) and (2.17), $G_{PVI}(s)$ is easy to realise using digital control chips, such as a digital signal processor (DSP), field programmable gate array (FPGA) and micro-



Figure 2.7: The System I.

Table 2.2: Parameters of the System I

	5					
Parameter	L_{f}	C_{f}	L_{f1}	C_{f1}	R_{Ld}	
Value	700 μH	68 μF	33 μH	2400 μF	1.44Ω	

controller unit (MCU).

2.2 Design Example

As illustrated in Chapter 1, the worst instability phenomena are most likely to occur when the source converter is a *LC* input filter. Buck and Boost converters are two typical DC/DC converters, so a 100 W cascaded system, which consists of a *LC* input filter and a 48 V - 12 V Buck converter, and a 200 W cascaded system, which includes a *LC* input filter and a 24 V - 48 V Boost converter, are designed in this section.

2.2.1 System I — a Buck Converter with a LC Input Filter

A 100 W cascaded system with a *LC* input filter (source stage) and a 48 V - 12 V Buck converter (load stage) has been designed, and are shown at Fig. 2.7. The Buck converter operates at 100 kHz. The cut-off frequency and phase margin of the voltage close loop of the Buck converter is set to 20 kHz and 45° , respectively. The corresponding system parameters are summarised in Table 2.2.

The output impedance Z_{oS} of the *LC* input filter is given in (2.19), where R_{Lf} is the parasitic resistor of L_f and the measured value is 0.1 Ω . According to the small-signal circuit model of the Buck converter (Cuk, 1976), the input impedance Z_{iL} of the Buck



Figure 2.8: Bode plots of Z_{oS} and Z_{iL} of the System I.

converter can be expressed as (2.20).

$$Z_{oS}(s) = \left(sL_f + R_{Lf}\right) / \left(s^2 L_f C_f + s R_{Lf} C_f + 1\right)$$
(2.19)
$$\int c_f C_f P_f^2 + P_f^2 P_f^2$$

$$Z_{iL}(s) = \left[\frac{1}{1+T_{\nu}(s)} \cdot \frac{sC_{f1}D^2 + \frac{D^2}{R_{Ld}}}{s^2 L_{f1}C_{f1} + s\frac{L_{f1}}{R_{Ld}} + 1} - \frac{T_{\nu}(s)}{1+T_{\nu}(s)} \cdot \frac{P_o}{V_{bus}^2}\right]$$
(2.20)

where P_o and D are the output power and the duty cycle of the Buck converter, respectively.

Fig. 2.8 shows the Bode plots of Z_{oS} and Z_{iL} in the example system. As seen above, when the load is lower than 35% full-load, the system is stable; otherwise, the cascaded system is unstable and the oscillation should occur at about 685 Hz theoretically (Wildrick, 1993).

To conveniently solve the instability problem, Z_{iL} is designed to a positive resistor within $[f_1, f_2]$ to ensure $|\varphi(Z_{oS}) - \varphi(Z_{iLP})| < 180^{\circ}$, i.e., θ is set to 0 at (2.17). By the small signal model of Buck converter (Cuk, 1976) and (2.17), the expression of $G_{PVI}(s)$ can be derived as

$$G_{PVI}(s) = G_{PVI1}(s) \cdot P_o \cdot G_{BPF}(s)$$
(2.21)



Figure 2.9: Software flowchart of the PVI control strategy.

where

$$G_{PVI1}(s) = \frac{2H_s(s)R_{Ld}}{V_{bus}V_o\left[s^2L_{f1}C_{f1} + s\left(R_{Ld}C_{f1} + L_{f1}/R_{Ld}\right) + 2\right]}$$
(2.22)

here, in $G_{BPF}(s)$, $f_1 = 685 Hz$ and $f_2 = 780 Hz$; $Q_H = 0.707$ and $Q_L = 0.707$.

The Bode plot of the modified input impedance, Z_{iLP} , is shown with dashed line in Fig. 2.8. As seen, $|\varphi(Z_{oS}) - \varphi(Z_{iLP})| < 180^{0}$ at f_{1} and f_{2} . Therefore, the cascaded system will be stable and the design of $G_{PVI}(s)$ is appropriate.

Since the PVI control strategy is realised by the DSP TMS320F28335 in practice, its software flowchart is presented in Fig. 2.9. Both input and output voltage of the load converter are sampled to provide the necessary information for the PVI control strategy and to calculate the final reference of output voltage. With the final output voltage reference, the load converter is regulated by the compensator $G_{\nu}(s)$.

2.2.2 System II — a Boost Converter with a *LC* Input Filter

Fig. 2.10 shows a 200 W cascaded system consisting of a *LC* input filter (source stage) and a 24 V - 48 V Boost converter (load stage) operated at 100 kHz. Since the Boost converter is a non-minimum phase system, a proportional–integral (PI) controller is employed as



Figure 2.10: The System II.

Table 2.3: Parameters of the System II

Parameter	L_{f}	C_{f}	L_{f1}	C_{f1}	R_{Ld}
Value	250 μH	330 μF	720 μH	450 μF	11.52 Ω

the compensator, and the cut-off frequency and phase margin of the voltage closed loop of the Boost converter is set to 2.5 kHz and 32° , respectively. The corresponding system parameters are summarised in Table 2.3.

The output impedance Z_{oS} of the *LC* input filter has been given in (2.19), thus it is not repeated here. Owing to the small-signal circuit model of the Boost converter (Cuk, 1976), the input impedance Z_{iL} of the Boost converter can be expressed as (2.23).

$$Z_{iL}(s) = \left[\frac{1}{1+T_{\nu}(s)} \cdot \frac{sC_{f1}R_{Ld}+1}{s^2L_{f1}C_{f1}R_{Ld}+sL_{f1}+R_{Ld}(1-D)^2} - \frac{T_{\nu}(s)}{1+T_{\nu}(s)} \cdot \frac{P_o}{V_{bus}^2}\right]^{-1}$$
(2.23)

where D and P_o are the duty cycle and output power of the Boost converter, respectively.

Fig. 2.11 shows the Bode plots of Z_{oS} and Z_{iL} in the example system, as seen when the load is lower than 20% full-load and the cascaded system is stable. Otherwise, the cascaded system is unstable and the oscillation occurs at theoretically about 520 Hz (Wildrick, 1993).

In order to solve the instability problem conveniently, Z_{iL} is designed to a positive resistor within $[f_1, f_2]$ to ensure $|\varphi(Z_{oS}) - \varphi(Z_{iLP})| < 180^{\circ}$, i.e., θ is set to 0 at (2.17). Then, according to the small signal model of Boost converter (Cuk, 1976) and (2.17), the expression of $G_{PVI}(s)$ can be derived as

$$G_{PVI}(s) \approx G_{PVI1}(s) \cdot G_{BPF}(s)$$
 (2.24)



Figure 2.11: Bode plots of Z_{oS} and Z_{iL} of the System II.

where

$$G_{PVI1}(s) = \frac{2H_s(s)P_o}{V_{bus}V_o} \cdot \frac{R_{Ld} - sL_{f1}/(1-D)^2}{sC_{f1}R_{Ld} + 2}$$
(2.25)

Also, for $G_{BPF}(s)$, $f_1 = 520$ Hz and $f_2 = 645$ Hz, $Q_H = Q_L = 0.707$.

The Bode plot of the modified input impedance, Z_{iLP} , is represented by a dashed line in Fig. 2.11, with $|\varphi(Z_{oS}) - \varphi(Z_{iLP})| < 180^{\circ}$ at f_1 and f_2 . Therefore, the improved cascaded system is stable and the design of $G_{PVI}(s)$ is appropriate.

In practice, the Boost converter is also controlled by DSP TMS320F28335, and its PVI control strategy can be realised according to the software flowchart in Fig. 2.9. Since the flowchart has already been explained in Section 2.2.1, it is not repeated here.

2.3 Experimental Verification

According to the design examples, two experimental systems have been fabricated and tested in the lab to verify the validity of the PVI control strategy.



Figure 2.12: Experimental waveforms of System I at full load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.13: Experimental waveforms of System I at half load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.14: Experimental waveforms of System I at light load: (a) without PVI control strategy, (b) with PVI control strategy.

2.3.1 System I — a Buck Converter with a *LC* Input Filter

System I is a 100 W cascaded system consisting of a *LC* input filter and a 48 V - 24 V digital-controlled Buck converter. Its topology, parameters and control algorithm are presented in Fig. 2.7, Table 2.2 and Fig. 2.9, respectively.

The experimental results of System I with/without the PVI control strategy at different loads are given in Figs. 2.12 to 2.14, where the waveforms of v_{bus} and v_o are their ac components to clearly show the oscillation. As seen in Figs. 2.12(a) and 2.13(a), System I is unstable at full load and half load when the PVI control strategy is absent, and the oscillating frequencies of the bus voltage and output voltage are both about 685 Hz, which concurs with the conclusion at Fig. 2.8. As shown in Fig. 2.12(b) and Fig. 2.13(b), when the PVI control strategy is incorporated, System I becomes stable at full load and half load, and no oscillation occurs in v_{bus} and v_o . Fig. 2.14(a) shows the experimental waveforms of i_o , v_{bus} and v_o at 35% full-load without the PVI control strategy. It shows that System I is stable, which is consistent with Fig. 2.8. Besides, as shown in Fig. 2.14(b), System I is also stable at 35% full-load when the PVI control strategy is utilised.

In order to further show the zoomed waveforms of System I with/without the PVI control strategy at full load, its simulation results are given at Figs. 2.15(a) and (b). According to the zoomed waveforms in Fig. 2.15, the PVI control strategy indeed solves the instability problem of System I.

Based on Fig. 2.15(a), the harmonic distribution histogram of System I without PVI control strategy at full load is presented at Fig. 2.16. It is shown that, without the PVI control strategy, there is an obvious harmonic component of i_o , v_{bus} and v_o at 700 Hz, which is around the oscillation frequency. According to Fig. 2.16, the oscillation deteriorates the quality of i_o , v_{bus} and v_o : with the total harmonic distortion (THD) of the unstable i_o , v_{bus} and v_o at 9.488%, 13.969% and 9.488%, respectively.

Based on Fig. 2.15(b), the harmonic distribution histogram of System I with PVI control strategy at full load is presented at Fig. 2.17. It can be seen that there is no obvious harmonic component of i_o , v_{bus} and v_o with the PVI control strategy. Therefore, the PVI control strategy indeed solves the instability problem of System I at full load. In addition, compared to Fig. 2.16, the PVI control strategy also improves the quality of i_o , v_{bus} and v_o : with the THD of stable i_o , v_{bus} and v_o reduced to 0.013%, 0.001% and 0.013%, respectively.

Fig. 2.18(a) shows the dynamic performance of System I when its load steps between 10% and 100% full-load at the rated input voltage (48 V) indicating that the dynamic performances of System I with PVI control strategy are good. Fig. 2.18(b) shows the dynamic performance of System I when its input voltage steps between 80% (38.4 V) and 120% (57.6 V) rated voltage at full load. It can be seen that the dynamic performance of System I with PVI control strategy is also good.

2.3.2 System II — a Boost Converter with a *LC* Input Filter

A 200W cascaded system with a *LC* input filter and a 24 V - 48 V digital-control-based Boost converter is fabricated. Its topology, parameters and control algorithm have already been described in Fig. 2.10, Table 2.3 and Fig. 2.9, respectively.

The experimental results of System II with/without the PVI control strategy at different loads are given in Figs. 2.19 to 2.21, where the waveforms of v_{bus} and v_o are their ac components to clearly show the oscillation. As seen in Figs. 2.19(a) and 2.20(a), System II is unstable at full load and half load when the PVI control strategy is absent, and the oscillating frequencies of the bus voltage and output voltage are both approximately 520 Hz, which concurs with the conclusion at Fig. 2.11. As shown in Fig. 2.19(b) and Fig.



Figure 2.15: Simulation waveforms of System I at full load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.16: Harmonic distribution histogram of System I without PVI control strategy at full load: (a) i_o , (b) v_{bus} , (c) v_o .


Figure 2.17: Harmonic distribution histogram of System I with PVI control strategy at full load: (a) i_o , (b) v_{bus} , (c) v_o .



Figure 2.18: Dynamic experimental waveforms of System I with PVI control strategy: (a) load changing between 10% load and 100% load, (b) input voltage changing between 80% rated voltage and 120% rated voltage.



Figure 2.19: Experimental waveforms of System II at full load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.20: Experimental waveforms of System II at half load: (a) without PVI control strategy, (b) with PVI control strategy.

2.20(b), when the PVI control strategy is incorporated, System II becomes stable at full load and half load, and no oscillation occurs in v_{bus} and v_o . Fig. 2.21(a) shows the experimental waveforms of i_o , v_{bus} and v_o at 20% full-load without the PVI control strategy. It is shown here that System II is stable, which is consistent with Fig. 2.11, and Fig. 2.21(b) indicates that System II is stable at 20% full-load when the PVI control strategy is utilised.

Fig. 2.22(a) shows the dynamic waveforms of System II when its load steps between 10% and 100% full-load at the rated input voltage (24 V). The dynamic performance of System II with PVI control strategy is good. Fig. 2.22(b) shows the dynamic performance of System II with the input voltage steps between 80% (19.2 V) and 120% (28.8 V) rated voltage at full load. It is shown that the dynamic performance of System II with PVI control strategy is good.

The above two examples indicate that the PVI control strategy is effective and feasible in practice.



Figure 2.21: Experimental waveforms of System II at light load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.22: Dynamic experimental waveforms of System II with PVI control strategy: (a) load changing between 10% load and 100% load, (b) input voltage changing between 80% rated voltage and 120% rated voltage.

2.4 Simulation Verification

2.4.1 Simulation Verification of System I with the Consideration of the Load Input Capacitor

The input capacitor is usually a necessary component of DC/DC converters in practice. Therefore, a simulink model adding a 110 μF input capacitor to the Buck converter of Fig. 2.7 has been built in Matlab to verify the accuracy of the PVI control strategy.

The simulation results of System I with consideration of load input capacitor at different loads are given in Figs. 2.23 to 2.25. As with the above experimental results, the simulation waveforms of v_{bus} and v_o are also their ac components and clearly show the oscillation. As seen in Figs. 2.23(a) and 2.24(a), System I is unstable at full load and half load when the PVI control strategy is absent. As shown in Fig. 2.23(b) and Fig. 2.24(b), when the PVI control strategy is incorporated, System I becomes stable at full load and half load, and no oscillation occurs in v_{bus} and v_o . Fig. 2.25(a) shows the simulation waveforms of i_o , v_{bus} and v_o at 35% full-load without the PVI control strategy. It shows that System I is stable. As shown in Fig. 2.25(b), System I is also stable at 35% full-load when the PVI control strategy is utilised. It is worth pointing out that thanks to the added input capacitor, the oscillation waveforms in Figs. 2.23(a) and 2.24(a) are smaller than those in Figs. 2.12(a) and 2.13(a).

Fig. 2.26(a) shows the load dynamic performance of System I with the PVI control strategy and the added load input capacitor. It shows that when the load steps between 10% and 100% full-load at the rated input voltage (48 V), the dynamic performance of System I is good. Fig. 2.26(b) shows the input voltage dynamic performance of System I with the PVI control strategy and the added load input capacitor. It also shows that with the input voltage steps between 80% (38.4 V) and 120% (57.6 V) rated voltage at full load, the dynamic performance of System I is also good.

2.4.2 Simulation Verification of System II with the Consideration of the Load Input Capacitor

Similarly, a simulink model adding a 110 μF input capacitor to the Boost converter in Fig. 2.10 has also been built in Matlab to verify the accuracy of the PVI control strategy.



Figure 2.23: Simulation waveforms of System I with the added load input capacitor at full load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.24: Simulation waveforms of System I with the added load input capacitor at half load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.25: Simulation waveforms of System I with the added load input capacitor at light load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.26: Dynamic simulation waveforms of System I with PVI control strategy and the added load input capacitor: (a) load changing between 10% load and 100% load, (b) input voltage changing between 80% rated voltage and 120% rated voltage.



Figure 2.27: Simulation waveforms of System II with the added load input capacitor at full load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.28: Simulation waveforms of System II with the added load input capacitor at half load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.29: Simulation waveforms of System II with the added load input capacitor at light load: (a) without PVI control strategy, (b) with PVI control strategy.



Figure 2.30: Dynamic simulation waveforms of System II with PVI control strategy and the added load input capacitor: (a) load changing between 10% load and 100% load, (b) input voltage changing between 80% rated voltage and 120% rated voltage.

The simulation results of System II with consideration of load input capacitor at different loads are given in Figs. 2.27 to 2.29. Similar to the above experimental results, the simulation waveforms of v_{bus} and v_o are also their ac components and clearly show the oscillation. As seen in Figs. 2.27(a) and 2.28(a), System II is unstable at full load and half load when the PVI control strategy is absent. As shown in Fig. 2.27(b) and Fig. 2.28(b), when the PVI control strategy is incorporated, System II becomes stable at full load and half load, and no oscillation occurs in v_{bus} and v_o . Fig. 2.29(a) shows the simulation waveforms of i_o , v_{bus} and v_o at 20% full-load without the PVI control strategy. It shows that System II is stable. In addition, as shown in Fig. 2.29(b), System II is also stable at 20% full-load when the PVI control strategy is utilised. It is worth pointing out that, thanks to the added input capacitor, the oscillation waveforms in Figs. 2.27(a) and 2.28(a) are smaller than those in Figs. 2.19(a) and 2.20(a).

Fig. 2.30(a) shows the load dynamic waveforms of System II with the PVI control strategy and the added load input capacitor. It shows that with load steps between 10% and 100% full-load at the rated input voltage (24 V), the dynamic performance of System II is good. Fig. 2.30(b) shows the input voltage dynamic performance of System II with the PVI control strategy and the added load input capacitor. It also shows that with the input voltage steps between 80% (19.2 V) and 120% (28.8 V) rated voltage at full load, the dynamic performance of System II is also good.

The above simulation results demonstrate that the PVI control strategy is effective and feasible when the load converter contains input capacitors.

2.5 Summary

In order to solve the instability problem of the cascaded system, a PVI control strategy is proposed in this chapter. This research shows that the PVI control strategy modifies the input impedance of the load converter only in a small range of frequencies. Therefore, it not only stabilises the whole system, but also ensures a good dynamic performance of the load converter. Furthermore, the proposed method can be realised by a simple digital control algorithm, thus this approach is feasible in practice. Finally, two example systems are fabricated to validate the correctness of the PVI control strategy.

Chapter 3

Series-Virtual-Impedance (SVI) Control for Load Converters

In Chapter 2, a PVI control strategy is proposed to stabilise the cascaded system with the minimised load compromise. Since the PVI control strategy can add a virtual impedance in parallel with the load converter to realise its function, it can be assumed that a corresponding virtual impedance also can be added in the series with the load converter to realise the same function. Therefore, a series-virtual-impedance (SVI) control strategy is proposed in this chapter, which can also be treated as a development of, and complement to the PVI control strategy. The concept, derivation and design examples of the SVI control strategy are discussed carefully in this chapter. Similar to the PVI control strategy, the SVI control strategy is also verified by two experimental/simulation cascaded systems.

The other parts of this chapter are arranged as follows: in Section 3.1, the basic idea of load stabilisation methods, the concept and realisation of the SVI control strategy are studied. Following that, two design examples of the SVI control strategy are presented in Section 3.2; and two experimental/simulation cascaded systems are fabricated to verify the effectiveness of the SVI control strategy in Sections 3.3 and 3.4, respectively. Finally, Section 3.5 concludes the chapter.

3.1 SVI Control Strategy

3.1.1 Basic Idea of Load Shaping Methods

As discussed in Chapter 2, when a designer wants to keep the cascaded system stable while keeping a better performance of load converter, the best load stabilisation method is to change Z_{iL} only in the vicinity of the intersection frequencies of $|Z_{oS}|$ and $|Z_{iL}|$. Here, Z_{oS} is the source output impedance, and Z_{iL} is the load input impedance.

As shown in Fig. 3.1, there are two better ways to regulate Z_{iL} :

(a) Make a total separation of $|Z_{oS}|$ and $|Z_{iL}|$ by increasing $|Z_{iL}|$ only during $[f_1, f_2]$ (See Fig. 3.1(a));

(b) Ensure $|\varphi(Z_{oS}) - \varphi(Z_{iL})| < 180^{\circ}$ at f_1 and f_2 by increasing $\varphi(Z_{iL})$ only during $[f_1, f_2]$ (See Fig. 3.1(b)). Here, f_1 and f_2 are the intersection frequencies of $|Z_{oS}|$ and $|Z_{iL}|$.

Note, in Fig. 3.1, Z_{iLS} is the input impedance of the load converter with the latter proposed SVI control strategy.

3.1.2 Expression of the Series-Virtual-Impedance $Z_{SVI}(s)$

The load shaping idea in Section 3.1.1 can be realised by adding a virtual impedance $Z_{SVI}(s)$ in series with the load input port. As shown in Fig. 3.2, the shaped load input impedance $Z_{iLS}(s)$ can be expressed as

$$Z_{iLS} = Z_{iL} + Z_{SVI} \tag{3.1}$$

where Z_{iL} is the original input impedance of the load converter and if $f < f_{CL}$ its expression is $-V_{bus}^2/P_o$. Here, V_{bus} and P_o are the input voltage and output power of the load converter, respectively.

a) As illustrated in Fig. 3.1(a), if a total separation between $|Z_{oS}|$ and $|Z_{iLS}|$ is needed and the minimised change of Z_{iL} is also required, Z_{iLS} should satisfy:

$$If \{ f \in [f_1, f_2] \} \rightarrow |Z_{iLS}| > Z_{oSP}$$

$$(3.2)$$

$$If \{ f \notin [f_1, f_2] \} \rightarrow Z_{iLS} = Z_{iL}$$

$$(3.3)$$



Figure 3.1: Basic idea of load shaping methods: (a) only increasing $|Z_{iL}|$ at $[f_1, f_2]$ to make a total separation between $|Z_{oS}|$ and $|Z_{iL}|$; (b) only increasing $\varphi(Z_{iL})$ at $[f_1, f_2]$ to ensure $|\varphi(Z_{oS}) - \varphi(Z_{iL})| < 180^0$ at $[f_1, f_2]$.



Figure 3.2: A cascaded system with series virtual impedance.

where Z_{oSP} is the peak value of $|Z_{oS}|$.

According to (3.2) and (3.3), Z_{iLS} could be further expressed as

$$Z_{iLS} = \begin{cases} -|Z_{iL1}| & f \in [f_1, f_2] \\ Z_{iL} & f \notin [f_1, f_2] \end{cases}$$
(3.4)

where $|Z_{iL1}| > Z_{oSP}$ is the amplitude of Z_{iLS} during $[f_1, f_2]$.

According to (3.1) and (3.4), in order to ensure $|Z_{iL}|$ successfully becomes $|Z_{iLS}|$ as shown in Fig. 3.1(a), Z_{SVI} can be selected as

$$Z_{SVI} = Z_{iLS} - Z_{iL} = \begin{cases} -|Z_{iL1}| - (-V_{bus}^2/P_o) & f \in [f_1, f_2] \\ Z_{iL} - Z_{iL} & f \notin [f_1, f_2] \end{cases}$$
$$= \begin{cases} (V_{bus}^2/P_o) - |Z_{iL1}| & f \in [f_1, f_2] \\ 0 & f \notin [f_1, f_2] \end{cases}$$
(3.5)

b) Similarly, according to Fig. 3.1(b), if $|\varphi(Z_{oS}) - \varphi(Z_{iLS})| < 180^{\circ}$ is to be ensured within $[f_1, f_2]$ and the minimised change of Z_{iL} is also required, Z_{iLS} should satisfy:

$$If \{ f \in [f_1, f_2] \} \rightarrow |\theta| < 90^0$$
 (3.6)

$$If \{ f \notin [f_1, f_2] \} \rightarrow \varphi(Z_{iLS}) = \varphi(Z_{iL})$$

$$(3.7)$$

where θ is the phase of Z_{iLS} during $[f_1, f_2]$.

According to (3.6) and (3.7), Z_{iLS} could be further expressed as

$$Z_{iLS} = \begin{cases} \left(V_{bus}^2 / P_o \right) e^{j\theta} & f \in [f_1, f_2] \\ Z_{iL} & f \notin [f_1, f_2] \end{cases}$$
(3.8)

According to (3.1) and (3.8), in order to ensure $|Z_{iL}|$ successfully becomes $|Z_{iLS}|$ as shown in Fig. 3.1(b), Z_{SVI} can be selected as

$$Z_{SVI} = Z_{iLS} - Z_{iL} = \begin{cases} \left(V_{bus}^2/P_o\right) e^{j\theta} - \left(-V_{bus}^2/P_o\right) & f \in [f_1, f_2] \\ Z_{iL} - Z_{iL} & f \notin [f_1, f_2] \end{cases}$$

$$= \begin{cases} \left(V_{bus}^2/P_o\right) \cdot \left(1 + e^{j\theta}\right) & f \in [f_1, f_2] \\ 0 & f \notin [f_1, f_2] \end{cases}$$
(3.9)

\hat{v}_{bus}	Perturbation of the bus voltage
î,	Perturbation of the load current
î _{bus}	Perturbation of the bus current
â	Perturbation of the duty cycle
v _o	Perturbation of the output voltage
\hat{v}_{or}	Perturbation of the output voltage reference
$G_{v}(s)$	Transfer function of the voltage regulator
$G_M(s)$	Transfer function of the modulator
$Z_{iOL}(s)$	Open-loop input impedance
$G_{vd}(s)$	Control to output voltage transfer function
$G_{id}(s)$	Control to input current transfer function
$Z_{oOL}(s)$	Open-loop output impedance
$H_s(s)$	Sampling coefficient of the output voltage
$G_{iOL}(s)$	Open-loop load to input current transfer function
$G_{vOL}(s)$	Open-loop input to output voltage transfer function

Table 3.1: Definitions of variables and transfer functions in load converter

In summary, if the Z_{SVI} , which is expressed by (3.5), is added in the series with the load input impedance, the total separation between $|Z_{oS}|$ and $|Z_{iLS}|$ shown in Fig. 3.1(a) can be achieved. If the Z_{SVI} , which expressed by (3.9), is added in the series with the load input impedance, $|\varphi(Z_{oS}) - \varphi(Z_{iLS})| < 180^{\circ}$ within $[f_1, f_2]$ as shown in Fig. 3.1(b) can be ensured.

3.1.3 Derivation of the SVI Control Strategy

According to Section 3.1.1, with Z_{SVI} , the cascaded system can be stabilised by increasing $|Z_{iL}|$ or $\varphi(Z_{iL})$ in a very small frequency range to minimise the load compromise. Hence, in this section, a SVI control strategy for the load converter is proposed to realise Z_{SVI} .

First of all, the small signal block diagram of the original load converter is presented in Fig. 3.3. v_{bus} and i_{bus} are the input voltage and current of the load converter, respectively. v_o and i_o are the output voltage and current of the load converter, respectively. d is the duty cycle of the load converter. $Z_{iOL}(s)$, $G_{iOL}(s)$, $G_{id}(s)$, $G_{vOL}(s)$, $Z_{oOL}(s)$ and $G_{vd}(s)$ are the six basic open-loop transfer functions of the load converter. In addition, for the original load converter, a voltage closed-loop is utilised, where $H_s(s)$ and $G_v(s)$ are the sampling coefficient and voltage regulator transfer function of the output voltage, respectively; and $G_M(s)$ is the transfer function of the modulator. All the variables and transfer functions of Fig. 3.3 are described in Table 3.1.

 $Z_{iL}(s)$ is the input impedance of the original load converter. Therefore, for the original



Figure 3.3: Small signal block diagram of the original load converter.



Figure 3.4: Small signal block diagram of the load converter with the paralleled virtual impedance.



Figure 3.5: Derivation of the SVI control strategy.

load converter, its v_{bus} and i_{bus} have the following relationship:

$$v_{bus} = Z_{iL}(s)i_{bus} \tag{3.10}$$

If Z_{SVI} is added in the series with the input port of the load converter, the relationship between v_{bus} and i_{bus} is changed to:

$$v_{bus} - Z_{SVI}(s)i_{bus} = Z_{iL}(s)i_{bus}$$

$$(3.11)$$

As indicated by (3.11), the small signal block diagram of the load converter can be changed to Fig. 3.4.

Fig. 3.4 illustrates that $-Z_{SVI}$ is introduced to the control block between the v_{bus} and i_{bus} directly. From the small signal block diagram point of view, Fig. 3.4 achieves the purpose of adding Z_{SVI} to the series with the input port of the load converter. However, from the control point of view, this method cannot be realised directly. It is because that any converter is only controlled by its duty cycle. In order to address this issue, the output of $-Z_{SVI}$ is moved to the output voltage reference, with equivalent adjustments to the transfer function to $G_{SVI}(s)$, as shown with the Fig. 3.5.

Fig. 3.5 shows that when $G_{SVI}(s)$ is required to realise the series virtual impedance Z_{SVI} , the following relationship can be obtained:

$$\hat{i}_{bus} \cdot [-Z_{SVI}(s)] \cdot \frac{1}{Z_{iL}(s)} = \hat{i}_{bus} \cdot G_{SVI}(s) \cdot G_M(s) \cdot G_{id}(s) \cdot \frac{1}{1 + T_v(s)}$$
(3.12)

where $T_{\nu}(s) = H_s(s)G_{\nu}(s)G_{\nu d}(s)$ is the voltage loop gain of the load converter.

According to (3.12), the expression of $G_{SVI}(s)$ can be derived as:

$$G_{SVI}(s) = -\frac{Z_{SVI}(s)}{Z_{iL}(s)} \cdot \frac{1 + T_v(s)}{G_M(s) \cdot G_{id}(s)}$$

= $Z_{SVI}(s) \cdot \frac{[1 + T_v(s)] \cdot P_o}{G_{id}(s) \cdot G_M(s) \cdot V_{bus}^2}$ (3.13)

Up to this point, Fig. 3.3 ~ Fig. 3.5 represent the whole derivation process of the SVI control strategy, and $G_{SVI}(s)$ is the input-impedance-regulator of the SVI control strategy.

3.1.4 Realisation of the SVI Control Strategy

According to Section 3.1.3, $G_{SVI}(s)$ is the key point of the SVI control strategy. Therefore, methods of realising $G_{SVI}(s)$ are discussed in this section.

According to (3.5), (3.9) and Fig. 3.5, the expression of $G_{SVI}(s)$ can be derived in the following ways:

a) To achieve a total separation between $|Z_{oS}|$ and $|Z_{iLS}|$, $G_{SVI1}(s)$ is

$$G_{SVI1}(s) = \begin{cases} \left[\left(V_{bus}^2 / P_o \right) - |Z_{iL1}| \right] \cdot \frac{[1 + T_v(s)] \cdot P_o}{G_{id}(s) \cdot G_M(s) \cdot V_{bus}^2} & f \in [f_1, f_2] \\ 0 & f \notin [f_1, f_2] \end{cases}$$
(3.14)

b) To ensure $|\varphi(Z_{oS}) - \varphi(Z_{iLS})| < 180^{\circ}$ at f_1 and f_2 , $G_{SVI2}(s)$ is

$$G_{SVI2}(s) = \begin{cases} \left(V_{bus}^{\prime 2}/P_{o}\right) \cdot \left(1 + e^{j\theta}\right) \cdot \frac{[1 + T_{v}(s)] \cdot P_{o}}{G_{id}(s) \cdot G_{M}(s) \cdot V_{bus}^{2}} & f \in [f_{1}, f_{2}] \\ 0 & f \notin [f_{1}, f_{2}] \end{cases}$$
(3.15)

According to (3.14) and (3.15), $G_{SVI}(s)$ is a frequency-based piecewise transfer function, where $|G_{SVI}(s)| > 0$ within $[f_1, f_2]$ and $|G_{PVI}(s)| = 0$ outside $[f_1, f_2]$ as shown in Fig. 3.6. Hence, a band-pass filter $G_{BPF}(s)$ can be utilised to help achieve $G_{SVI}(s)$, i.e., (3.14) and (3.15) can be re-expressed as (3.16) and (3.17), respectively.

$$G_{SVI1}(s) = \left[\left(V_{bus}^{2} / P_{o} \right) - |Z_{iL1}| \right] \cdot \frac{[1 + T_{v}(s)] \cdot P_{o}}{G_{id}(s) \cdot G_{M}(s) \cdot V_{bus}^{2}} \cdot G_{BPF}(s)$$
(3.16)

$$G_{SVI2}(s) = \left(V_{bus}^{\prime 2}/P_o\right) \cdot \left(1 + e^{j\theta}\right) \cdot \frac{\left[1 + T_v(s)\right] \cdot P_o}{G_{id}(s) \cdot G_M(s) \cdot V_{bus}^2} \cdot G_{BPF}(s)$$
(3.17)

where $G_{BPF}(s)$ adopts a typical second order band-pass filter, which is composed of a second order high-pass filter and a second order low-pass filter, given as

$$G_{BPF}(s) = \frac{s^2}{s^2 + (2\pi f_1/Q_H)s + (2\pi f_1)^2} \cdot \frac{(2\pi f_2)^2}{s^2 + (2\pi f_2/Q_L)s + (2\pi f_2)^2}$$
(3.18)

where Q_H and Q_L are quality factors of the high-pass and low-pass filter, respectively.

According to (3.16) and (3.17), $G_{SVI}(s)$ can be easily realised by digital control chips, such as a digital signal processor (DSP), field programmable gate array (FPGA) and micro-



Figure 3.6: Characteristic of $|G_{SVI}(s)|$.



Figure 3.7: The System I.

controller unit (MCU).

3.2 Design Example

As discussed in Chapter 1, the worst instability phenomena are most likely to occur when the source converter is a *LC* input filter. As Buck and Boost converters are two typical DC/DC converters, a 100 W cascaded system, which consists of a *LC* input filter and a 48 V - 12 V Buck converter, and a 200 W cascaded system, which includes a *LC* input filter and a 24 V - 48 V Boost converter, are designed in this section.

3.2.1 System I — a Buck Converter with a *LC* Input Filter

A 100 W cascaded system with a *LC* input filter (source stage) and a 48 V - 12 V Buck converter (load stage) is designed, which is shown in Fig. 3.7. The Buck converter is operated at 100 kHz. The cut-off frequency and phase margin of the voltage close loop of the Buck converter are set to 20 kHz and 45° , respectively. The corresponding system parameters are summarised in Table 3.2.

The output impedance Z_{oS} of the *LC* input filter is given in (3.19), where R_{Lf} is the parasitic resistor of L_f and the measured value is 0.1 Ω . According to the small-signal



Table 3.2: Parameters of the System I

Figure 3.8: Bode plots of Z_{oS} and Z_{iL} of the System I.

circuit model of the Buck converter (Cuk, 1976), the input impedance Z_{iL} of the Buck converter can be expressed as (3.20).

$$Z_{oS}(s) = \frac{sL_f + R_{Lf}}{s^2 L_f C_f + sR_{Lf} C_f + 1}$$
(3.19)

$$Z_{iL}(s) = \left[\frac{1}{1+T_{\nu}(s)} \cdot \frac{sC_{f1}D^2 + \frac{D^2}{R_{Ld}}}{s^2 L_{f1}C_{f1} + s\frac{L_{f1}}{R_{Ld}} + 1} - \frac{T_{\nu}(s)}{1+T_{\nu}(s)} \cdot \frac{P_o}{V_{bus}^2}\right]^{-1}$$
(3.20)

where P_o and D are the output power and the duty cycle of the Buck converter, respectively.

Fig. 3.8 shows the Bode plots of Z_{oS} and Z_{iL} in the example system: when the load is lower than 35% full-load, the cascaded system is stable; otherwise, the cascaded system is unstable and the oscillation occurs at approximately 685 Hz theoretically (Wildrick, 1993).

To solve the instability problem conveniently, Z_{iL} is designed as a positive resistor within $[f_1, f_2]$ to ensure $|\varphi(Z_{oS}) - \varphi(Z_{iLS})| < 180^0$, i.e., θ is set to 0 at (3.17). By the



Figure 3.9: Software flowchart of the SVI control strategy.

small signal model of the Buck converter (Cuk, 1976) and (3.17), the expression of $G_{SVI}(s)$ can be derived as

$$G_{SVI}(s) = G_{SVI1}(s) \cdot G_{BPF}(s) \tag{3.21}$$

where

$$G_{SVI1}(s) \approx \left(1 - \frac{1 + sR_{Ld}C_{f1}}{s^2 L_{f1}C_{f1} + s\left(R_{Ld}C_{f1} + L_{f1}/R_{Ld}\right) + 2}\right) \cdot \frac{2V_M R_{Ld}}{V_{bus}D}$$
(3.22)

here, in $G_{BPF}(s)$, $f_1 = 685 Hz$ and $f_2 = 780 Hz$; $Q_H = 0.707$ and $Q_L = 0.707$.

The Bode plot of the modified input impedance, Z_{iLS} , is shown with double dotteddashed line in Fig. 3.8, with $|\varphi(Z_{oS}) - \varphi(Z_{iLS})| < 180^{\circ}$ at f_1 and f_2 . Therefore, the cascaded system will be stable and the design of $G_{SVI}(s)$ is appropriate.

As the SVI control strategy is realised by a DSP TMS320F28335 in practice, its software flowchart is presented in Fig. 3.9. The input current i_{bus} and the output voltage v_o of the load converter are sampled to provide the necessary information for the SVI control strategy and output voltage regulation. The algorithm obtained the modulation signal by adding the output of $G_{SVI}(s)$ to the output of $G_v(s)$. It then generated the PWM signal by



Figure 3.10: The System II.

Table 3.3: Parameters of the System II

Parameter	L_{f}	C_{f}	L_{f1}	C_{f1}	R_{Ld}
Value	250 μH	330 μF	720 μH	450 μF	11.52 Ω

PWM module.

3.2.2 System II — a Boost Converter with a *LC* Input Filter

Fig. 3.10 shows a 200 W cascaded system consisting of a *LC* input filter (source stage) and a 24 V - 48 V Boost converter (load stage) operated at 100 kHz. Since the Boost converter is a non-minimum phase system, a proportional–integral (PI) controller is employed as the compensator and the cut-off frequency and phase margin of the voltage closed loop of the Boost converter are set to 2.5 kHz and 32° , respectively. The corresponding system parameters are summarised in Table 3.3.

The output impedance Z_{oS} of the *LC* input filter is given in (3.19), and is threefore not repeated here. According to the small-signal circuit model of the Boost converter (Cuk, 1976), the input impedance Z_{iL} of the Boost converter can be expressed as (3.23).

$$Z_{iL}(s) = \left[\frac{1}{1+T_{\nu}(s)} \cdot \frac{sC_{f1}R_{Ld} + 1}{s^2L_{f1}C_{f1}R_{Ld} + sL_{f1} + R_{Ld}(1-D)^2} - \frac{T_{\nu}(s)}{1+T_{\nu}(s)} \cdot \frac{P_o}{V_{bus}^2}\right]^{-1}$$
(3.23)

where D and P_o are the duty cycle and output power of the Boost converter, respectively.

Fig. 3.11 shows the Bode plots of Z_{oS} and Z_{iL} in the example system: when the load is lower than 20% full-load, the cascaded system is stable; otherwise, the cascaded system is



Figure 3.11: Bode plots of Z_{oS} and Z_{iL} of the System II.

unstable and the oscillation occurs at approximately 520 Hz theoretically (Wildrick, 1993).

In order to conveniently solve the instability problem, Z_{iL} is designed as a positive resistor within $[f_1, f_2]$ to ensure $|\varphi(Z_{oS}) - \varphi(Z_{iLS})| < 180^0$, i.e., θ is set to 0 at (3.17). According to the small signal model of Boost converter (Cuk, 1976) and (3.17), the expression of $G_{SVI}(s)$ can then be derived as

$$G_{SVI}(s) \approx G_{SVI1}(s) \cdot G_{BPF}(s) \tag{3.24}$$

where

$$G_{SVI1}(s) = \frac{2V_M[s^2 L_{f1} C_{f1} R_{Ld} + s L_{f1} + R_{Ld} (1-D)^2]}{v_o(s C_{f1} R_{Ld} + 2)}$$
(3.25)

Also, for $G_{BPF}(s)$, $f_1 = 520$ Hz and $f_2 = 645$ Hz, $Q_H = Q_L = 0.707$.

The Bode plot of the modified input impedance, Z_{iLS} , is represented by a double dotteddashed line in Fig. 3.11, with $|\varphi(Z_{oS}) - \varphi(Z_{iLS})| < 180^{\circ}$ at f_1 and f_2 . Therefore, the improved cascaded system is stable and the design of $G_{SVI}(s)$ is appropriate.

In practice, the Boost converter is also controlled by DSP TMS320F28335, and its SVI control strategy can be realised by the software flowchart in Fig. 3.9. Since the flowchart



Figure 3.12: Experimental waveforms of the System I at full load: (a) without SVI control strategy, (b) with SVI control strategy.

has already been explained in Section 3.2.1, it is not repeated here.

3.3 Experimental Verification

According to the design examples, two experimental systems have been fabricated and tested in the lab to verify the validity of the SVI control strategy.

3.3.1 System I — a Buck Converter with a *LC* Input Filter

System I is a 100 W cascaded system consisting of a *LC* input filter and a 48 V - 12 V digital-controlled Buck converter. Its topology, parameters and control algorithm are presented in Fig. 3.7, Table 3.2 and Fig. 3.9, respectively.

The experimental results of System I with/without the SVI control strategy at different loads are given in Figs. 3.12 to 3.14 respectively, where the waveforms of v_{bus} and v_o are their ac components to clearly show the oscillation. As seen in Figs. 3.12(a) and 3.13(a), System I is unstable at full load and half load when the SVI control strategy is absent, and the oscillating frequencies of the bus voltage and output voltage are both about 685 Hz, which concurs with the conclusion at Fig. 3.8. As shown in Fig. 3.12(b) and Fig. 3.13(b),



Figure 3.13: Experimental waveforms of the System I at half load: (a) without SVI control strategy, (b) with SVI control strategy.

when the SVI control strategy is incorporated, System I becomes stable at full load and half load, and no oscillation occurs in v_{bus} and v_o . Fig. 3.14(a) shows the experimental waveforms of i_o , v_{bus} and v_o at 35% full-load without the SVI control strategy. This shows that System I is stable, which is consistent with Fig. 3.8. As shown in Fig. 3.14(b), System I is also stable at 35% full-load when SVI control strategy is utilised.

Fig. 3.15(a) shows the dynamic performance of System I when its load steps between 10% and 100% full-load at the rated input voltage (48 V). It shows that the dynamic performance of System I with SVI control strategy is good. Fig. 3.15(b) shows the dynamic performance of System I when its input voltage steps between 80% (38.4 V) and 120% (57.6 V) rated voltage at full load; and shows that the dynamic performance of System I with SVI control strategy is also good.

3.3.2 System II — a Boost Converter with a *LC* Input Filter

A 200 W cascaded system with a *LC* input filter and a 24 V - 48 V digital-control-based Boost converter is fabricated. Its topology, parameters and control algorithm are described in Fig. 3.10, Table 3.3 and Fig. 3.9, respectively.

The experimental results of System II with/without the SVI control strategy at different



Figure 3.14: Experimental waveforms of the System I at light load: (a) without SVI control strategy, (b) with SVI control strategy.



Figure 3.15: Dynamic experimental waveforms of the System I with SVI control strategy: (a) load changing between 10% load and 100% load, (b) input voltage changing between 80% rated voltage and 120% rated voltage.



Figure 3.16: Experimental waveforms of the System II at full load: (a) without SVI control strategy, (b) with SVI control strategy.



Figure 3.17: Experimental waveforms of the System II at half load: (a) without SVI control strategy, (b) with SVI control strategy.



Figure 3.18: Experimental waveforms of the System II at light load: (a) without SVI control strategy, (b) with SVI control strategy.

loads are given in Figs. 3.16 to 3.18, where the waveforms of v_{bus} and v_o are their ac components to clearly show the oscillation. As seen in Figs. 3.16(a) and 3.17(a), System II is unstable at full load and half load when the SVI control strategy is absent, and the oscillating frequencies of the bus voltage and output voltage are both approximately 520 Hz, which concurs with the conclusion in Fig. 3.11. As shown in Fig. 3.16(b) and Fig. 3.17(b), when the SVI control strategy is incorporated, System II becomes stable at full load and half load, and no oscillation occurs in v_{bus} and v_o . Fig. 3.18(a) shows the experimental waveforms of i_o , v_{bus} and v_o at 20% full-load without the SVI control strategy. System II is stable, which is consistent with Fig. 3.11. In addition, as shown in Fig. 3.18(b), System II is stable at 20% full-load when the SVI control strategy is utilised.

Fig. 3.19(a) shows the dynamic waveforms of System II when its load steps between 10% and 100% full-load at the rated input voltage (24 V). The dynamic performance of System II with SVI control strategy is good. Fig. 3.19(b) shows the dynamic performance of System II with the input voltage steps between 80% (19.2 V) and 120% (28.8 V) rated voltage at full load. It is also indicated that, the dynamic performance of System II with SVI control strategy is good.

The above two examples indicate that the SVI control strategy is effective and feasible



Figure 3.19: Dynamic experimental waveforms of the System II with SVI control strategy: (a) load changing between 10% load and 100% load, (b) input voltage changing between 80% rated voltage and 120% rated voltage.

in practice.

3.4 Simulation Verification

3.4.1 Simulation Verification of System I with the Consideration of the Load Input Capacitor

The input capacitor is usually a necessary component of DC/DC converters in practice. Therefore, a simulink model which adds a 110 μF input capacitor to the Buck converter of Fig. 3.7 has been built in Matlab to verify the accuracy of the SVI control strategy.

The simulation results of System I with consideration for the load input capacitor at different loads are given in Figs. 3.20 to 3.22. Similar to the above experimental results, the simulation waveforms of v_{bus} and v_o are also their ac components to clearly show the oscillation. As seen in Figs. 3.20(a) and 3.21(a), System I is unstable at full load and half load when the SVI control strategy is absent. As shown in Fig. 3.20(b) and Fig. 3.21(b), when the SVI control strategy is incorporated, System I becomes stable at full load and half load, and no oscillation occurs in v_{bus} and v_o . Fig. 3.22(a) shows the simulation waveforms



Figure 3.20: Simulation waveforms of the System I with the added load input capacitor at full load: (a) without SVI control strategy, (b) with SVI control strategy.



Figure 3.21: Simulation waveforms of the System I with the added load input capacitor at half load: (a) without SVI control strategy, (b) with SVI control strategy.



Figure 3.22: Simulation waveforms of the System I with the added load input capacitor at light load: (a) without SVI control strategy, (b) with SVI control strategy.
of i_o , v_{bus} and v_o at 35% full-load without the SVI control strategy. It shows that System I is stable. In addition, as shown in Fig. 3.22(b), System I is also stable at 35% full-load when the SVI control strategy is utilised. It is worth pointing out that, thanks to the added input capacitor, the oscillation waveforms in Figs. 3.20(a) and 3.21(a) are smaller than those in Figs. 3.12(a) and 3.13(a).

Fig. 3.23(a) shows the load dynamic performance of System I with the SVI control strategy and added load input capacitor. It shows that with the load steps between 10% and 100% full-load at the rated input voltage (48 V), the dynamic performance of System I is good. Fig. 3.23(b) shows the input voltage dynamic performance of System I with the SVI control strategy and the added load input capacitor. It also shows that with the input voltage steps between 80% (38.4 V) and 120% (57.6 V) rated voltage at full load, the dynamic performance of System I is also good.

3.4.2 Simulation Verification of System II with the Consideration of the Load Input Capacitor

Similar to Section 3.4.1, a simulink model adding a 110 μ *F* input capacitor to the Boost converter of Fig. 2.10 has been built in Matlab to verify the accuracy of the SVI control strategy.

The simulation results of System II with consideration for the load input capacitor at different loads are given in Figs. 3.24 to 3.26. Similar to the above experimental results, the simulation waveforms of v_{bus} and v_o are also their ac components to clearly show the oscillation. As seen from Figs. 3.24(a) and 3.25(a), System II is unstable at full load and half load when the SVI control strategy is absent. As illustrated by Fig. 3.24(b) and Fig. 3.25(b), when the SVI control strategy is incorporated, System II becomes stable at full load and half load, and no oscillation occurs in v_{bus} and v_o . Fig. 3.26(a) shows the simulation waveforms of i_o , v_{bus} and v_o at 20% full-load without the SVI control strategy. It shows that System II is stable. In addition, as shown in Fig. 3.26(b), System II is also stable at 20% full-load when the SVI control strategy is utilised. It is worth pointing out that, thanks to the added input capacitor, the oscillation waveforms in Figs. 3.24(a) and 3.25(a) are smaller than those in Figs. 3.16(a) and 3.17(a).

Fig. 3.27(a) shows the load dynamic waveforms of System II with the SVI control



Figure 3.23: Dynamic simulation waveforms of the System I with SVI control strategy and the added load input capacitor: (a) load changing between 10% load and 100% load, (b) input voltage changing between 80% rated voltage and 120% rated voltage.



Figure 3.24: Simulation waveforms of the System II with the added load input capacitor at full load: (a) without SVI control strategy, (b) with SVI control strategy.



Figure 3.25: Simulation waveforms of the System II with the added load input capacitor at half load: (a) without SVI control strategy, (b) with SVI control strategy.



Figure 3.26: Simulation waveforms of the System II with the added load input capacitor at light load: (a) without SVI control strategy, (b) with SVI control strategy.



Figure 3.27: Dynamic simulation waveforms of the System II with SVI control strategy and the added load input capacitor: (a) load changing between 10% load and 100% load, (b) input voltage changing between 80% rated voltage and 120% rated voltage.

strategy and the added load input capacitor. It shows that with the load steps between 10% and 100% full-load at the rated input voltage (24 V), the dynamic performance of System II is good. Fig. 3.27(b) shows the input voltage dynamic performance of System II with the SVI control strategy and the added load input capacitor. It also shows that with the input voltage steps between 80% (19.2 V) and 120% (28.8 V) rated voltage at full load, the dynamic performance of System II is also good.

The above simulation results indicate that the SVI control strategy is also effective and feasible when the load converter contains the input capacitor.

3.5 Summary

Based on the idea of the PVI control strategy, another load stabilisation method is proposed in this chapter: the SVI control strategy. The SVI control strategy adds a virtual impedance in the series with the load converter to modify the input impedance in a very small frequency range. As a result, it can not only stabilise the cascaded system, but also ensure a good dynamic performance of the load converter. Similar to the PVI control strategy, the proposed SVI control strategy can also be realised by a simple digital control algorithm. Thus, the SVI control strategy is feasible in practice as well. In addition, the SVI control strategy can be treated as a development of, and complement to the PVI control strategy. Finally, two example systems are fabricated to validate the accuracy of the SVI control strategy.

Chapter 4

Adaptive-PVI (APVI) Control for Load Converters

In Chapters 2 and 3, a family of load stabilisation methods, PVI and SVI control strategies, are proposed. Compared to the existing stabilisation methods, both PVI and SVI control strategies take into account the stability of the cascaded system and the dynamic performance of the load converter.

However, although the proposed PVI and SVI control strategies are more competitive than the other existing stabilisation methods, they require detailed information about the source converter when designing their impedance regulator. In other words, if the source converter is changed, either the PVI or SVI control strategy must re-design its impedance regulator. Obviously, this creates conflict with the purpose of the modularity design of the cascaded system, hence increasing the system development cycles. It is worth pointing out that this problem also occurs in the existing stabilisation methods.

In order to solve the above problem, the adaptive function is first introduced into the PVI control strategy in this chapter to generate a new stabilisation method: Adaptive PVI (APVI) control strategy. The APVI control strategy connects a self-regulated virtual impedance in parallel with the input port of the load converter. Thanks to this adaptive virtual impedance, the load converter can change its input impedance adaptively with different source converters to stabilise the whole system. The APVI control strategy can therefore make all the load converters into stable modules in different cascaded systems. Furthermore, the APVI control strategy inherits the advantage of the PVI control strategy, such as only modifying the load input impedance in a very small frequency range to keep the

original dynamic performance of the load converter. Therefore, the APVI control strategy can be treated as an improved method of the PVI control strategy, which makes the PVI control strategy more competitive in practice.

The rest of this chapter is organised as follows: in Section 4.1, the stability preferred load input impedance for the APVI control strategy is discussed. The adaptive concept, realisation method and operation flow of the APVI control strategy are then discussed in Section 4.2. Following that, two different cascaded systems composed by two different *LC* input filters and the same load converter, are designed and experimentally implemented to verify the effectiveness of the proposed APVI method in Section 4.3. Finally, Section 4.4 concludes the chapter.

4.1 Stability-preferred Input Impedance of the Load Converter with the APVI Control Strategy

Fig. 4.1(a) shows the Bode plots of the source output impedance Z_{oS} and the load input impedance Z_{iL} in a typical instability case. Here, if the source converter is a switchingmode power supply, f_{CS} is the cut-off frequency of its voltage loop; if the source converter is a *LC* input filter, f_{CS} is the filter resonant frequency. In addition, f_{CL} is the cut-off frequency of the voltage loop of the load converter. Z_{oSP} is the peak value of $|Z_{oS}|$. According to Fig. 4.1(a), the cause of the instability can be summarised as follows: if $|Z_{oS}|$ is intersected with $|Z_{iL}|$, and if f_{CS} is less than f_{CL} , the cascaded system is unstable even if the subsystems can work well individually (Sun, 2009). The -180° phase resulting from the load negative resistor characteristic is the main cause of the instability in the cascaded system (Zhang et al., 2015). According to Fig. 4.1(a), the worst instance of instability problems in a cascaded system occurs at full load and with the source converter being a *LC* input filter.

As seen in Chapters 2 and 3, there is nothing more desirable than changing Z_{iL} only in the vicinity of the intersection frequencies $(f_1 \text{ and } f_2)$ of $|Z_{oS}|$ and $|Z_{iL}|$, to stabilise the cascaded system while keeping a better dynamic performance from the load converter. Therefore, this chapter only increases the phase of Z_{iL} during $[f_1, f_2]$, which can ensure $|\varphi(Z_{oS}) - \varphi(Z_{iL})| < 180^0$ at the intersection frequencies range. Such shaping $\varphi(Z_{iL})$ method not only keeps the advantage of the existing PVI control strategy, but is also con-



Figure 4.1: Instability reason and stability-preferred input impedance of the load converter with the APVI control strategy: (a) instability reason; (b) stability-preferred input impedance of the load converter.



Figure 4.2: A cascaded system with adaptive parallel-virtual-impedance.

ducive to the realisation of the latter proposed APVI control strategy. As shown in Fig. 4.1(b), the stability-preferred input impedance of the load converter Z_{iLP} can be expressed as

$$Z_{iLP} = \begin{cases} -|Z_{iL1}| e^{j\theta} & f \in [f_1, f_2] \\ Z_{iL} & f \notin [f_1, f_2] \end{cases}$$
(4.1)

where, Z_{iL} is the original input impedance of the load converter, $|Z_{iL1}|$ and θ are the magnitude and phase of the improved input impedance of the load converter within $[f_1, f_2]$. In order to ensure $|\varphi(Z_{oS}) - \varphi(Z_{iLP})| < 180^{\circ}$, θ should satisfy $\theta \in (-90^{\circ}, 90^{\circ})$. Since $|\varphi(Z_{oS}) - \varphi(Z_{iLP})| < 180^{\circ}$ at $[f_1, f_2]$, $|Z_{iL1}|$ can be any values. In order to minimise the resulting effect on the load converter, $|Z_{iL1}|$ is designed to be equal to $|Z_{iL}|$ (see Fig. 4.1(b)).

4.2 APVI Control Strategy

4.2.1 Adaptive-Parallel-Virtual-Impedance *Z*_{APVI} and its Adaptive Approach

As shown in Fig. 4.2, if a virtual impedance Z_{APVI} is added in parallel with the input port of the load converter, the load input impedance is changed to

$$Z_{iLP} = Z_{iL} \parallel Z_{APVI} = \frac{Z_{iL} \cdot Z_{APVI}}{Z_{iL} + Z_{APVI}}$$

$$(4.2)$$

where Z_{iL} is the original load input impedance and equal to $-V_{bus}^2/P_o$ when $f < f_{CL}$. Here, V_{bus} and P_o are the input voltage and output power of the load converter, respectively.

According to (4.1) and (4.2), where the load input impedance is needed to meet (4.1), Z_{APVI} should be

$$Z_{APVI} = \begin{cases} \left(V_{bus}^2/P_o\right) \cdot \left[e^{j\theta}/(1+e^{j\theta})\right] & f \in [f_1, f_2] \\ +\infty & f \notin [f_1, f_2] \end{cases}$$
(4.3)

As seen from (4.3), within $[f_1, f_2]$, Z_{APVI} is a constant value, which is determined by the load converter, otherwise, $Z_{APVI} = +\infty$. Since f_1 and f_2 are very close to f_{CS} (see Fig. 4.1(a)), the frequency characteristics of Z_{APVI} are also affected by the source converter. In other words, for an actual load converter, if its source converter is changed, the frequency



Figure 4.3: Adaptively parallel-virtual-impedance $Z_{APVI}(f_{CS})$.



if $|\Delta v_{bus}| = 0$, PI_{out} finds the right f_{CS} , then locked this frequency

Figure 4.4: Basic idea of the adaptive approach of $Z_{APVI}(f_{CS})$.

characteristics of Z_{APVI} must be changed accordingly. It is presumed that if Z_{APVI} can be changed adaptively according to different source converters, as seen in Fig. 4.3, the load converter does not necessarily need to be changed any more. Therefore, realising the adaptive virtual impedance $Z_{APVI}(f_{CS})$ is the main aim of the proposed APVI control strategy.

Fig. 4.3 illustrates that $Z_{APVI}(f_{CS})$ is a function with respect to f_{CS} . Therefore, the key purpose of realising $Z_{APVI}(f_{CS})$ is to find f_{CS} when the source converter is changing. In order to solve this problem, an adaptive approach, as shown in Fig. 4.4, is proposed in this chapter. There are four basic steps to this approach. Firstly, $|\triangle v_{bus}|$ is sampled and sent to the positive input of a proportional–integral (PI) controller and compared with the PI negative input 0. Secondly, the output of the PI controller is set as the f_{CS} of $Z_{APVI}(f_{CS})$. Thirdly, according to (4.3) and the PVI control strategy in Chapter 2, $Z_{APVI}(f_{CS})$ is added to the input port of the load converter via control method. Finally, if $|\triangle v_{bus}|$ becomes 0, it means $Z_{APVI}(f_{CS})$ has found the right f_{CS} via the PI controller. At this point, a frequency-



Figure 4.5: Concept of the APVI control strategy.

locked block starts working to lock the right f_{CS} for the $Z_{APVI}(f_{CS})$. According to Fig. 4.4, with the proposed adaptive approach, $Z_{APVI}(f_{CS})$ can find its right f_{CS} when the source converter changes.

It is worth pointing out that Fig. 4.4 not only shows the basic adaptive approach of realising $Z_{APVI}(f_{CS})$, but also the main idea of the proposed APVI control strategy. Using Fig. 4.4, the APVI control strategy and its detailed realisation method will be discussed in Sections 4.2.2 and 4.2.3, respectively.

4.2.2 The APVI Control Strategy

Fig. 4.5 shows the small-signal control block of the original load converter. Its variables and transfer functions are described in Table 4.1. If Z_{APVI} must be added in parallel with the input port of the load converter, one intuitive way is to introduce $1/Z_{APVI}$ to the control block between the input voltage and input current (as represented with dashed lines in Fig. 4.5). However, this method cannot be achieved by control directly. In order to address this issue, the output of $1/Z_{APVI}$ is moved to the output voltage reference and equivalent adjustments to the transfer function to $G_{APVI}(s)$, as shown by the dot-dashed lines. Fig. 4.5 is the concept of the proposed APVI control strategy, and $G_{APVI}(s)$ is expressed as

$$G_{APVI}(s) = \frac{1}{Z_{APVI}(s)} \cdot \frac{1 + T_v(s)}{G_v(s) \cdot G_M(s) \cdot G_{id}(s)}$$
(4.4)

лс т .т.	variables and transfer functions of foad converte
\hat{v}_{bus}	Perturbation of the bus voltage
î,	Perturbation of the load current
î _{bus}	Perturbation of the bus current
â	Perturbation of the duty cycle
\hat{v}_o	Perturbation of the output voltage
\hat{v}_{or}	Perturbation of the output voltage reference
$G_{v}(s)$	Transfer function of the voltage regulator
$G_M(s)$	Transfer function of the modulator
$Z_{iOL}(s)$	Open-loop input impedance
$G_{vd}(s)$	Control to output voltage transfer function
$G_{id}(s)$	Control to input current transfer function
$Z_{oOL}(s)$	Open-loop output impedance
$H_s(s)$	Sampling coefficient of the output voltage
$G_{iOL}(s)$	Open-loop load to input current transfer function
$\overline{G_{vOL}(s)}$	Open-loop input to output voltage transfer function

Table 4.1: Variables and transfer functions of load converter



Figure 4.6: Adaptive characteristic of $|G_{APVI}(s)|$.

where $T_v(s) = H_s(s)G_v(s)G_M(s)G_{vd}(s)$ is the voltage-closed loop gain of the load converter.

Note: since the derivation of $G_{APVI}(s)$ is very similar to $G_{PVI}(s)$ in Chapter 2, it will not be repeated here.

4.2.3 Realisation of the APVI Control Strategy

Substituting (4.3) for (4.4), the adaptive regulator $G_{APVI}(s)$ can be further derived as

$$G_{APVI}(s) = \begin{cases} \frac{P_o(\frac{1+e^{j\theta}}{e^{j\theta}})[1+T_v(s)]}{V_{bus}^2 G_v(s) G_M(s) G_{id}(s)} = \frac{P_o(1+e^{-j\theta})[1+T_v(s)]}{V_{bus}^2 G_v(s) G_M(s) G_{id}(s)} & f \in [f_1, f_2] \\ 0 & f \notin [f_1, f_2] \end{cases}$$
(4.5)

As seen in (4.5), $G_{APVI}(s)$ is an adaptive frequency-based piecewise function, which is shown in Fig. 4.6, where $|G_{APVI}(s)|$ is a non-zero value within $[f_1, f_2]$ and zero outsides $[f_1, f_2]$. Since f_1 and f_2 are very close to f_{CS} , an adaptive band-pass filter $G_{BPFa}(s)$, whose



Figure 4.7: Realisation of the $G_{APVI}(s)$: (a) control circuit of $G_{APVI}(s)$; (b) adaptive mechanism of $G_{APVI}(s)$.

centre frequency is f_{CS} , can be utilised to realise $G_{APVI}(s)$, i.e.,

$$G_{APVI}(s) = G_{APVI1}(s) \cdot G_{BPFa}(s) \tag{4.6}$$

where

$$G_{APVI1}(s) = \frac{P_o}{V_{bus}^2} \frac{(1 + e^{-j\theta}) [1 + T_v(s)]}{G_v(s) G_M(s) G_{id}(s)}$$
(4.7)

$$G_{BPFa}(s) = -\frac{s \left(2\pi f_{CS}/Q\right)}{s^2 + s \left(2\pi f_{CS}/Q\right) + \left(2\pi f_{CS}\right)^2}$$
(4.8)

where, Q is the quality factor of the band pass filter, whose initial value is recommended as 0.707 and can be changed manually as needed; f_{CS} is changed adaptively according to different source converters. For (Cuk, 1976), the numerator degree of $G_{APVI}(s)$ is always lower than its denominator degree for all DC/DC converters. As a result, $G_{APVI}(s)$ is a proper transfer function that can be fully implemented by digital chips. To ensure $G_{APVI}(s)$ can realise $Z_{APVI}(s)$ and find f_{CS} adaptively, a specific control circuit for $G_{APVI}(s)$ is proposed in this section. Fig. 4.7(a) illustrates this control circuit which contains five parts and operates as follows:

In Part A, v_{bus} first goes through a high-pass-filter (HPF) and an absolute value block to extract $|\triangle v_{bus}|$. Following this, $|\triangle v_{bus}|$ goes through Parts B and C to find f_{CS} adaptively. As shown in Figs. 4.7(a) and (b), the adaptive mechanism of Parts B and C can be explained as: in Part B, $|\triangle v_{bus}|$ is firstly compared with 0, meanwhile, its error is amplified by a PI controller. Therefore, the output of the PI controller, PI_{out} , can be expressed as:

$$PI_{out}(t) = K_p \left| \triangle v_{bus}(t) \right| + K_i \int_0^t \left| \triangle v_{bus}(t) \right| dt$$
(4.9)

where K_p and K_i are the proportional and integral coefficients of PI controller, respectively.

According to (4.9), if the system is unstable, $|\triangle v_{bus}| > 0$, PI_{out} would be increased from zero. By Fig. 4.7(a), PI_{out} also decides the centre frequency of $Z_{APVI}(s)$. Therefore, once PI_{out} is increased to f_{CS} , $Z_{APVI}(s)$ will find its right centre frequency, stabilise the cascaded system and make $|\triangle v_{bus}| = 0$. Then according to Part C and Fig. 4.7(b), the input 3 of S_1 is equal to $0 \rightarrow$ the output of S_1 becomes $0 \rightarrow$ the output of *Relay* changes to $1 \rightarrow$ the input 2 of S_1 changes to $1 \rightarrow$ the output of S_1 is changed to the input 1 of S_1 and becomes 0 again \rightarrow the output of *Relay* is locked as 1. Therefore, the output of S_2 , $S2_{Out}$, is always equal to its previous value

$$S2_{Out}(n+1) = S2_{Out}(n) = f_{CS}$$
(4.10)

where n + 1 and n represent the time n + 1 and its previous time n, respectively. Both are after PI_{out} finding f_{CS} .

According to Fig. 4.7 and (4.10), as $S2_{Out}$ is equal to f_{CS} before the relay is locked, the centre frequency of $Z_{APVI}(s)$ is finally locked as f_{CS} and will not be changed any more. At the same time, v_{bus} is sent to Part D to realise $G_{APVI1}(s)$ in (4.7). Both of the outputs of Parts C and D are then sent to Part E, where the $G_{BPFa}(s)$ can be achieved. Finally, the output of $G_{APVI}(s)$ can be obtained from the output of Part E.

Note that, though $G_{APVI}(s)$ can lock f_{CS} as the final centre frequency of $G_{BPFa}(s)$ after finding it, too large K_p and K_i parameters should be avoided in the design of the PI con-



Figure 4.8: Impact of K_p and K_i on $G_{APVI}(s)$: (a) large K_p and K_i ; (b) small K_p and K_i .

troller in Part B of Fig. 4.7. As can be seen in Fig. 4.8(a), this is because if K_p and K_i are too large, this may cause the output of the PI controller to increase so much during a sampling period, and consequently for f_{CS} to be skipped. For this reason, small K_p and K_i are preferred for the PI controller (Fig. 4.8(b)). The only compromise is that the $G_{APVI}(s)$ may spend a slightly longer time before the right f_{CS} is founded. However, if the APVI controller finds the right f_{CS} , the PI controller will be disabled and the K_p and K_i do not affect any steady or dynamic performance of the load converter. Therefore, small K_p and K_i are accepted in practice.

4.2.4 Operation Flow of the APVI Control Strategy

Fig. 4.9 gives the operation flow of the APVI control strategy. First of all, the user needs to check both source and load converters to make sure they can work well individually. Then, if the cascaded system is unstable, $G_{APVI}(s)$ will be utilised. For $G_{APVI}(s)$, its initial value of Q is set as 0.707. This is because Q=0.707 can ensure a relatively narrow passband for $G_{BPFa}(s)$, and if 0.707 is also suitable for Q, the $G_{APVI}(s)$ not only can stabilise the cascaded system, but also keep the original dynamic performance of the load converter. However, if the system is still unstable, this means that 0.707 is too much larger than Q, i.e., the bandwidth of $G_{BPFa}(s)$ is smaller than $(f_2 - f_1)$. As a result, the user should reduce Q and test the system again until the suitable Q is found.

It is worth pointing out that, though the type of the typical input voltage feed-forward



Figure 4.9: Operation flow of the APVI control method.

control loop seems similar to the proposed APVI control block, they are in fact entirely different. The purpose of the typical input voltage feed-forward control loop is to regulate the input impedance of the load converter to a pure negative resistor within the whole frequency range. However, the purpose of the APVI control method is to remove the negative resistor characteristic from the input impedance of the load converter. As a result, the proposed APVI control method can stabilise the cascaded system, but the typical input voltage feed-forward method cannot.

4.3 Experimental Verification

As the worst instability phenomena are most likely to arise in a cascaded system whose source converter is a *LC* input filter, the APVI control strategy is applied into two such unstable 100 W cascaded systems. As shown in Fig. 4.10, the two cascaded systems utilise the same load stage, which is a 48 V-24 V Buck converter with a 20 kHz switching frequency. If the load stage is connected to the source stage 1, which is an input filter formed by a 6 mH inductor and a 150 μ F capacitor, the first unstable system is formed. On the other hand, if the load converter is connected to the source stage 2, which is an input filter whose inductor and capacitor are 6 mH and 60 μ F, respectively, the second unstable



Figure 4.10: Example cascaded systems.



Figure 4.11: Bode plots of example systems at full load.

system is formed. It is worth pointing out that, since the load switching frequency in this chapter is different to that discussed in Chapter 2, the parameters of the *LC* input filter are also different.

Fig. 4.11 shows the Bode plots of the example systems, where Z_{oS1} , Z_{oS2} and Z_{iL} are the output impedance of source stage 1, the output impedance of source stage 2 and the input impedance of the load converter at full load, respectively. It can be seen that both of the two cascaded systems are unstable.

According to Figs. 4.5 and 4.7, an adaptive input impedance regulator $G_{APVI}(s)$ is introduced to the load converter. Here, θ is selected as 0. Then, as shown in Fig. 4.11, Z_{iL}

is modified into Z_{iLP1} in the first cascaded system and Z_{iLP2} in the second cascaded system adaptively. In both cases, $|\varphi(Z_{oS}) - \varphi(Z_{iLP})| < 180^{0}$ at the intersection frequencies of $|Z_{oS}|$ and $|Z_{iL}|$. Therefore, the modified cascaded system with the APVI control method is stable and the design of $G_{APVI}(s)$ is appropriate. In addition, Fig. 4.11 shows that, $G_{APVI}(s)$ can achieve a modified input impedance in Fig. 4.1(b), i.e., only changing the phase of the load input impedance in a very small frequency range, which minimise the resulting effect on the load converter.

The experimental results of the first example system are given in Fig. 4.12, where the waveforms of v_{bus} and v_o are their ac components to clearly show the oscillation (i_L , v_{bus} , i_{L1} and v_o are illustrated in Fig. 4.10). As seen in Fig. 4.12(a), this cascaded system is unstable without the APVI control strategy. After the APVI control strategy was utilised, it spent about 45.56 ms to adjust the parameters of $G_{APVI}(s)$ and to stabilise the cascaded system automatically. Fig. 4.12(b) shows the dynamic waveforms of the modified system with the APVI control strategy when its input voltage steps down from 100% rated to 80% rated voltage at the full load. It seems that, the APVI control strategy can work well during an input voltage. Fig. 4.12(c) shows the dynamic waveforms of the modified system with the APVI control strategy when its load increases from 10% rated to 100% rated load at the rated input voltage. This demonstrates that the APVI control strategy can also work well during a load change.

Similarly, Figs. 4.13(a), (b) and (c) give the steady state waveforms, input voltage and load dynamic waveforms of the second example system. Again, it verifies that the APVI control strategy can not only solve the instability problem of the cascaded system adaptively, but also work well during the dynamic process. The above experimental results can be said to have verified the effectiveness of the proposed APVI control strategy.

It is worth pointing out that, though the APVI control strategy introduces the adaptive function to the PVI control strategy and can be treated as an improved PVI control strategy, it needs relatively more time to search the centre frequency of the APVI. Therefore, the execution time of the APVI algorithm is longer than that of the PVI algorithm. Since the switching period of a converter should be larger than the algorithm execution time, the load experimental switching frequency is decreased from 100 kHz in Chapter 2 to 20 kHz in this chapter. Since low switching frequency always leads to a larger inductor and capacitor



Figure 4.12: Experimental waveforms of the first cascaded system with the APVI control strategy: (a) steady state waveforms with rated input voltage and rated load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated to 100% rated load at rated input voltage.



Figure 4.13: Experimental waveforms of the second cascaded system with the APVI control strategy: (a) steady state waveforms with rated input voltage and rated load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated to 100% rated load at rated input voltage.

of the *LC* input filter, the parameters of the *LC* input filter in this chapter are also different from that discussed in Chapter 2. Therefore, the relatively longer algorithm execution time is the restriction of the APVI control strategy.

4.4 Summary

In order to solve the instability problem and maintain the modularisation characteristics of the cascaded system, an APVI control strategy is proposed in this chapter. This demonstrates that the APVI control strategy can adaptively regulate the input impedance of the load converter for different source converters by adding a self-regulated virtual impedance in parallel with the load converter. As a result, the APVI control strategy means that the load converter can be designed as a standard module stably connected to any source converter. Additionally, the APVI control strategy also inherits the advantage of the PVI control strategy to keep most of the dynamic performance of the load converter. Therefore, the APVI control strategy can be treated as an improved method of the PVI control strategy. Finally, the APVI control strategy has been experimentally verified on two cascaded systems.

Chapter 5

Adaptive-SVI (ASVI) Control for Load Converters

As demonstrated in Chapters 2 and 3, the cascaded system can be stabilised via amplitude compensation (SAC) or phase compensation (SPC) for the input impedance of the load converter. Since both SAC and SPC can be realised by both PVI and SVI control strategies, a further analysis of SAC/SPC and a further investigation of the PVI / SVI control strategies are carried out in this chapter. Firstly, it is shown that the cascaded system adopting the SAC is unconditionally stable, but conditionally stable when adopting the SPC, i.e., the cascaded system with SAC is more stable than that with SPC. It is then noted that the PVI control strategy has inevitable limitations when realising the SAC during the whole load and input voltage range of the load converter, but the SVI control strategy does not have limitations. In other words, the SVI control strategy is more competent for the SAC. Furthermore, in order to stably connect a load converter to different source converters, such as LC input filters and traditional DC/DC converters, without changing its internal structure, the adaptive function is also introduced to the traditional SVI control strategy in this chapter. With the Adaptive SVI (ASVI) control strategy, the series-virtual-impedance is able to adaptively regulate its characteristic to stabilise the cascaded system according to different source converters.

It should be stressed that although the adaptive characteristics of the ASVI control strategy are similar to those of the APVI control strategy, they are entirely different from in terms of physical concept, stabilisation method and realisation approach. The APVI control strategy adds a virtual impedance in parallel with the load converter by an APVI



Figure 5.1: Bode plots of SAC and SPC: (a) SAC; (b) SPC.

controller to realise SPC. However, the ASVI control strategy adds a virtual impedance in the series with the load converter by an ASVI controller to realise SAC. Since SAC is more stable than SPC, the ASVI control strategy is superior to the APVI control strategy.

The remaining parts of this chapter are organised as follows: in Section 5.1, SAC and SPC are compared. The PVI and SVI control strategies are then compared in Section 5.2. Following this, the concept, realisation and impact of the ASVI control strategy are discussed in Section 5.3. Section 5.4 gives the experimental verification of the proposed ASVI control strategy. Finally, Section 5.5 concludes the chapter.

5.1 Stabilisation via Amplitude Compensation (SAC) and Phase Compensation (SPC)

As SAC and SPC are two effective stabilisation methods for the load converter, they are reviewed and compared carefully in this section. As shown in Figs. 5.1(a) and (b), SAC increases $|Z_{iL}|$ to keep a total separation with $|Z_{oS}|$, while SPC increases $\varphi(Z_{iL})$ to ensure $|\varphi(Z_{oS}) - \varphi(Z_{iL})| < 180^{\circ}$ at the intersection frequencies (f_1 and f_2 in Fig. 5.1) of $|Z_{iL}|$ and $|Z_{oS}|$. As both SAC and SPC only change Z_{iL} in a very small frequency range, they can stabilise the cascaded system with the minimised load performance compromise.

However, though both SAC and SPC can stabilise the cascaded system with minimised performance degradation, their stabilising effects are essentially different, as can be seen



Figure 5.2: Nyquist plots of Z_{oS}/Z_{iL} with SAC and SPC: (a) with SAC; (b) with SPC.

from the Nyquist plots of two typical cases shown in Fig. 5.2. For the SAC, since it increases $|Z_{iL}|$ and ensures $|Z_{iL}| > |Z_{oS}|$ in the whole frequency range, the system loop gain Z_{oS}/Z_{iL} always stays inside the unit cycle, as shown in Fig. 5.2(a) (Middlebrook, 1979). Hence, with the SAC, the improved cascaded system is an unconditionally stable system. However, for the SPC, it only increases $\varphi(Z_{iL})$ to ensure $|\varphi(Z_{oS}) - \varphi(Z_{iL})| < 180^{\circ}$ during (f_1, f_2) but does not change $|Z_{iL}|$. As a result, for the SPC, though Z_{oS}/Z_{iL} does not encircle (-1, j0), it intersects with the unit cycle, i.e., the stability of this system still depends on the phase margin of Z_{oS}/Z_{iL} , see PM_1 and PM_2 in Fig. 5.2(b). In other words, with the SPC, the improved cascaded system is actually a conditionally stable system. Therefore, the SAC is more stable than the SPC and should be considered the preferred stabilisation method to shape the load input impedance.

5.2 Comparison of PVI and SVI Control Strategies

SAC is the preferred load stabilisation method. As noted in Chapters 2 and 3, as it can be realised by both PVI and SVI control strategies, a comparison of PVI and SVI control strategies is carried out in this section.

5.2.1 Limitation of PVI Control Strategy When Realising the SAC

As illustrated by Fig. 5.3, SAC can be realised by the PVI control strategy via adding a virtual impedance Z_{PVI} in parallel with the load converter. As the input impedance of the original load converter is $-V_{bus}^2/P_o$, the improved load input impedance Z_{iLP} can be expressed as

$$Z_{iLP} = \frac{-Z_{PVI} \cdot (V_{bus}^2/P_o)}{Z_{PVI} - (V_{bus}^2/P_o)}$$
(5.1)

where P_o and V_{bus} are the output power and voltage of the load converter, respectively. According to the PVI control strategy in Chapter 2, Z_{PVI} is a constant positive resistor inside $[f_1, f_2]$ and $+\infty$ outside $[f_1, f_2]$. Thus, $|Z_{iLP}|$ is only increased during $[f_1, f_2]$.

According to (5.1), during $[f_1, f_2]$, the curve of $|Z_{iLP}|$ to $\frac{P_o}{V_{bus}^2}$ is depicted in Fig. 5.4. As seen, $|Z_{iLP}|$ has three characteristics: 1) $|Z_{iLP}|$ is monotonically increased from $|Z_{PVI}|$ to $+\infty$ when $\frac{P_o}{V_{bus}^2}$ increases from 0 to $\left|\frac{1}{Z_{PVI}}\right|$; 2) $|Z_{iLP}|$ is monotonically decreased from $+\infty$ to 0 when $\frac{P_o}{V_{bus}^2}$ increases from $\left|\frac{1}{Z_{PVI}}\right|$ to $+\infty$; 3) $|Z_{iLP}| = |Z_{PVI}|$ when $\frac{P_o}{V_{bus}^2} = 0$ or $\frac{P_o}{V_{bus}^2} = \left|\frac{2}{Z_{PVI}}\right|$ respectively. Therefore, if the peak value of $|Z_{oS}|$ is Z_{oSP} , and if a total separation between $|Z_{oS}|$ and $|Z_{iLP}|$ (i.e., SAC) during the full load and input voltage range of the load converter is required, the curve of $|Z_{iLP}|$ during $[f_1, f_2]$ should be limited in the shadow part of Fig. 5.4, i.e., $|Z_{PVI}|$ should satisfy the following conditions:

$$|Z_{PVI}(j2\pi f)| > |Z_{oSP}| 10^{\frac{GM}{20}} \qquad f \in [f_1, f_2]$$
(5.2)

$$\frac{2}{|Z_{PVI}(j2\pi f)|} > Max\left(\frac{P_o}{V_{bus}^2}\right)_L \qquad f \in [f_1, f_2]$$
(5.3)

where *GM* is the gain margin of $\frac{Z_{oS}}{Z_{iLP}}$ and its unit is $dB\Omega$. $Max\left(\frac{P_o}{V_{bus}^2}\right)_L$ is equal to $\frac{P_{oM}}{V_{bM}^2}$, where P_{oM} and V_{bM} are the maximum power and the minimum input voltage of the load converter, respectively.

According to (5.2) and (5.3), the selection range of $|Z_{PVI}|$ can be derived as

$$|Z_{oSP}| 10^{\frac{GM}{20}} < |Z_{PVI}(j2\pi f)| < \frac{2V_{bM}^2}{P_{oM}} \quad f \in [f_1, f_2]$$
(5.4)

By (5.4), during $[f_1, f_2]$, the value of $|Z_{PVI}|$ is effective if and only if the following



Figure 5.3: The PVI control strategy and its physical concept.



Figure 5.4: Requirement of Z_{PVI} during $[f_1, f_2]$ when realising the SAC

condition is satisfied

$$|Z_{oSP}| 10^{\frac{GM}{20}} < \frac{2V_{bM}^2}{P_{oM}}$$
(5.5)

In other words, if (5.5) is not satisfied, the PVI control strategy cannot help the cascaded system realise the SAC during the whole load and input voltage range of the load converter. As a result, (5.5) is the limitation of the PVI control strategy when realising the SAC.

5.2.2 Advantage of SVI Control Strategy When Realising the SAC

Though PVI control strategy has inevitable limitations when achieving the SAC, the SVI control strategy is fully competent for the SAC without limitation. As shown in Fig. 5.5, SAC can be realised by the SVI control strategy via adding a virtual impedance Z_{SVI} in series with the load converter. The improved load input impedance Z_{iLS} is expressed as

$$Z_{iLS} = Z_{SVI} - \left(V_{bus}^2 / P_o\right) \tag{5.6}$$

where Z_{SVI} is a constant negative resistor inside $[f_1, f_2]$ and 0 outside $[f_1, f_2]$ according to analysis in Chapter 3. Similar to $|Z_{iLP}|$, $|Z_{iLS}|$ is also only increased during $[f_1, f_2]$.

According to (5.6), during $[f_1, f_2]$, the curve of $|Z_{iLS}|$ to $\frac{P_o}{V_{bus}^2}$ is depicted in Fig. 5.6. As seen, $|Z_{iLS}|$ is monotonically decreased from $+\infty$ to $|Z_{SVI}|$ when $\frac{P_o}{V_{bus}^2}$ increases from



Figure 5.5: The SVI control strategy and its physical concept.



Figure 5.6: Requirement of $|Z_{SVI}|$ during $[f_1, f_2]$ when realising the SAC.

0 to $+\infty$. In addition, the value of $|Z_{iLS}|$ is equal to $Min\left(\frac{V_{bus}^2}{P_o}\right)_L + |Z_{SVI}|$ when $\frac{P_o}{V_{bus}^2} = Max\left(\frac{P_o}{V_{bus}^2}\right)_L$. Here $Min\left(\frac{V_{bus}^2}{P_o}\right)_L = \frac{V_{bM}^2}{P_{oM}}$, $Max\left(\frac{P_o}{V_{bus}^2}\right)_L = \frac{P_{oM}}{V_{bM}^2}$. Therefore, if the SAC must be realised by the SVI control strategy during the full load and input voltage range of the load converter, the curve of $|Z_{iLS}|$ during $[f_1, f_2]$ should be limited in the shadow part of Fig. 5.6, i.e., $|Z_{SVI}|$ should satisfy the following condition:

$$Min\left(\frac{V_{bus}^{2}}{P_{o}}\right)_{L} + |Z_{SVI}(j2\pi f)| > |Z_{oSP}| \, 10^{\frac{GM}{20}} \qquad f \in [f_{1}, f_{2}]$$
(5.7)

According to (5.7), the requirement of $|Z_{SVI}|$ can be derived as

$$|Z_{SVI}(j2\pi f)| > |Z_{oSP}| \, 10^{\frac{GM}{20}} - \frac{V_{bM}^2}{P_{oM}} \qquad f \in [f_1, f_2]$$
(5.8)

Obviously, for the cascaded system, during $[f_1, f_2]$, a higher $|Z_{SVI}(j2\pi f)|$ can always be found to meet the requirement of (5.8). As a result, the SVI control strategy can help the cascaded system to realise SAC during the whole load and input voltage range of the load converter without limitation. This is also the advantage of the SVI control strategy when realising SAC.

Therefore, the SVI control strategy is more suitable for the SAC than the PVI control

strategy.

5.3 The ASVI Control Strategy

5.3.1 Adaptive-Series-Virtual-Impedance Z_{ASVI} and its Adaptive Approach

According to (5.6) and Fig. 5.5, if $|Z_{iL}|$ must be changed to $|Z_{iLS}|$, Z_{SVI} can be selected as

$$Z_{SVI} = \begin{cases} Z_{iLS} + (V_{bus}^2/P_o) & f \in [f_1, f_2] \\ 0 & f \notin [f_1, f_2] \end{cases}$$
(5.9)

(5.8) and (5.9) llustrate that during $[f_1, f_2]$, $|Z_{SVI}|$ should be large enough to ensure that $|Z_{iLS}| > |Z_{oS}|$. In addition, as f_1 and f_2 are very close to f_{CS} (See Fig. 5.1(a)), the frequency characteristics of Z_{SVI} are also affected by the source converter. In other words, if Z_{SVI} wants to become an adaptive series-virtual-impedance Z_{ASVI} to help the load converter realise SAC with different source converters, it should satisfy three basic requirements: 1) when $f \in [f_1, f_2]$, $|Z_{ASVI}|$ should be large enough to ensure a total separation between $|Z_{iLS}|$ and $|Z_{oS}|$; 2) when $f \notin [f_1, f_2]$, $|Z_{ASVI}|$ should be small enough to minimise the impact of the ASVI control strategy on the original load converter; 3) $[f_1, f_2]$ could be changed adaptively according to different source converters. In line with these requirements, the adaptive characteristics can be introduced into a well known non-ideal resonant controller to achieve the improved adaptive resonant controller. As shown in Fig. 5.7, this adaptive controller can mimic Z_{ASVI} as

$$Z_{ASVI}(s) = -\frac{2K_r (2\pi f_{RC})s}{s^2 + 2(2\pi f_{RC})s + (2\pi f_{CS})^2}$$
(5.10)

where f_{CS} is determined by the source converter and can be changed adaptively. f_{RC} is the bandwidth at -3dB cut-off frequency of the $Z_{ASVI}(s)$, whose value is recommended as 5Hz. $|Z_{ASVI}(s)|$ during $(f_{CS} - f_{RC}, f_{CS} + f_{RC})$ is equal to $\frac{K_r}{\sqrt{2}}$, whose value is not affected by f_{CS} and can choose 100Ω as its initial value. It is worth pointing out that in most cases $\frac{K_r}{\sqrt{2}} = 100\Omega$ is large enough for $|Z_{ASVI}(s)|$ to avoid the intersection between $|Z_{iLS}(s)|$ and $|Z_{oS}(s)|$ during the whole load and input voltage range of the load converter. However, even



Figure 5.7: Characteristic of $|Z_{ASVI}(f_{CS})|$.



if $|\Delta v_{bus}| = 0$, PI_{out} finds the right f_{CS} , then locked this frequency

Figure 5.8: Basic idea of the adaptive approach of $Z_{ASVI}(f_{CS})$.

if it is not enough, K_r can be increased and tested by the user flexibly until the suitable K_r is found.

As discussed above, the realising of the adaptive virtual impedance $Z_{ASVI}(f_{CS})$ is the main aim of the proposed ASVI control strategy. According to Fig. 5.7, $Z_{ASVI}(f_{CS})$ is a function with respect to f_{CS} . Therefore, the key purpose of realising $Z_{ASVI}(f_{CS})$ is finding f_{CS} when the source converter changes. In order to solve this problem, this chapter utilises a similar adaptive approach to Chapter 4. As illustrated in Fig. 5.8, there are four basic steps to this adaptive approach. Firstly, $|\triangle v_{bus}|$ is sampled and sent to the positive input of a proportional–integral (PI) controller and compared with the PI negative input 0. Secondly, the output of the PI controller is set as f_{CS} of $Z_{ASVI}(f_{CS})$. Thirdly, as shown in (5.10) and the SVI control strategy in Chapter 3, $Z_{ASVI}(f_{CS})$ is added to the input port of the load converter via the control method. Finally, if $|\triangle v_{bus}|$ becomes 0, it indicates that $Z_{ASVI}(f_{CS})$ has found the right f_{CS} via the PI controller. At this point, a frequency-locked block begins



Figure 5.9: The ASVI control strategy

to lock the right f_{CS} for the $Z_{ASVI}(f_{CS})$. As a result, the above adaptive approach allows $Z_{ASVI}(f_{CS})$ to find the right f_{CS} when the source converter is changing.

It is worth pointing out that Fig. 5.8 not only shows the basic adaptive approach of realising $Z_{ASVI}(f_{CS})$, but also the main idea of latter proposed ASVI control strategy. Based on Fig. 5.8, the ASVI control strategy and its detailed realisation method will be discussed in Sections 5.3.2 and 5.3.3, respectively.

5.3.2 Concept of the ASVI Control Strategy

Fig. 5.9 shows the small-signal control block diagram of the original load converter. Its variables and transfer functions are described in Table 5.1 as well. If $Z_{ASVI}(s)$ is to be added to the series with the input port of the load converter, one intuitive method is to introduce $Z_{ASVI}(s)$ to the control block between the load input current and load input voltage (as shown by the dashed lines in Fig. 5.9). This is also the basic idea of the ASVI control strategy. However, this method cannot be achieved by control directly. In order to address this issue, the output of $Z_{ASVI}(s)$ is moved to the output of $G_v(s)$, adjusting the transfer function to $G_{ASVI}(s)$, as shown by the dot-dashed lines in 5.9. Here, Fig. 5.9 is the concept

· · · · · · · · · · · · · · · · · · ·
Perturbation of the bus voltage
Perturbation of the load current
Perturbation of the bus current
Perturbation of the duty cycle
Perturbation of the output voltage
Perturbation of the output voltage reference
Transfer function of the voltage regulator
Transfer function of the modulator
Open-loop input impedance
Control to output voltage transfer function
Control to input current transfer function
Open-loop output impedance
Sampling coefficient of the output voltage
Open-loop load to input current transfer function
Open-loop input to output voltage transfer function

Table 5.1: Variables and transfer functions of load converter



Figure 5.10: Control system of the load converter with the ASVI control strategy.

of the ASVI control strategy and $G_{ASVI}(s)$ is expressed as:

$$G_{ASVI}(s) = Z_{ASVI} \cdot \frac{[1 + T_v(s)]P_o}{G_{id}(s)G_M(s)V_{bus}^2} = \frac{-2K_r (2\pi f_{RC})s}{s^2 + 2(2\pi f_{RC})s + (2\pi f_{CS})^2} \cdot \frac{[1 + T_v(s)]P_o}{G_{id}(s)G_M(s)V_{bus}^2}$$
(5.11)

where $T_v(s) = H_s(s)G_v(s)G_M(s)G_{vd}(s)$ is the loop gain of the voltage closed-loop of the load converter.

Note: since the derivation of $G_{ASVI}(s)$ is very similar to $G_{SVI}(s)$ in Chapter 3, it will not be repeated here.



Figure 5.11: Realisation of $G_{ASVI}(s)$.

5.3.3 Realisation of the ASVI Control Strategy

According to Fig. 5.9, the control system of the load converter with the ASVI control strategy is further presented in Fig. 5.10. In contrast to the control system of the original load converter, the ASVI control strategy adds only $G_{ASVI}(s)$ to the original control system (as shown with dashed lines in Fig. 5.10). Therefore, realising $G_{ASVI}(s)$ is the key to the ASVI control strategy. In addition, as shown in Fig. 5.10, both i_{bus} and v_{bus} are sampled to help achieve $G_{ASVI}(s)$ in practice.

According to (5.11) and Fig. 5.9, $G_{ASVI}(s)$ can be achieved by Fig. 5.11. As shown in Fig. 5.11, in Part A, v_{bus} first goes through a high-pass filter $\frac{s}{s+15}$ and an absolute value block to extract $\triangle v_{bus}$. Following this, $|\triangle v_{bus}|$ is sent to Part B and compared with zero, whilst, its error is amplified by a proportional–integral (PI) controller. If the system is unstable, the output of this PI controller is increased from zero, regulating the centre frequency of $Z_{ASVI}(s)$. If the output of the PI controller arrives at f_{CS} , $Z_{ASVI}(s)$ will find its right frequency characteristics, stabilise the cascaded system and make $|\triangle v_{bus}| = 0$. Then according to Part C, the input 3 of S_1 is equal to $0 \rightarrow$ the output of S_1 becomes $0 \rightarrow$ the output of *Relay* changes to $1 \rightarrow$ the input 2 of S_1 changes to $1 \rightarrow$ the output of S_1 is changed to the input 1 of S_1 and becomes 0 again \rightarrow the output of *Relay* is locked as 1, which locks the output of S_2 as the final centre frequency of $Z_{ASVI}(s)$ and does not change it during the remaining running time. At the same time, i_{bus} is sent to Part D to realise $\frac{[1+T_v(s)]P_o}{G_{id}(s)G_{PWM}(s)V_{bus}^2}$. Then, the output of Part D is sent to Part E, where $Z_{ASVI}(s)$ can be achieved by taking



Figure 5.12: Bode plots of the load converter with/without the ASVI control strategy: (a) $Z_{iL}(s)$; (b) $G_{iL}(s)$; (c) $G_{\nu L}(s)$; (d) $Z_{oL}(s)$.

its centre frequency from Part C. Finally, the output of $G_{ASVI}(s)$ can be obtained from the output of Part E. It is worth pointing out that, according to the analysis in Chapter 4, in order to ensure $Z_{ASVI}(s)$ finds f_{CS} effectively, too large K_p and K_i parameters should be avoided during the design process of the PI controller in Part B. In addition, as Chapter 4 has described how the PI control scheme is used to get the frequency adaptive scheme of the $Z_{ASVI}(s)$, it is only briefly explained here.

In summary, the proposed ASVI control strategy can be realised by Figs. 5.10 and 5.11.

5.3.4 Impact of the ASVI Control Strategy on the Load Converter

The performance of the load converter can be evaluated using two-port network analysis with four typical transfer functions (Arnedo, 2008): the closed-loop input impedance $Z_{iL}(s)$, the closed-loop load to input current transfer function $G_{iL}(s)$, the closed-loop input to out-
put voltage transfer function $G_{vL}(s)$ and the closed-loop output impedance $Z_{oL}(s)$. Therefore, to evaluate the impact of the ASVI control strategy on the load converter in a clear way, the Bode plots of $Z_{iL}(s) \sim Z_{oL}(s)$ of a specific load converter with/without the ASVI control strategy are depicted in Fig. 5.12. Here, the source and load converters correspond to the source converter I and the load converter in Fig. 5.13.

According to Fig. 5.12, though the proposed ASVI control strategy changes the features of $Z_{iL}(s) \sim Z_{oL}(s)$, it keeps most of the dynamic performance of the original load converter. This phenomenon can be explained as follows: since $Z_{ASVI}(s)$ only plays its role during $[f_1, f_2]$, but becomes zero outside $[f_1, f_2]$, the ASVI control strategy only affects the performance of the load converter during $[f_1, f_2]$. As a result, the ASVI control strategy can be considered an acceptable stabilisation method for the load converter.

5.4 Experimental Verification

In this section, the ASVI control strategy is applied to a 100 W unstable cascaded system, which contains one load converter and three different source converters. As shown in Fig. 5.13, the source converter I is a 100 V - 48 V / 50 kHz Buck converter, source converters II & III are two different *LC* input filters, and the load converter is a 48 V-24 V / 20 kHz Buck converter. Here, in order to compare the ASVI control strategy and the APVI control strategy, the source converters II & III and the load converter are the same as the converters in the experimental system of Chapter 4. The main circuit and parameters of the experimental system are also presented in Fig. 5.13. For convenience, the system is referred to work at Cases I, II and III when the load converter is connected to source converters I, II and III, respectively.

In Case I, the Bode plots of the cascaded system when the input voltage of the load converter is varied between 80% rated input voltage (38.4*V*) and 120% rated input voltage (57.6*V*) are presented in Figs. 5.14(a) and (b), where Z_{oS} , Z_{iL} and Z_{iLS} are the output impedance of the source converter, the original load input impedance and the load input impedance with the ASVI control strategy, respectively. As seen, when v_{bus} is changed from 38.4*V* to 57.6*V*, though $|Z_{iL}(s)|$ is intersected with $|Z_{oS}(s)|$, $|Z_{iLS}(s)|$ is always larger than $|Z_{oS}(s)|$. Similarly, the Bode plots of the experimental system at Case I when the load



Figure 5.13: The experimental system.

is varied during 10% full load (10*W*) and 100% full load (100*W*) are presented in Figs. 5.14(c) and (d). It can be noted that, when p_o varies from 10*W* to 100*W*, the ASVI control strategy can ensure a total separation between $|Z_{oS}(s)|$ and $|Z_{iLS}(s)|$ as well.

The experimental results in Case I of the cascaded system are given in Fig. 5.15. Fig. 5.15(a) shows the unstable cascaded system can be regulated to a stable system by the ASVI control strategy. In addition, by Figs. 5.15(b) and (c), the cascaded system can work well with the ASVI control strategy whether during input voltage changing or load changing process. One might ask if the ASVI control strategy cannot find its right centre frequency, what will happen to the cascaded system. Is the oscillation of the cascaded system serious or mitigated? To answer these questions, two definitions are firstly defined: 1) the wrong ASVI control strategy — the ASVI control strategy with a wrong centre frequency; 2) the right ASVI control strategy — the ASVI control strategy with a right centre frequency. Fig. 5.16 presents the simulation waveforms in Case I of the cascaded system with the wrong and right ASVI control strategies. In Fig. 5.16, the cascaded system works without ASVI control strategy before 0.2s. During [0.2s, 0.25s], the cascaded system works with a wrong ASVI control strategy. After 0.25 s, the cascaded system works with a right ASVI control strategy. Three conclusions can be obtained from Fig. 5.16: 1) the cascaded system is unstable without the ASVI control strategy; 2) even if the ASVI does not find its right centre frequency, it will not make the system more unstable, but still can suppresses the



Figure 5.14: Bode plots in Case I of the experimental system with different working conditions: (a) without the ASVI control strategy, $v_{bus} : 38.4V \rightarrow 57.6V$, $p_o = 100W$; (b) with the ASVI control strategy, $v_{bus} : 38.4V \rightarrow 57.6V$, $p_o = 100W$; (c) without the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow 100W$; (d) with the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow 100W$; (d) with the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow 100W$; (d) with the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow 100W$; (d) with the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow 100W$; (d) with the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow 100W$.



Figure 5.15: Experimental waveforms in Case I of the cascaded system: (a) steady state waveforms with rated input voltage and rated load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated to 100% rated load at rated input voltage.



Figure 5.16: Simulation waveforms in Case I of the cascaded system with the wrong and right ASVI control strategies.

system oscillation; 3) if the ASVI finds its right centre frequency, the cascaded system can become stable.

Similarly, both the Bode plots and experimental results in Cases II & III are given in Figs. 5.17 ~ 5.18 and Figs. 5.20 ~ 5.21, respectively. In both cases, the ASVI control strategy can ensure a total separation of $|Z_{oS}(s)|$ and $|Z_{iLS}(s)|$ and make the whole system work well both at steady-state and in dynamic conditions. It is worth pointing out that compared to the experimental results in Chapter 4, in Cases II and III, the APVI method requires 45.56 ms and 32.2 ms to stabilise the unstable system, respectively, but the ASVI method needs only 20.08 ms and 8.79 ms. Moreover, in Cases II and III, the ASVI control method reduces both the overshoot and regulation time of the load converter when its input voltage and load changing. Therefore, the ASVI control strategy is superior to the APVI control strategy.

The simulation waveforms in Cases II and III with the wrong and right ASVI control strategies are also presented at Figs. 5.19 and 5.22, respectively. In both cases, the cascaded system is unstable without the ASVI control strategy. However, even if the ASVI does not find its right centre frequency, the ASVI control strategy will not make the system more



Figure 5.17: Bode plots in Case II of the experimental system with different working conditions: (a) without the ASVI control strategy, v_{bus} : 38.4V \rightarrow 57.6V, $p_o = 100W$; (b) with the ASVI control strategy, v_{bus} : 38.4V \rightarrow 57.6V, $p_o = 100W$; (c) without the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W; (d) with the ASVI control strategy, $v_{bus} = 48V$, p_o : 10W \rightarrow 100W.



Figure 5.18: Experimental waveforms in Case II of the cascaded system: (a) steady state waveforms with rated input voltage and rated load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated to 100% rated load at rated input voltage.



Figure 5.19: Simulation waveforms in Case II of the cascaded system with the wrong and right ASVI control strategies.

unstable, but can still suppress the system oscillation. In addition, when the ASVI finds the right centre frequency, the cascaded system becomes stable.

According to Fig. 5.14 ~ Fig. 5.21, the ASVI control strategy can not only stabilise the load converter cascaded with a *LC* input filter, but also the load converter cascaded with a Buck converter. These experimental results also can be explained by the theory. Fig. 5.10 illustrates that the ASVI controller $G_{ASVI}(s)$ needs only the information of i_{bus} and v_{bus} to shape the input impedance of the load converter, thus, it is independent of the internal structure of the source converters. Therefore, no matter the source converter is a *LC* input filter or Buck converter, the same information (i_{bus} and v_{bus}) is provided to $G_{ASVI}(s)$ to help the ASVI control strategy stabilise the whole system. The above experimental results also show that the ASVI control strategy is feasible and effective in practice.

5.5 Summary

SAC and SPC, two typical load shaping stabilisation methods, are analysed and compared in this chapter. This analysis reveals that although both SAC and SPC can stabilise the



Figure 5.20: Bode plots in Case III of the experimental system with different working conditions: (a) without the ASVI control strategy, $v_{bus} : 38.4V \rightarrow 57.6V$, $p_o = 100W$; (b) with the ASVI control strategy, $v_{bus} : 38.4V \rightarrow 57.6V$, $p_o = 100W$; (c) without the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow 100W$; (d) with the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow 100W$; (d) with the ASVI control strategy, $v_{bus} = 48V$, $p_o : 10W \rightarrow 100W$.



Figure 5.21: Experimental waveforms in Case III of the cascaded system: (a) steady state waveforms with rated input voltage and rated load, (b) dynamic waveforms when the input voltage steps down from 100% rated to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% rated to 100% rated load at rated input voltage.



Figure 5.22: Simulation waveforms in Case III of the cascaded system with the wrong and right ASVI control strategies.

cascaded system, the system utilising the SAC is unconditionally stable, but conditionally stable when utilising the SPC. The cascaded system with SAC is therefore more stable than that with SPC. In addition, as the SAC can be realised by both the PVI and SVI control strategies, the PVI/SVI control strategies are also compared here. The results shows that the PVI control strategy has inevitable limitations when realising the SAC during the whole load and input voltage range of the load converter, but that the SVI control strategy does not have limitations. Therefore, the SVI control strategy is more suitable for the SAC than the PVI control strategy. Moreover, in order to stably connect the load converter to different source converters, such as *LC* input filters and traditional DC/DC converters, without changing its internal structure, an ASVI control strategy is further proposed in this chapter. This proposed ASVI control strategy allows the load converter can shape its input impedance to stabilise the cascaded system with different source converters. Finally, the ASVI strategy has been experimentally verified on a cascaded system composed by a load converter and three different source converters.

Chapter 6

Improved ASVI Control with Minimum Ripple Point Tracking for the Load Converter

The ASVI control strategy is an attractive load stabilisation method for cascaded systems thanks to its adaptive capacity for different source converters and better performance for load converters. However, as discussed in Chapters 4 and 5, this ASVI control strategy has a potential problem in that, as it utilises a proportional-integral (PI) controller to find the centre frequency of the ASVI, the PI controller may miss it with excessive proportional or integral coefficients, leading to the failure of the ASVI control strategy. Although this potential problem may not arise with small proportional and integral coefficients, it is a potential issue for the ASVI control strategy. To make the ASVI control strategy more reliable, a minimum-ripple-point-tracking (MRPT) controller is proposed in this chapter to replace the original PI controller. With the MRPT controller, the centre frequency of ASVI is found via the perturb and observe (P&O) algorithm, which ensures the ASVI control strategy does not have the potential for this problem. Furthermore, in order to quickly find the centre frequency of ASVI, the sinusoidal-tracking-algorithm (STA) is further incorporated into the MRPT controller to improve its processing speed. Finally, a load converter cascaded with different LC input filters is fabricated to validate the effectiveness of the proposed MRPT controller.

6.1 The ASVI Control Strategy and the Motivation of the MRPT Controller

6.1.1 Review of the ASVI Control Strategy

According to Chapter 5, there are three basic purposes of the ASVI control strategy: (a) make a total separation between $|Z_{iL}|$ and $|Z_{oS}|$ (See Fig. 6.1(a)); (b) only increase $|Z_{iL}|$ during a very small frequency range (See Fig. 6.1(a)); (c) change $|Z_{iL}|$ adaptively according to different source converters. Each of these purposes brings its own benefit to the ASVI control strategy. The first purpose ensures the cascaded system works as an unconditionally stable system whose system loop gain Z_{oS}/Z_{iL} always stays inside the unit cycle as shown in Fig. 6.1(b). The second purpose guarantees that the ASVI control strategy can stabilise the cascaded system with minimised load performance compromise. The final purpose is for the load converter to stably connect to different source converters without changing its internal structure. The above benefits make the ASVI control strategy the more competitive among the existing stabilisation methods of the cascaded system.

In theory, the ASVI control strategy can be realised by adding an adaptive virtual impedance Z_{ASVI} into the series with the load input port. As shown in Fig. 6.1(c), the improved load input impedance Z_{iLS} can be expressed as

$$Z_{iLS} = Z_{ASVI} + Z_{iL} \tag{6.1}$$

where $Z_{iL} = -V_{bus}^2/P_o$. Here, V_{bus} and P_o are the input voltage and output power of the load converter, respectively.

According to Fig. 6.1(a), Z_{iLS} should satisfy:

$$If \{ f \in [f_1, f_2] \} \rightarrow |Z_{iLS}| > |Z_{oS}|$$

$$(6.2)$$

$$If \{ f \notin [f_1, f_2] \} \rightarrow Z_{iLS} = Z_{iL}$$

$$(6.3)$$

where f_1 and f_2 are the intersection frequencies of $|Z_{iL}|$ and $|Z_{oS}|$ as shown in Fig. 6.1(a).

As observed in Chapter 5, in order to ensure that $|Z_{iL}|$ successfully becomes $|Z_{iLS}|$,



Figure 6.1: Purposes and concept of the ASVI control strategy: (a) purposes of the ASVI control strategy — described by Bode plots; (b) purposes of the ASVI control strategy — described by Nyquist curve; (c) concept of the ASVI control strategy.

 Z_{ASVI} can be selected as

$$Z_{ASVI}(s) = -\frac{2K_r (2\pi f_{RC})s}{s^2 + 2(2\pi f_{RC})s + (2\pi f_{CS})^2}$$
(6.4)

where f_{CS} is the cut-off frequency of the source converter and is also the centre frequency of Z_{ASVI} . By Fig. 6.1(a), f_{CS} is very close to f_1 and f_2 . f_{RC} is the bandwidth at -3dBcut-off frequency of the $Z_{ASVI}(s)$. During $(f_{CS} - f_{RC}, f_{CS} + f_{RC})$, $|Z_{ASVI}(s)| = K_r/\sqrt{2}$. As noted in Chapter 5, both f_{RC} and K_r are flexible determined by the user and their initial values are recommended as $5H_z$ and 100Ω , respectively. The characteristics of Z_{ASVI} are also presented in Fig. 6.1(c).

According to (6.4) and Fig. 6.1(c), Z_{ASVI} has three characteristics: (a) when $f \in [f_1, f_2]$, $|Z_{ASVI}|$ is large enough to ensure a total separation of $|Z_{iLS}|$ and $|Z_{oS}|$; (b) when $f \notin [f_1, f_2]$, $|Z_{ASVI}|$ is small enough to minimise the impact of the ASVI control strategy on the original load converter; (c) when the source converters changes, $|Z_{ASVI}|$ only needs to regulate its centre frequency to the new f_{CS} to adapt to this change. Therefore, conceptually, the ASVI control strategy indeed can be realised perfectly by adding Z_{ASVI} to the series, with the load input impedance shown in Fig. 6.1(c).

It is clear that realising Z_{ASVI} is the key point of the ASVI control strategy. As discussed in Chapter 5, this Z_{ASVI} can be realised by adding an impedance regulator $G_{ASVI}(s)$ to the original load converter. 6.2 illustrates that $G_{ASVI}(s)$ samples the input voltage and current of the load converter and adds its output to the output of the voltage regulator $G_v(s)$. Here, $G_{ASVI}(s)$ can be expressed as

$$G_{ASVI}(s) = Z_{ASVI}(s) \cdot \frac{[1 + T_v(s)]P_o}{G_{id}(s)G_M(s)V_{bus}^2}$$
(6.5)

where $T_{\nu}(s)$ is the loop gain of the voltage closed-loop of the original load converter; $G_{id}(s)$ is the control to the input current transfer function of the original load converter; and $G_M(s)$ is the modulator transfer function of the original load converter.

To ensure $G_{ASVI}(s)$ can realise $Z_{ASVI}(s)$ and especially can find f_{CS} adaptively, a specific control circuit for $G_{ASVI}(s)$ is proposed in Chapter 5. As shown in Fig. 6.3(a), this control circuit contains five parts and operates as follows.

In Part A, v_{bus} first goes through a high-pass-filter (HPF) and an absolute value block to extract $|\triangle v_{bus}|$. Following this, $|\triangle v_{bus}|$ goes through Parts B and C to find f_{CS} adaptively. As shown in Figs. 6.3(a) and (b), the adaptive mechanism of Parts B and C can be explained as: in Part B, $|\triangle v_{bus}|$ is firstly compared with 0, whilst, its error is amplified by a PI controller. Therefore, the output of the PI controller, PI_{out} , can be expressed as:

$$PI_{out}(t) = K_p \left| \triangle v_{bus}(t) \right| + K_i \int_0^t \left| \triangle v_{bus}(t) \right| dt$$
(6.6)

where K_p and K_i are the proportional and integral coefficients of PI controller, respectively.

According to (6.6), if the system is unstable, $|\triangle v_{bus}| > 0$, PI_{out} is increased from zero. By Fig. 6.3(a), PI_{out} also decides the centre frequency of $Z_{ASVI}(s)$. Therefore, once PI_{out} is increased to f_{CS} , $Z_{ASVI}(s)$ will find its right centre frequency, stabilise the cascaded system and make $|\triangle v_{bus}| = 0$. Then, as observed in Part C and Fig. 6.3(b), the input 3 of S_1 is equal to $0 \rightarrow$ the output of S_1 becomes $0 \rightarrow$ the output of *Relay* changes to $1 \rightarrow$ the input



Figure 6.2: Realisation of the ASVI control strategy via adding $G_{ASVI}(s)$.



Figure 6.3: Realisation of the $G_{ASVI}(s)$: (a) control circuit of $G_{ASVI}(s)$; (c) adaptive mechanism of $G_{ASVI}(s)$.

2 of S_1 changes to $1 \rightarrow$ the output of S_1 is changed to the input 1 of S_1 and becomes 0 again \rightarrow the output of *Relay* is locked as 1. Therefore, the output of S_2 , $S2_{Out}$, is always equal to its previous value

$$S2_{Out}(n+1) = S2_{Out}(n) = f_{CS}$$
(6.7)

where n + 1 and n represent the time n + 1 and its previous time n, respectively. Both of them are after PI_{out} finding f_{CS} .

According to Fig. 6.3 and (6.7), as $S2_{Out}$ is equal to f_{CS} before the relay is locked, the centre frequency of $Z_{ASVI}(s)$ is finally locked as f_{CS} and will not be changed further. At the same time, i_{bus} is sent to Part D to realise $\frac{[1+T_v(s)]P_o}{G_{id}(s)G_{PWM}(s)V_{bus}^2}$. Then, both the outputs of Parts C and D are sent to Part E, where the $Z_{ASVI}(s)$ can be achieved. Finally, the output of $G_{ASVI}(s)$ can be obtained from the output of Part E.

Note that in the ASVI control strategy, the PI controller plays an important role in helping $G_{ASVI}(s)$ find f_{CS} adaptively.

6.1.2 Potential Problem of the ASVI Control Strategy

As described in Section 6.1.1, the PI controller is the key controller for helping the ASVI control strategy to find the centre frequency of ASVI. However, as discussed in Chapter 5, if K_p or K_i is too large, the PI controller may miss the right frequency and cause the failure of the ASVI control strategy. This is a potential problem of the ASVI control strategy.

As shown in Fig. 6.4(a), when K_p and K_i are too large, this can cause PI_{out} to increase to such a degree during a sampling period that f_{CS} is skipped. Unfortunately, for the ASVI control strategy, if f_{CS} is skipped, $G_{ASVI}(s)$ is unable to find it again. In other words, the ASVI control strategy fails with large K_p and K_i .

The detailed failure mechanism is explained in Fig. 6.4(b). As noted, if PI_{out} exceeds f_{CS} , $Z_{ASVI}(s)$ misses its right centre frequency, the cascaded system is still unstable and $|\triangle v_{bus}| > 0$. According to the control Part C of Fig. 6.3(a), the input 3 of S_1 is larger than $0 \rightarrow$ the output of S_1 is larger than $0 \rightarrow$ the output of Relay is equal to $0 \rightarrow$ the input 2 of S_1 is equal to $0 \rightarrow$ the output of S_1 is still equal to the input 3 of S_1 and larger than $0 \rightarrow$ the output of Relay is always larger than $0 \rightarrow$ the output of S_2 is always equal to the input 3 of S_2 at this moment can



Figure 6.4: The potential problem and its existing prevention method of the ASVI control strategy: (a) miss f_{CS} with excessive PI control parameters; (b) failure mechanism; (c) existing prevention method in Chapter 5 — using small K_p and K_i .

be expressed as

$$S2_{Out} = PI_{out} > f_{CS} \tag{6.8}$$

According to (6.6), as $|\triangle v_{bus}| > 0$, PI_{out} will keep continue to increase due to its integral function. By (6.8), $S2_{Out}$ will also continue to increase. As $S2_{Out}$ also provides the centre frequency for $Z_{ASVI}(s)$, the centre frequency of $Z_{ASVI}(s)$ will be farther away from f_{CS} and never return, as shown in Fig. 6.4(b).

Therefore, the PI controller may miss the right centre frequency of $Z_{ASVI}(s)$ with large K_p and K_i and lead to the failure of the ASVI control strategy. To avoid this problem, small K_p and K_i are recommended in Chapter 5 to prevent the permanent unstable situation. As shown in Fig. 6.4(c), if the K_p and K_i are small enough, the PI controller can always help $Z_{ASVI}(s)$ to find the right centre frequency. This is also the only existing method for



Figure 6.5: Experimental system and its failure Cases of the ASVI control strategy with the same PI controller in Chapter 5: (a) experimental system; (b) Case I; (c) Case II.

prevention the potential problem of the ASVI control strategy.

6.1.3 The Motivation of the MRPT Controller

As noted in Section 6.1.2, although small K_p and K_i may prevent the fault of the ASVI control strategy, this is only a temporary solution. This is because the engineer or user cannot have a clear concept of K_p and K_i for the ASVI controls strategy, i.e., how small is small, or how large is large. Different cascaded systems give different answers.

In order to further elaborate on the limitations of the prevention method proposed in Chapter 5, a 120 W, 48 V - 24 V / 50 kHz Buck converter cascaded with two different *LC* input filters is fabricated. As illustrated in Fig. 6.5(a), if the Buck converter is connected to the filter with $L_f = 2mH$, $C_f = 100 \mu F$, the system is referred to Case I. If the Buck converter is connected to the filter with $L_f = 3mH$, $C_f = 150 \mu F$, the system is referred to Case II. Since this cascaded system is unstable at both Cases I and II, the ASVI control strategy is applied into such unstable system and selects $K_p = 10$, $K_i = 20$ for its PI controller, whose parameters are the same as the PI parameters in Chapter 5, and already proved to be small enough in their original experimental systems. However, as shown in Figs. 6.5(b) and (c), the K_p and K_i are still too large for the example system in Fig. 6.5(a). Whether the example system is working in Case I or Case II, the ASVI controller has failed to stabilise the cascaded system. Note the dashed lines in Figs. 6.5(b) and (c): $|\Delta v_{bus}|$ first decreases and then increases, after enabling the ASVI control strategy. It clearly shows that in this example system, $K_p = 10$, $K_i = 20$ are indeed too large for the PI controller, which causes the *PI*_{out} to close, miss and be farther away from the right centre frequency of $Z_{ASVI}(s)$.

Therefore, in practice, the engineer should spend a lot of time testing the PI parameters many times to find a sufficiently small one for the ASVI control strategy. Clearly, the proposed prevention method in Chapter 5 increases the time cost for the cascaded system, and reduces the competitiveness of the ASVI control strategy. Since the PI controller is the root of the potential problem of the ASVI control strategy and changing the PI parameters is merely a palliative solution, the PI controller could instead by replaced by another reliable controller. It is for this reason that the current chapter proposes the MRPT controller.

6.2 The Improved ASVI Control Strategy with the MRPT Controller

6.2.1 The Relationship Between the Centre Frequency of $Z_{ASVI}(s)$ and $|\triangle v_{bus}|$ — the Inspiration of the MRPT Controller

As noted in Section 6.1.2, the potential problem of the ASVI control strategy is its PI controller. It may miss the right centre frequency of $Z_{ASVI}(s)$ and cause the failure of the ASVI control strategy. Therefore, the relationship between the centre frequency of $Z_{ASVI}(s)$ and $|\triangle v_{bus}|$ is carefully analysed in this section. For convenience, the centre frequency of $Z_{ASVI}(s)$ is hereinafter referred to as f_o .

To reveal the relationship between f_o and $|\triangle v_{bus}|$ clearly, the Bode plots of Z_{iL} , Z_{iLS} and Z_{oS} of a specific cascaded system with different f_o are depicted in Figs. 6.6(a) and (b). Here, this example system is the same as the experimental system in Fig. 6.5(a).

Figs. 6.6(a) and (b) are the Bode plots of this system in Cases I and II, respectively. For both cases, it is only when $f_o = f_{CS}$, that $|Z_{iL}|$ is increased by $Z_{ASVI}(s)$ correctly and a total separation is made between $|Z_{iLS}|$ and $|Z_{oS}|$. Otherwise, whether f_o is increased or decreased, as long as f_o is far away from f_{CS} , $|Z_{iLS}|$ and $|Z_{oS}|$ are intersected and their intersection becomes increasingly serious.

It is understood that the more serious the intersection between $|Z_{iLS}|$ and $|Z_{oS}|$, the larger the oscillation bus voltage of the cascaded system which occurs (Wildrick, 1993). Therefore, Figs. 6.6(a) and (b) suggest that $|\triangle v_{bus}|$ will increase when f_o is far away from f_{CS} . In order to verify this idea, the measured $\frac{|\triangle v_{bus}|}{V_{bus}}$ versus $\frac{f_o}{f_{CS}}$ of the above experimental system are also presented in Fig. 6.6(c). Whether at Case I or Case II, the minimum $|\triangle v_{bus}|$ always appears at $f_o = f_{CS}$ and $|\triangle v_{bus}|$ increases when f_o is far away from f_{CS} .

It is worth pointing out that, Fig. 6.6(c) actually presents a very significance phenomenon, i.e., the $\frac{|\triangle v_{bus}|}{V_{bus}}$ vs $\frac{f_o}{f_{CS}}$ curve behaves like an inverted power-voltage curve of a PV cell (Liu et al., 2008). As the maximum power point of the PV cell can be found by the maximum power point tracking (MPPT) controller, it seems likely that the minimum $|\triangle v_{bus}|$ of the cascaded system could be found by a similar MPPT controller. By Fig. 6.6(c), if the minimum $|\triangle v_{bus}|$ is found, the cascaded system also becomes stable. This is the initial inspiration for the proposed MRPT controller.

6.2.2 Basic Idea of the MRPT Controller

In Fig. 6.6(c), a MRPT controller is proposed to help the ASVI control strategy find f_{CS} instead of the original PI controller. With this MRPT controller, the ASVI control strategy could eliminate the potential problem caused by the original PI controller. The basic principle of the MRPT controller is described in this section.

Fig. 6.7(a) illustrates that the MRPT controller actually learns the perturb and observe (P&O) algorithm (Liu et al., 2008) from the MPPT controller, with f_o is perturbed by a small increment in the MRPT controller, and the resulting change of $|\triangle v_{bus}|$ measured. Where the resulting change of $|\triangle v_{bus}|$ is negative, the perturbation of f_o moves $|\triangle v_{bus}|$ closer to the minimum ripple point (MRP). Thus, further f_o perturbations in the same direction (that is, with the same algebraic sign) should move $|\triangle v_{bus}|$ toward the MRP. If the resulting change of $|\triangle v_{bus}|$ is positive, $|\triangle v_{bus}|$ has been moved away from the MRP, and



Figure 6.6: Impact of f_o change on the ASVI control strategy: (a) Bode plots of Z_{iL} , Z_{iLS} and Z_{oS} at Case I; (b) Bode plots of Z_{iL} , Z_{iLS} and Z_{oS} at Case II; (c) the relationship between f_o and $|\Delta v_{bus}|$.



Figure 6.7: Basic idea and algorithm flowchart of the MRPT controller: (a) basic idea; (b) algorithm flowchart.

the algebraic sign of the perturbation should be reversed to move back toward the MRP. Fig. 6.7(b) illustrates this algorithm in a detailed flowchart. Here, $|\triangle v_{bus}(k-1)|$ and $|\triangle v_{bus}(k)|$ are $|\triangle v_{bus}|$ at time k-1 and k respectively. $f_o(k-1)$, $f_o(k)$ and $f_o(k+1)$ are f_o at time k-1, k and k+1, respectively. Between time k and time k+1, the MRPT controller works as follows:

$$If \{ |\triangle v_{bus}(k)| > |\triangle v_{bus}(k-1)| \& f_o(k) > f_o(k-1) \} \to f_o(k+1) = f_o(k) - \triangle f_o \quad (6.9)$$

$$If \{ |\triangle v_{bus}(k)| > |\triangle v_{bus}(k-1)| \& f_o(k) < f_o(k-1) \} \to f_o(k+1) = f_o(k) + \triangle f_o \quad (6.10)$$

$$If \{ |\triangle v_{bus}(k)| < |\triangle v_{bus}(k-1)| \& f_o(k) > f_o(k-1) \} \to f_o(k+1) = f_o(k) + \triangle f_o \quad (6.11)$$

$$If \{ |\triangle v_{bus}(k)| < |\triangle v_{bus}(k-1)| \& f_o(k) < f_o(k-1) \} \to f_o(k+1) = f_o(k) - \triangle f_o \quad (6.12)$$

where $\triangle f_o$ is the perturbation size of f_o . In order to avoid the oscillation caused by the P&O algorithm (Liu et al., 2008), a smaller $\triangle f_o$ is recommended in the MRPT controller.

As indicated by (6.9) \sim (6.12), even when the output of the MRPT controller is increased rapidly and skips f_{CS} , it is still able to return and find it. Therefore, the MRPT



Figure 6.8: Improvement of the MRPT controller: (a) slow processing speed of the MRPT controller; (b) improvement of the MRPT controller; (c) control block of STA.

controller indeed eliminates the potential problem of the original ASVI control strategy.

6.2.3 Improvement of the MRPT Controller with Sinusoidal-Tracking-Algorithm (STA)

As observed in Section 6.2.2, a smaller $\triangle f_o$ is needed in the MRPT controller to avoid the oscillation caused by the P&O algorithm. However, a smaller $\triangle f_o$ also means a slower processing speed of the MRPT controller as shown in Fig. 6.8(a). In order to solve this problem, the MRPT controller is further improved with STA in this section.

It is understood that if the system is unstable due to the intersection between $|Z_{iLS}|$ and $|Z_{oS}|$, the fundamental frequency of $\triangle v_{bus}$ will be very close to f_{CS} (Zhang et al., 2015; Wildrick, 1993). Therefore, if the fundamental frequency of $\triangle v_{bus}$ can be pre-measured

and pre-set as the initial frequency of f_o before enabling the MRPT controller, it can reduce much of the processing time for the MRPT controller to find the f_{CS} . This is the key point to improve the MRPT controller. For convenience, the fundamental frequency of $\triangle v_{bus}$ and the initial frequency of f_o are hereafter referred to as f_{osf} and f_{oi} , respectively.

As shown in Fig. 6.8(b), the above idea can be realised simply by a HPF and a STA: v_{bus} first goes through the HPF to extract $\triangle v_{bus}$, then the STA (Zhong and Hornik, 2013) measures f_{osf} from $\triangle v_{bus}$. f_{osf} is then set as f_{oi} . Therefore, before starting the MRPT controller, its output is already very close to f_{CS} , which can greatly shorten the processing time of the MRPT controller to find the f_{CS} . As a result, the MRPT controller is indeed improved by a fast processing speed.

As discussed before, the STA plays an important role in improving the MRPT controller. Therefore, its detailed operational principle and design criteria are specially explained as follow.

First, the STA is essentially a phase-locked-loop (PLL), whose objective and estimated signals are $\triangle v_{bus}(t)$ and its fundamental component e(t), respectively. $\triangle v_{bus}(t)$ and e(t) are assumed to be

$$\triangle v_{bus}(t) = \sum V_{mi} \sin \theta_{gi} + n(t)$$
(6.13)

$$e(t) = E(t)\sin\theta(t)$$

= $E(t)\sin(\int_0^t \omega(\tau)d\tau + \delta(t))$ (6.14)

where V_{mi} and θ_{gi} are the amplitude and phase of the *i*-th harmonic component of $\triangle v_{bus}(t)$; n(t) represents the noise on $\triangle v_{bus}(t)$; E(t) and $\theta(t) = \int_0^t \omega(\tau) d\tau + \delta(t)$ are the amplitude and phase of e(t).

Then, the state vector of the STA is defined as $\psi(t) = [E(t) \omega(t) \delta(t)]^T$. So the problem of designing the STA can be formulated as finding the optimal vector $\psi(t)$ that minimises the cost function

$$J(\boldsymbol{\psi}(t),t) = d^{2}(t)$$

= $[\Delta v_{bus}(t) - e(t)]^{2}$ (6.15)

where $d(t) = \triangle v_{bus}(t) - e(t)$ is the tracking error of the STA.

In order to solve the optimisation problem, formulate

$$\frac{d\psi(t)}{dt} = -\mu \frac{\partial [J\psi(t), t]}{\partial \psi(t)} \Rightarrow \begin{cases} \frac{dE(t)}{dt} = 2\mu_1 d\sin\theta\\ \frac{d\omega(t)}{dt} = 2\mu_2 E d\cos\theta\\ \frac{d\theta(t)}{dt} = \omega + \mu_3 \frac{d\omega}{dt} \end{cases}$$
(6.16)

(6.16) is the basis for implementing the STA as shown in Fig. 6.8(c). As illustrated in (6.16) and Fig. 6.8(c), there are three control parameters in STA: μ_1 is the integral coefficient of the integral controller of the STA amplitude loop. μ_2 and μ_3 form a PI controller $\mu_2(\mu_3 + \frac{1}{s})$ of the STA frequency loop. In general, the larger μ_1 , μ_2 and μ_3 , the higher the bandwidth of STA and the faster the response. However, too large a μ_2 should be avoided (Zhong and Hornik, 2013) as it may cause instability of STA if a larger phase jump of Δv_{bus} occurs. In this chapter, μ_1 , μ_2 and μ_3 are selected as 10, 35 and 2 respectively for the experimental system in Fig. 6.5(a).

According to Fig. 6.8(c), when the frequency loop of STA enters its steady state, STA can extract f_{osf} from Δv_{bus} and also provide f_{osf} as the initial frequency for f_o . Therefore, with the STA, the improved MRPT controller not only inherits the advantages of the basic MRPT controller to eliminate the potential problem of the original ASVI control strategy, but also ensures the improved ASVI control strategy to quickly stabilise the cascaded system.

6.2.4 Realisation of the MRPT Controller

According to Section 6.1.1, the ASVI control strategy adds only an impedance regulator $G_{ASVI}(s)$ to the original control system of the load converter. Therefore, Fig. 6.9 shows the practical implementation circuit of $G_{ASVI}(s)$ with the MRPT controller. When the single-pole-double-throw switcher S_0 is connected to its input 1, STA is disabled and the circuit is for the $G_{ASVI}(s)$ with the basic MRPT controller. When S_0 is connected to its input 2, STA is enabled and the circuit is for the $G_{ASVI}(s)$ with the improved MRPT controller. As observed earlier, this circuit utilises Parts B and C to realise the function of STA and the basic MRPT controller, respectively. Also, when STA is enabled, Part B extracts the f_{osf} from Δv_{bus} and sends it into Part C to provide the initial frequency for f_o . As the detailed operational principle of the basic MRPT controller and STA were explained in Sections



Figure 6.9: Implementation circuit of $G_{ASVI}(s)$ with the MRPT controller.

6.2.2 and 6.2.3 respectively, they will not be repeated here. In Fig. 6.9, the logic of $S_1 \sim S_5$, *Relay* and *Start* are also presented.

6.3 Experimental Verification

In this section, the improved ASVI control strategy with the proposed MRPT controller are applied to the experimental system in Fig. 6.5(a). In this experiment, the sampling time step and Δf_o of the MRPT controller are 10 ms and 25 Hz, respectively. The control parameters μ_1 , μ_2 and μ_3 of the STA are 10, 35 and 2, respectively.

Firstly, the improved ASVI control strategy with the basic MRPT controller is utilised by the experimental system. The experimental waveforms in Cases I and II are presented at Figs. 6.10(a) and (b) respectively, whilst the definition of i_L , v_{bus} , i_{L1} and v_o are illustrated in Fig. 6.5(a). It is noted that, in both cases, the improved ASVI control strategy with the basic MRPT controller works well and is able to stabilise the cascaded system. In contrast to the experimental results in Figs. 6.5(b) and (c), the basic MRPT controller eliminates the potential problem of the original ASVI control strategy. However, in Figs. 6.10, its stabilisation time is a little longer. As discussed above, this is caused by the slow processing speed of the basic MRPT controller.

The improved ASVI control strategy with the improved MRPT controller utilising STA, is then applied to the experimental system. The experimental waveforms in Cases I and



Figure 6.10: Experimental waveforms at Cases I & II with the improved ASVI control strategy incorporating the basic MRPT controller: (a) Case I; (b) Case II.

II are presented at Figs. 6.11 and 6.12, respectively. In Case I, the steady state waveforms, input voltage dynamic waveforms and load dynamic waveforms of experimental systems are presented in Figs. 6.11(a), (b) and (c), respectively. According to Fig. 6.11(a), the improved MRPT controller and the improved ASVI control strategy not only stabilise the cascaded system, but also require much less stabilisation time than that of the basic MRPT controller. Additionally, Figs. 6.11(b) and (c) indicate that the cascaded system can work well with the improved ASVI control strategy incorporating the improved MRPT controller, whether during input voltage changing (v_{bus} : $48V \rightarrow 38.4V$) or load changing (p_o : $10W \rightarrow 120W$) processes. Similarly, in Case II, the steady state waveforms, input voltage dynamic waveforms and load dynamic waveforms of experimental system are presented in Figs. 6.12(a), (b) and (c), respectively. Again, it has been verified that the improved ASVI control strategy with the improved MRPT controller not only stabilises the cascaded system with a fast speed, but also works well during the dynamic processes.

These experimental results thus verified the effectiveness of the ASVI control strategy with the MRPT controller.



Figure 6.11: Experimental waveforms at Cases I with the improved ASVI control strategy incorporating the improved MRPT controller: (a) steady state waveforms with rated input voltage and load, (b) dynamic waveforms when the input voltage steps down from 100% to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% to 100% rated load at rated input voltage.



Figure 6.12: Experimental waveforms at Cases II with the improved ASVI control strategy incorporating the improved MRPT controller: (a) steady state waveforms with rated input voltage and load, (b) dynamic waveforms when the input voltage steps down from 100% to 80% rated voltage at full load, (c) dynamic waveforms when the load increases from 10% to 100% rated load at rated input voltage.

6.4 Summary

The traditional ASVI control strategy utilises a PI controller to find the centre frequency of the ASVI. However, this PI controller may miss the right frequency with excessive proportional or integral coefficients and cause the failure of the ASVI control strategy. Though this problem has been discussed in Chapters 4 and 5, it has not yet been solved, thus this chapter looks again at the problem. The relationship between the bus voltage ripple and the centre frequency of the ASVI is firstly analysed. From this, it is clear that their relationship curve behaves like an inverted power-voltage curve of a PV cell. Therefore, a P&O algorithm based MRPT controller is proposed to find the centre frequency of ASVI, in place of the original PI controller. With the MRPT controller, the ASVI control strategy can find the centre frequency of the ASVI without the above problem. Furthermore, in order to shorten the stabilisation time of the improved ASVI control strategy, the STA is also utilised to improve the processing speed of the proposed MRPT controller. In this chapter, both the potential problem of the original ASVI control strategy with the PI controller and the effectiveness of the improved ASVI control strategy with the PI controller and the effectiveness of the improved ASVI control strategy with the proposed MRPT controller are experimentally verified on an actual cascaded system.

Chapter 7

Source-side SVI (SSVI) Control to Stabilise Cascaded Systems and Improve the Source Performance

Based on the analysis in Chapters 2 to 6, the ASVI control strategy is so far the best load stabilisation method for the cascaded system. It can not only make the cascaded system an unconditional stable system by shaping the load input impedance, it also changes just the load input impedance in a very small frequency range, minimising its impact on the original load converter. However, though the ASVI control strategy has already greatly reduced its impact on the load converter, its remaining impact is negative. In order to solve this problem, this chapter moves the ASVI from the load side to the source side via a source-side SVI (SSVI) control strategy for the source converter. The proposed SSVI control strategy not only has the same stabilisation function as the ASVI control strategy, it also improves the performance of the source converter. In addition, since the SSVI control strategy is realised by changing the control block of the source converter, the performance of the load converter is not affected. Therefore, the SSVI control strategy can be treated as a supplement and expansion of the ASVI control strategy. Moreover, depending on the method of realisation, the SSVI control strategy can be divided into the source stabilisation methods of the cascaded system. It is also worth pointing out that as the SSVI control strategy stabilises the system by reducing the source output impedance, it is not only suitable for the cascaded system with one upstream converter and one downstream converter, but also for the distributed power system with multiple upstream converters and one downstream



Figure 7.1: Distributed power system with multiple upstream converters and one down-stream converter.

converter as shown in Fig. 7.1.

In this chapter, the impact of the ASVI control strategy on the load converter and the impact of the SSVI control strategy on the source converter are analysed via two-port network analysis. The concept and realisation approach of the SSVI control strategy are also discussed in detail. The remaining parts of this chapter are organised as follows: the impact of the ASVI control strategy on the load converter is analysed via two-port network analysis in Section 7.1. The SSVI control strategy is then proposed and its impact on the source converter discussed in Section 7.2. Following this, a 100 W, 48 V - 32 V - 24 V cascaded system is experimentally employed to verify the effectiveness of the SSVI control strategy in Section 7.3. Section 7.4 then concludes this chapter.

7.1 Impact of the ASVI Control Strategy on the Load Converter via Two-Port Network Analysis

In this section, the generic two-port network models of the load converter with and without the ASVI control strategy are first derived. Then, the impact of the ASVI control strategy on the load converter is analysed based on a specific cascaded system.



Figure 7.2: Two-port network model of the original load converter: (a) block diagram; (b) two-port network model.

7.1.1 Two-Port Network Model of the Original Load Converter

In Fig. 7.2(a), the small signal block diagram of the original load converter is presented, presenting v_{bus} and i_{bus} as the input voltage and current of the load converter, respectively. v_o and i_o are the output voltage and current of the load converter, respectively. d_L is the duty cycle of the load converter. $Z_{iOL}(s)$, $G_{iOL}(s)$, $G_{vOL}(s)$, $Z_{oOL}(s)$ and $G_{vdL}(s)$ are the six basic open-loop transfer functions of the load converter, whose definitions are referred to Fig. 7.2(a). In addition, for the original load converter, a voltage closed-loop is utilised, where $H_{sL}(s)$ and $G_{cL}(s)$ are the sampling coefficient and voltage regulator transfer function of the output voltage, respectively, and $G_{ML}(s)$ is the transfer function of the modulator.

According to Fig. 7.2(a), the generic two-port network model of the original load converter is derived and presented in Fig. 7.2(b), where $Z_{iL}(s)$ is the closed loop input impedance; $G_{iL}(s)$ is the closed loop load to input current transfer function; $G_{vL}(s)$ is the closed loop output to output voltage transfer function; $Z_{oL}(s)$ is the closed loop output impedance. The performance of the original load converter can be fully described by $Z_{iL}(s) \sim Z_{oL}(s)$



Figure 7.3: Two port network model of the load converter with the ASVI control strategy: (a) block diagram; (b) two-port network model.

(Arnedo, 2008), whose expressions can be derived as

$$Z_{iL}(s) = \left. \frac{\hat{v}_{bus}(s)}{\hat{i}_{bus}(s)} \right|_{\hat{i}_o(s)=0} = \left[\frac{1}{Z_{iOL}(s)} \frac{1}{1 + T_{vL}(s)} - \frac{T_{vL}(s)}{1 + T_{vL}(s)} \left(\frac{1}{Z_{iOL}(s)} - \frac{G_{idL}(s)G_{vOL}(s)}{G_{vdL}(s)} \right) \right]^{-1}$$
(7.1)

$$G_{iL}(s) = \frac{\hat{i}_{bus}(s)}{\hat{i}_{o}(s)} \bigg|_{\hat{v}_{bus}(s)=0} = G_{iOL}(s) - \frac{Z_{oOL}(s)G_{idL}(s)}{G_{vdL}(s)} \frac{T_{vL}(s)}{1 + T_{vL}(s)}$$
(7.2)

$$G_{\nu L}(s) = \frac{\hat{v}_o(s)}{\hat{v}_{bus}(s)}\Big|_{\hat{i}_o(s)=0} = \frac{G_{\nu L}(s)}{1 + T_{\nu L}(s)}$$
(7.3)

$$Z_{oL}(s) = -\frac{\hat{v}_o(s)}{\hat{i}_o(s)}\Big|_{\hat{v}_{bus}(s)=0} = \frac{Z_{oOL}(s)}{1+T_{vL}(s)}$$
(7.4)

where $T_{vL}(s) = H_{sL}(s)G_{cL}(s)G_{ML}(s)G_{vdL}(s)$ is the loop gain of the voltage closed-loop of the original load converter.
7.1.2 Two-Port Network Model of the Load Converter With the ASVI Control Strategy

According to Chapter 5, in order to stabilise the cascaded system, the ASVI control strategy adds an adaptive virtual impedance $Z_{ASVI}(s)$ in series with $Z_{iL}(s)$ to avoid the intersection between $|Z_{oS}(s)|$ and $|Z_{iL}(s)|$. As shown in Fig. 7.3(a), this $Z_{ASVI}(s)$ is actually realised by introducing an impedance regulator $G_{ASVI}(s)$ to the control block of the load converter. The expression of $G_{ASVI}(s)$ is:

$$G_{ASVI}(s) = Z_{ASVI}(s) \cdot \frac{[1 + T_{\nu L}(s)]P_o}{G_{idL}(s)G_{ML}(s)V_{bus}^2}$$
(7.5)

where

$$Z_{ASVI}(s) = \begin{cases} Z_{iLS}(s) - Z_{iL}(s) & f \in [f_1, f_2] \\ 0 & f \notin [f_1, f_2] \end{cases} \approx \frac{-2K_r (2\pi f_{RC}) s}{s^2 + 2(2\pi f_{RC}) s + (2\pi f_{CS})^2}$$
(7.6)

here, $Z_{iLS}(s)$ is the improved $Z_{iL}(s)$ by the ASVI control strategy. $|Z_{iLS}(s)| > Z_{oSP}$ during (f_1, f_2) and $Z_{iLS}(s) = Z_{iL}(s)$ outside (f_1, f_2) . $f_{RC} = Max[(f_{CS} - f_1), (f_2 - f_{CS})]$ is the bandwidth at -3dB cut-off frequency of the $Z_{ASVI}(s)$. $|Z_{ASVI}(s)|$ during $(f_{CS} - f_{RC}, f_{CS} + f_{RC})$ is equal to $\frac{K_r}{\sqrt{2}}$ and also is equal to $|Z_{iLS}(j2\pi f_{CS}) - Z_{iL}(j2\pi f_{CS})|$.

According to Fig. 7.3(a), the generic two-port small signal model of the load converter with SVI control strategy is presented in Fig. 7.3(b). $Z_{iLS}(s)$, $G_{iLS}(s)$, $G_{vLS}(s)$ and $Z_{oLS}(s)$ are the improved $Z_{iL}(s)$, $G_{iL}(s)$, $G_{vL}(s)$ and $Z_{oL}(s)$ with the ASVI control strategy, respectively. Their expressions can be derived as

$$Z_{iLS}(s) = \frac{\hat{v}_{bus}(s)}{\hat{i}_{bus}(s)}\Big|_{\hat{i}_o(s)=0} = Z_{iL}(s) + Z_{ASVI}(s)$$
(7.7)

$$G_{iLS}(s) = \frac{\hat{i}_{bus}(s)}{\hat{i}_o(s)}\Big|_{\hat{v}_{bus}(s)=0} = G_{iL}(s) \cdot \left(1 - \frac{T_{iL}(s)}{1 + T_{vL}(s)}\right)^{-1}$$
(7.8)

$$\begin{aligned} G_{vLS}(s) &= \frac{\hat{v}_o(s)}{\hat{v}_{bus}(s)} \Big|_{\hat{i}_o(s)=0} &= \left(G_{vOL}(s) + \frac{G_{vdL}(s)}{Z_{iOL}(s)G_{idL}(s)} \frac{T_{iL}(s)}{1 - T_{iL}(s)} \right) \left(1 + \frac{T_{vL}(s)}{1 - T_{iL}(s)} \right)^{-1} (7.9) \\ Z_{oLS}(s) &= -\frac{\hat{v}_o(s)}{\hat{i}_o(s)} \Big|_{\hat{v}_{bus}(s)=0} &= \left(Z_{oOL}(s) + \frac{G_{vdL}(s)G_{iOL}(s)}{G_{idL}(s)} \frac{T_{iL}(s)}{1 - T_{iL}(s)} \right) \left(1 + \frac{T_{vL}(s)}{1 - T_{iL}(s)} \right)^{-1} (7.9) \end{aligned}$$

where $T_{iL}(s) = G_{ASVI}(s)G_{idL}(s)G_{ML}(s)$.

According to (7.7), the ASVI control strategy adds $Z_{ASVI}(s)$ to the series with $Z_{iL}(s)$ to



Figure 7.4: Example system with the ASVI control strategy.

]	[abl	le 7.1	: Pa	aramet	ers	of the	e ez	kampl	e systen	n witl	h the ASV	[control stra	tegy
v_{in}	48V	V ₀	24V	f _{sS}	20kHz	L_S	1 <i>m</i> H	L_L	450 д Н	H_{sS}, H_{sL}	0.1	$C = 5^{+100}$	C 2722-9	$(s + 6.356e^3)^2$
v_{bus}	32V	P_o	100W	f_{sL}	100kHz	C_S	150µF	C_L	220 <i>µ</i> F	G_{MS}, G_{ML}	1/2.34	$O_{cS} = 3 + \frac{1}{s}$	$G_{cL} = 5.725e$ ·	$\overline{s+6.283e^5}$

form the final input impedance of the load converter $Z_{iLS}(s)$. By (7.6), $|Z_{iLS}(s)|$ is always larger than $|Z_{oS}(s)|$, thus, the ASVI control strategy can ensure the system stability.

However, according to (7.8) ~ (7.10), though the ASVI control strategy shapes $Z_{iL}(s)$ to ensure the system stability, all the transfer functions of the load converter have already been changed by it. Here, in order to show the changed transfer functions clearly, all of the changed transfer functions are also marked by shadow areas in Fig. 7.3(b).

7.1.3 Impact of the ASVI Control Strategy on the Load Converter

According to Sections 7.1.1 and 7.1.2, though the ASVI control strategy can stabilise the cascaded system by shaping $Z_{iL}(s)$, it in fact changes all the transfer functions of the original load converter. In order to evaluate the impact of ASVI control strategy on the load converter clearly, a detailed cascaded system is taken as an example to aid the analysis in this section. The main circuit and control block of this example system are depicted in Fig. 7.4, and the parameters of the example system are presented in Table 7.1.

According to (7.1) and (7.7), Bode plots of $Z_{iL}(s)$ and $Z_{iLS}(s)$ of the example system are depicted in Fig. 7.5(a). These suggest that the ASVI control strategy can increase $|Z_{iL}(s)|$ during (f_1, f_2) to ensure a total separation between $|Z_{iLS}(s)|$ and $|Z_{oS}(s)|$. Therefore, the stability of the cascaded system is guaranteed. However, it is worthy pointing out that, since the negative resistor characteristic of $Z_{iL}(s)$ is also changed by the ASVI control strategy during (f_1, f_2) (See Fig. 7.5(a)), the ASVI control strategy actually has negative impact on $Z_{iL}(s)$ from the load performance point of view. According to (7.2) and (7.8), Bode plots of $G_{iL}(s)$ and $G_{iLS}(s)$ in the example system can be depicted and presented in Fig. 7.5(b). As seen, $G_{iLS}(s)$ is different with $G_{iL}(s)$ during (f_1, f_2) , however, the difference is very small and can be ignored.

According to (7.3) and (7.9), Bode plots of $G_{vL}(s)$ and $G_{vLS}(s)$ in the example system can be plotted and presented in Fig. 7.5(c). These show that $|G_{vLS}(s)|$ is larger than $|G_{vL}(s)|$ at a lower frequency. Since $G_{vL}(s)$ reflects the suppression ability of v_o to the disturbance of v_{bus} and the smaller $|G_{vL}(s)|$ the better, the ASVI control strategy has a negative impact on $G_{vL}(s)$.

According to (7.4) and (7.10), Bode plots of $Z_{oL}(s)$ and $Z_{oLS}(s)$ in the example system can be plotted and presented in Fig. 7.5(d). Similar to $G_{vLS}(s)$, $|Z_{oLS}(s)|$ is larger than $|Z_{oL}(s)|$ at a lower frequency. As a larger $|Z_{oL}(s)|$ also means a worse dynamic performance of v_o to i_o , the ASVI control strategy has a negative impact on $Z_{oL}(s)$.

In summary, though the ASVI control strategy can stabilise the cascaded system by shaping $Z_{iL}(s)$, this affects all the transfer functions and deteriorates the performance of $Z_{iL}(s)$, $G_{\nu L}(s)$ and $Z_{oL}(s)$. This is the limitation of the ASVI control strategy.

7.2 The SSVI Control Strategy

7.2.1 The Concept of the SSVI Control Strategy

As shown in Fig. 7.6, the ASVI control strategy adds $Z_{ASVI}(s)$ to the series with $Z_{iL}(s)$, to form the final load input impedance $Z_{iLS}(s)$ and ensure $|Z_{iLS}(s)|$ is always larger than $|Z_{oS}(s)|$ to stabilise the cascaded system. However, as discussed above, the ASVI control strategy deteriorates the performance of the load converter. In order to solve this problem, the $Z_{ASVI}(s)$ is moved from load side to source side, thus changing the control block of the source converter instead of the load converter in this section. To distinguish it from the ASVI control strategy, this proposed solution is called the 'SSVI control strategy'.

As depicted in Fig. 7.6, since $Z_{ASVI}(s)$ is in a series with $Z_{iL}(s)$ and $Z_{oS}(s)$ simultaneously, the SSVI control strategy has the same stabilisation function as the ASVI control strategy. The only difference between the SSVI and ASVI control strategies is in their physical concepts, i.e., the ASVI control strategy keeps a total separation between $|Z_{oS}(s)|$ and $|Z_{iL}(s)|$ by increasing $|Z_{iL}(s)|$, but the SSVI control strategy realises this by decreasing



Figure 7.5: Bode plots of the load converter with or without the ASVI control strategy: (a) $Z_{iL}(s)$ and $Z_{iLS}(s)$; (b) $G_{iL}(s)$ and $G_{iLS}(s)$; (c) $G_{\nu L}(s)$ and $G_{\nu LS}(s)$; (d) $Z_{oL}(s)$ and $Z_{oLS}(s)$.



Figure 7.6: Concept of the SSVI control strategy.

 $|Z_{oS}(s)|$. However, in later sections, it will be shown that, this small difference helps the SSVI control strategy to overcome the drawback of the ASVI control strategy, i.e., not deteriorating but in fact improving the performance of the source converter without bringing any impact to the load converter when stabilising the cascaded system.

7.2.2 Realisation of the SSVI Control Strategy

The small signal control block of the original source converter is presented in Fig. 7.7(a). v_{in} and i_{in} are the input voltage and current of the source converter, respectively. d_S is the duty cycle of the source converter. $Z_{iOS}(s) \sim G_{MS}(s)$ have the same definitions as $Z_{iOL}(s) \sim G_{ML}(s)$ in Fig. 7.2(a), where the only difference is the subscript *S* denotes 'source converter'. As with the dotted line in Fig. 7.7(a), the intuitive way to add $Z_{ASVI}(s)$ to a series with $Z_{oS}(s)$ is by introducing $Z_{ASVI}(s)$ between \hat{i}_{bus} and $-\hat{v}_{bus}$ directly. However, this method cannot be achieved by control directly. In order to address this issue, the output of $Z_{ASVI}(s)$ moves to the output of the voltage regulator $G_{cS}(s)$ and equivalently adjusts the transfer function to $G_{SSVI}(s)$, as shown by the dot-dashed lines.

According to Fig. 7.7(a), if $G_{SSVI}(s)$ is required to realise $Z_{ASVI}(s)$, the following relationship is in force:

$$-\hat{i}_{bus} \cdot Z_{ASVI}(s) = \hat{i}_{bus} \cdot G_{SSVI}(s) \cdot G_{MS}(s) \cdot G_{vdS}(s) \cdot \frac{1}{1 + T_{vS}(s)}$$
(7.11)

where $T_{vS}(s) = H_{sS}(s)G_{cS}(s)G_{MS}(s)G_{vdS}(s)$ is the loop gain of the voltage closed-loop of the original source converter.



Figure 7.7: The SSVI control strategy: (a) control block; (b) algorithm.



Figure 7.8: Two-port network model of the source converter with the SSVI control strategy.

According to (7.11), the expression of $G_{SSVI}(s)$ can be derived as

$$G_{SSVI}(s) = -Z_{ASVI}(s) \cdot \frac{1 + T_{\nu S}(s)}{G_{\nu dS}(s)G_{MS}(s)}$$
(7.12)

Up to this point, Fig. 7.7(a) presents the fully controlled block of the proposed SSVI control strategy.

The proposed SSVI control strategy can be implemented by a digital signal processor, such as DSP TMS320F28335. The mechanism for this is presented in Fig. 7.7(b), with both v_{bus} and i_{bus} sampled to provide the necessary information for the SSVI control strategy and the output voltage regulation. The algorithm obtains the modulation signal by adding the output of $G_{SSVI}(s)$ to the output of $G_{cS}(s)$, and then generates d_S by pulse-width modulation (PWM).

7.2.3 Impact of the SSVI Control Strategy on the Source Converter

According to Fig. 7.7(a), the generic two-port small signal model of the source converter with the SSVI control strategy can be derived and presented in Fig. 7.8. $Z_{iS}(s)$, $G_{iS}(s)$, $G_{vS}(s)$ and $Z_{oS}(s)$ are the four basic input-to-output closed-loop transfer functions of the original source converters. $Z_{iSS}(s)$, $G_{iSS}(s)$, $G_{vSS}(s)$ and $Z_{oSS}(s)$ are the improved $Z_{iS}(s)$, $G_{iS}(s)$, $G_{vS}(s)$ and $Z_{oS}(s)$ by the SSVI control strategy, respectively. Their expressions can be derived as

$$Z_{iSS}(s) = \frac{\hat{v}_{in}(s)}{\hat{i}_{in}(s)}\Big|_{\hat{i}_{bus}(s)=0} = \left[\frac{1}{Z_{iOS}(s)}\frac{1}{1+T_{vS}(s)} - \frac{T_{vS}(s)}{1+T_{vS}(s)}\left(\frac{1}{Z_{iOS}(s)} - \frac{G_{idS}(s)G_{vOS}(s)}{G_{vdS}(s)}\right)\right]^{-1} = Z_{iS}(s)$$
(7.13)

$$G_{iSS}(s) = \frac{\hat{i}_{in}(s)}{\hat{i}_{bus}(s)}\Big|_{\hat{v}_{in}(s)=0} = \begin{bmatrix} G_{iOS}(s) - \frac{Z_{oOS}(s)G_{idS}(s)}{G_{vdS}(s)} \frac{T_{vS}(s)}{1 + T_{vS}(s)} \end{bmatrix} + \begin{bmatrix} \frac{T_{iS}(s)}{1 + T_{vS}(s)} \end{bmatrix}$$

$$= G_{vs}(s) + G_{vs}(s)$$
(7.14)

$$= G_{iS}(s) + G_{iSa}(s)$$
(7.14)
$$G_{vSS}(s) = \frac{\hat{v}_{bus}(s)}{\hat{s}_{c}(s)} = \frac{G_{vOS}(s)}{1 + T_{c}(s)} = G_{vS}(s)$$
(7.15)

$$Z_{oSS}(s) = -\frac{\hat{v}_{bus}(s)}{\hat{i}_{bus}(s)}\Big|_{\hat{v}_{in}(s)=0} = \frac{Z_{oOS}(s)}{1+T_{vS}(s)} + Z_{ASVI}(s) = Z_{oS}(s) + Z_{ASVI}(s)$$
(7.16)

where $T_{iS}(s) = G_{SSVI}(s)G_{idS}(s)G_{MS}(s)$.

According to (7.16), the SSVI control strategy adds $Z_{ASVI}(s)$ into the series with $Z_{oS}(s)$



Figure 7.9: Example system with SSVI control strategy.

to form the final output impedance of the source converter $Z_{oSS}(s)$. By Fig. 7.6, the SSVI control strategy realises the same stabilisation function with the ASVI control strategy, so the SSVI control strategy can guarantee the stability of the cascaded system. According to (7.13) and (7.15), the SSVI control strategy keeps the original performance of $Z_{iS}(s)$ and $G_{vS}(s)$. According to (7.14), the SSVI control strategy changes the original performance of $G_{iS}(s)$. In Fig. 7.8, the above changed transfer functions are marked by dash borders.

As with the ASVI control strategy, in order to evaluate the impact of the SSVI control strategy on the source converter, the same example system shown in Fig. 7.4 is utilised to aid the analysis in this section. The only difference is that this example system utilises the SSVI control strategy instead of the ASVI control strategy. The main circuit and control block of the example system with the SSVI control strategy are depicted in Fig. 7.9. The parameters of the example system can be found in Table 7.1.

According to (7.13) and (7.15), Bode plots of $Z_{iS}(s)$ & $Z_{iSS}(s)$ and $G_{vS}(s)$ & $G_{vSS}(s)$ of the example system are depicted in Figs. 7.10(a) and (c), respectively. It can be observed that the SSVI control strategy does not make any changes of $Z_{iS}(s)$ and $G_{vS}(s)$.

According to (7.14), Bode plots of $G_{iS}(s) \& G_{iSS}(s)$ of the example system can be plotted and presented in Fig. 7.10(b). The SSVI control strategy increases the amplitude of $G_{iS}(s)$ at a higher frequency. As $G_{iS}(s)$ reflects the suppression ability of i_{in} to the disturbance of i_{bus} and the smaller $|G_{iS}(s)|$ the better, the SSVI control strategy has a negative impact on $G_{iS}(s)$. However, it is worth mentioning that since the SSVI control strategy only increases $|G_{iS}(s)|$ on the high frequency range, the impact is minimal in practice.

According to (7.16), Bode plots of $Z_{oS}(s)$ & $Z_{oSS}(s)$ of the example system are depicted in Fig. 7.10(d). The SSVI control strategy decreases $|Z_{oS}(s)|$ in the whole frequency range



Figure 7.10: Bode plots of the source converter without or with the SSVI control strategy: (a) $Z_{iS}(s)$ and $Z_{iSS}(s)$; (b) $G_{iS}(s)$ and $G_{iSS}(s)$; (c) $G_{vS}(s)$ and $G_{vSS}(s)$; (d) $Z_{oS}(s)$ and $Z_{oSS}(s)$.

Table 7.2: Comparison for ASVI and SSVI control strategie	Table 7.2:	Comparison	for ASVI	and SSVI	control strategies
---	------------	------------	----------	----------	--------------------

	Affected converter	$Z_{i(L/S)}(s)$	$G_{i(L/S)}(s)$	$G_{\nu(L/S)}(s)$	$Z_{o(L/S)}(s)$	Cascaded system	Evaluation
ASV	Load converter	Negative	Little change	Negative	Negative	Stable	SSVI is better
SSV	Source converter	No change	Little change	No change	Postitive	Stable	than ASVI

to avoid the intersection with $|Z_{iL}(s)|$. Therefore, the stability of the cascaded system is guaranteed. In addition, since $Z_{oS}(s)$ is the closed-loop output impedance of the source converter, the smaller $|Z_{oS}(s)|$ the better. Hence, the SSVI control strategy also has positive impact on $Z_{oS}(s)$ from the source performance point of view.

In summary, the SSVI control strategy can stabilise the cascaded system by reducing $|Z_{oS}(s)|$. It has no impact on $Z_{iS}(s)$ and $G_{vS}(s)$, a negligible negative impact on $G_{iS}(s)$ and a positive impact on $Z_{oS}(s)$.

According to Figs. 7.5 and 7.10, both the impact of the ASVI control strategy on the load converter and the impact of the SSVI control strategy on the source converter are concluded and compared in Table 7.2. The SSVI control strategy not only has the same stabilisation function as the ASVI control strategy, but also has a more positive impact on the original cascaded system than the ASVI control strategy. From this point of view, the SSVI control strategy is superior to the ASVI control strategy.

7.3 Experimental Results

In this chapter, the example cascaded system presented in Fig. 7.9, is fabricated to validate the SSVI control strategy. Its circuit, control block, parameters and Bode plots can be found in Section 7.2.3, and will not be repeated here. As shown in Fig. 7.11, to make the verification more available, the source converter is controlled by DSP TMS320F28335, and the load converter is a special custom power module from TRACO POWER company.

The experimental results of the cascaded system with the SSVI control strategy are given in Fig. 7.12. As shown in Fig. 7.12(a), this cascaded system is unstable without the SSVI control strategy. After enabling the SSVI control strategy, the cascaded system becomes stable. This phenomenon is also consistent with Bode plots in Fig. 7.10(d). The input voltage dynamic waveforms of the cascaded system with the SSVI control strategy at full load are presented in Figs. 7.12(b) and (c). In Fig. 7.12(b), the input voltage of the



Figure 7.11: Experimental system.

cascaded system changes from 80% rated voltage (38.4 V) to 100% rated voltage (48 V). In Fig. 7.12(c), the input voltage of the cascaded system changes from 48 V to 38.4 V. This indicates that the SSVI control strategy can work well during any input voltage changes. The load dynamic waveforms of the cascaded system with the SSVI control strategy at rated input voltage are presented in Figs. 7.12(d) and (e). In Fig. 7.12(d), the load of the cascaded system changes from 10% full load (10 W) to 100% full load (100 W). In Fig. 7.12(e), the load of the cascaded system is changing from 100 W to 10 W. This also indicates that the SSVI control strategy can work well during any load changes

The dynamic experimental waveforms of the source converter with or without the SSVI control strategy are shown in Figs. 7.13 and 7.14, respectively. It is shown that, with the SSVI control strategy, the input voltage dynamic performance of the source converter is the same as the original performance. Moreover, with the SSVI control strategy, the load dynamic performance of the source converter is better than the original performance of the source of the source of the an impact on $Z_{iS}(s)$ and $G_{vS}(s)$, and has a negligible negative impact on $G_{iS}(s)$.

According to Fig. 7.12 \sim Fig. 7.14, the SSVI control strategy not only stabilises the cascaded system, but also improves the performance of the source converter.

7.4 Summary

The ASVI control strategy is further investigated in this chapter. It is understood that though the ASVI control strategy can stabilise the cascaded system and limit its impact on the load converter in a very small frequency range, it brings a negative impact to the load



Figure 7.12: Experimental waveforms of the cascaded system with the SSVI control strategy: (a) steady state waveforms when $V_{in} = 48V$, $P_o = 100W$, (b) dynamic waveforms when V_{in} steps up from 38.4 V to 48 V, (c) dynamic waveforms when V_{in} steps down from 48 V to 38.4 V, (e) dynamic waveforms when P_o increases from 10 W to 100 W, (f) dynamic waveforms when P_o decreases from 100 W to 10 W.



Figure 7.13: Dynamic waveforms of the original source converter without the SSVI control strategy: (a) V_{in} steps up from 38.4 V to 48 V, (b) V_{in} steps down from 48 V to 38.4 V, (c) P_o increases from 10 W to 100 W, (d) P_o decreases from 100 W to 10 W.



Figure 7.14: Dynamic waveforms of the source converter with the SSVI control strategy: (a) V_{in} steps up from 38.4 V to 48 V, (b) V_{in} steps down from 48 V to 38.4 V, (c) P_o increases from 10 W to 100 W, (d) P_o decreases from 100 W to 10 W.

converter. In order to solve this problem, the SSVI control strategy is proposed. The SSVI control strategy moves the ASVI from load side to source side by changing the control block of the source converter instead of the load converter. As a result, the SSVI control strategy not only has the same stabilisation function as the ASVI control strategy, but also improves the performance of the source converter. In addition, since the SSVI control strategy is realised by changing the control block of the source converter, the performance of the load converter is not affected. Therefore, the SSVI control strategy is an improvement on the ASVI control strategy. It is also worth pointing out that as the SSVI control strategy stabilises the system by reducing the source output impedance, it is not only suitable for the cascaded system with one upstream converter and one downstream converter, but also for the distributed power system with multiple upstream converters and one downstream converter. Finally, the proposed SSVI strategy has been experimentally verified on a 100 W, 48 V - 32 V - 24 V cascaded system.

Chapter 8

Virtual *RLC* Damper for the *LC* Input Filter to Stabilise Cascaded Systems and Improve the Filter Performance

As discussed in Chapters 2 to 7, the *LC* filter at the input of a DC/DC converter may cause instability when the converter is controlled as a constant power load (CPL). As the reduced system damping caused by the CPL is the root of the instability problem in cascaded systems (Middlebrook, 1979), the most intuitive solution is to increase the system damping by using dampers (Erickson and Maksimov, 2001; Cespedes et al., 2011; Marx et al., 2012). Three typical dampers are depicted in Fig. 8.1, where the elements in the dashed line are used for damping. According to the structure and position of the damper, the three dampers are called the *RC* parallel damper (Fig. 8.1(a)), the *RL* parallel damper (Fig. 8.1(b)) and the *RL* series damper (Fig. 8.1(c)), respectively. For the system shown in Fig. 8.1(d) with these dampers, the peak of the output impedance of the *LC* input filter can be damped and a total separation between $|Z_o|$ and $|Z_{iL}|$ ensured to fulfil Middlebrook stability criterion (Middlebrook, 1979). Here, Z_o is the output impedance of the damped *LC* input filter and Z_{iL} is the input impedance of the CPL.

Although these filter dampers can stabilisze the cascaded system, their impact on the *LC* input filter is always ignored and rarely reported. In this chapter, their impact was carefully analysed via two-port network analysis, and it was found that all these dampers degrade the original *LC* input filter performance to some extent. To solve this problem, a *RLC* damper is proposed which could not only can stabilise the system whilst improving



Figure 8.1: Typical dampers and their application: (a) *RC* parallel damper; (b) *RL* parallel damper; (c) *RL* series damper; (d) application.

the performance of the *LC* input filter, but also achieve high robustness to the parameter variations of the *LC* input filter. Based on the *RLC* damper, a virtual *RLC* (*VRLC*) damper is further proposed to avoid the power loss of the passive components by changing the control block of the CPL. The actual effectiveness of this *VRLC* damper and its impact on the CPL are also discussed. It should be stressed that although the *VRLC* damper is similar to the PVI control strategy in Chapter 2, they are fundamentally different. The PVI control strategy stabilises the system by shaping the load input impedance, without considering the source performance. However, the *VRLC* damper stabilises the system by shaping the proposed *VRLC* damper is experimentally verified by a 100 W 48 V - 24 V Buck converter with a *LC* input filter.

The remaining parts of this chapter are organised as follows: in Section 8.1, the existing dampers and their impact on the *LC* input filter are discussed. The *RLC* damper is proposed in Section 8.2, where its design consideration, advantages/disadvantages and impact on the *LC* input filter are also analysed. Following that, the *RLC* damper is virtual realised in Section 8.3, where the control and effectiveness of the *VRLC* damper are also discussed. In Section 8.4, both the existing dampers and proposed *VRLC* damper are utilised as an actual cascaded system, whose experimental results verify the effectiveness of the *VRLC* damper. Finally, Section 8.5 concludes the chapter.

8.1 Existing Dampers and Their Impact on the *LC* Input Filter

8.1.1 Review of the Existing Dampers

In this section, the existing dampers are reviewed under the same design principle: $6 dB\Omega$ system gain margin (GM) (Wildrick, 1993).

8.1.1.1 RC Parallel Damper

According to (Erickson and Maksimov, 2001), Fig. 8.1(a) and $6 dB\Omega$ GM stability requirement, the components C_1 and R_1 of the *RC* parallel damper can be designed as

$$C_1 = n_1 C_f \tag{8.1}$$

$$R_1 = R_{of} \cdot \sqrt{\frac{(2+n_1) \cdot (4+3n_1)}{2 \cdot n_1^2 \cdot (4+n_1)}}$$
(8.2)

where

$$R_{of} = \sqrt{\frac{L_f}{C_f}} \tag{8.3}$$

$$n_{1} = \frac{R_{of}^{2}}{\left(Z_{iL}/10^{\frac{6}{20}}\right)^{2}} \left\{ 1 + \sqrt{1 + 4\frac{\left(Z_{iL}/10^{\frac{6}{20}}\right)^{2}}{R_{of}^{2}}} \right\}$$
(8.4)

8.1.1.2 RL Parallel Damper

According to (Erickson and Maksimov, 2001), Fig. 8.1(b) and $6dB\Omega$ GM stability requirement, the components L_2 and R_2 of the *RL* parallel damper can be designed as

$$L_2 = n_2 L_f \tag{8.5}$$

$$R_2 = R_{of} \cdot \sqrt{\frac{n_2 \cdot (3 + 4n_2) \cdot (1 + 2n_2)}{2 \cdot (1 + 4n_2)}}$$
(8.6)

where

$$n_2 = \frac{1}{4} \left\{ \sqrt{1 + 4 \frac{\left(Z_{iL}/10^{\frac{6}{20}}\right)^2}{R_{of}^2}} - 1 \right\}$$
(8.7)

8.1.1.3 *RL* Series Damper

According to (Erickson and Maksimov, 2001), Fig. 8.1(c) and $6 dB\Omega$ GM stability requirement, the components L_3 and R_3 of the *RL* series damper can be designed as

$$L_3 = n_3 L_f \tag{8.8}$$

$$R_3 = R_{of} \cdot \left[\frac{(1+n_3)}{n_3} \cdot \sqrt{\frac{2 \cdot (1+n_3) \cdot (4+n_3)}{(2+n_3) \cdot (4+3n_3)}} \right]^{-1}$$
(8.9)

where

$$n_{3} = \left[\sqrt{1+4 \cdot \frac{\left(Z_{iL}/10^{\frac{6}{20}}\right)^{2}}{R_{of}^{2}}} + 3\right] \cdot \left\{\frac{\left(Z_{iL}/10^{\frac{6}{20}}\right)^{2}}{R_{of}^{2}} - 2\right\}^{-1}$$
(8.10)

All the above dampers can ensure a total separation of $|Z_o|$ and $|Z_{iL}|$ to stabilise the system (Erickson and Maksimov, 2001; Cespedes et al., 2011). However, their impact on the *LC* input filter has not been analysed sufficiently in the existing literature. For this reason, the impact of existing dampers on the *LC* input filter is carefully analysed here using the two-port network model in Sections 8.1.2 and 8.1.3.

		-
<i>LC</i> input filter	A_x ———	B_x ————
Original (<i>x</i> =0)		
<i>RC</i> parallel damper ($x=1$)		
<i>RL</i> parallel damper (x=2)		
<i>RL</i> series damper (x=3)		
Proposed damper (x=4)		

Table 8.1: Definitions of $A_x(s)$ and $B_x(s)$ in different dampers

8.1.2 Two-Port Network Model of the *LC* Input Filter with Different Dampers

In Fig. 8.2(a), the generic two-port network circuit model of the *LC* input filter with different dampers is presented, where v_{in} and v_{bus} are the filter input and output voltage, respectively; i_{in} and i_{bus} are the filter input and output current, respectively; and $A_x(s)$ and $B_x(s)$ represent the impedance of the filter inductor branch and capacitor branch, respectively. A further description of $A_x(s)$ and $B_x(s)$ is presented in Table 8.1. In Fig. 8.2(b), the two-port block diagram of Fig. 8.2(a) is given, where $Z_{ox}(s)$ is the output impedance; $1/Z_{ix}(s)$ is the input voltage to input current transfer function; $G_{vx}(s)$ is the input to output voltage transfer function; $G_{ix}(s)$ is the load to input current transfer function. Here, the subscript $x = 0 \sim 4$ denotes with no damper, *RC* parallel damper, *RL* parallel damper, *RL* series damper and the latter proposed damper, respectively. As is known, the performance of the *LC* input filter can be fully described by $Z_{ox}(s) \sim G_{ix}(s)$, which can be derived as

$$Z_{ox}(s) = -\frac{\hat{v}_{bus}(s)}{\hat{i}_{bus}(s)}\Big|_{\hat{v}_{in}(s)=0} = \frac{A_x(s) \cdot B_x(s)}{A_x(s) + B_x(s)}$$
(8.11)

$$1/Z_{ix}(s) = \frac{\hat{i}_{in}(s)}{\hat{v}_{in}(s)}\Big|_{\hat{i}_{bus}(s)=0} = \frac{1}{A_x(s) + B_x(s)}$$
(8.12)

$$G_{vx}(s) = \frac{\hat{v}_{bus}(s)}{\hat{v}_{in}(s)}\Big|_{\hat{i}_{bus}(s)=0} = \frac{B_x(s)}{A_x(s) + B_x(s)}$$
(8.13)

$$G_{ix}(s) = \frac{\hat{i}_{in}(s)}{\hat{i}_{bus}(s)}\Big|_{\hat{v}_{in}(s)=0} = \frac{B_x(s)}{A_x(s) + B_x(s)}$$
(8.14)

It is worth noting that according to (8.13) and (8.14), $G_{vx}(s)$ and $G_{ix}(s)$ have the same expression.



Figure 8.2: Generic two-port network model of the *LC* input filter with different dampers: (a) circuit model; (b) block diagram.

8.1.3 Impact of Existing Dampers on the *LC* Input Filter via Two-port Network Analysis

According to Section 8.1.2, the performance of the LC input filter can be fully evaluated by two-port network analysis with four transfer functions: $Z_{ox}(s)$, $1/Z_{ix}(s)$, $G_{vx}(s)$ and $G_{ix}(s)$, where $Z_{ox}(s)$ reflects the impact of the (load) current i_{bus} on the (load) voltage v_{bus} ; $1/Z_{ix}(s)$ reflects the impact of (input) voltage variations of v_{in} on the (input) current i_{in} ; $G_{vx}(s)$ reflects the impact of (input) voltage variations of v_{in} on the (load) voltage v_{bus} ; $G_{ix}(s)$ reflects the impact of the (load) current i_{bus} on the (input) current i_{in} . However, all the existing dampers are only focused on how to damp $|Z_{ox}(s)|$ to ensure the system stability while ignoring their other impact on $Z_{ox}(s)$, $1/Z_{ix}(s)$, $G_{ix}(s)$ and $G_{vx}(s)$. Therefore, in order to analyse the impact of the existing dampers on the LC input filter, the existing dampers impact on $Z_{ox}(s)$, $1/Z_{ix}(s)$, $G_{vx}(s)$ and $G_{ix}(s)$ are thoroughly analysed in this section. Moreover, in order to show the results in a clear way, a specific LC input filter is taken as an example to facilitate the analysis. The parameters of the example LC input filter and its dampers are given in Table 8.2, where the existing dampers are designed, according to (Erickson and Maksimov, 2001), with $6 dB\Omega$ system GM. In addition, For the CPL, its Z_{iL} is equal to $-V_{bus}^2/P_o$, where P_o is the output power of the CPL and its value is also listed in Table 8.2.

According to (8.11), Table 8.1 and Table 8.2, the Bode plots of $Z_{ox}(s)$ with different



Figure 8.3: Impact of different dampers on the *LC* input filter: (a) $Z_{ox}(s)$; (b) $1/Z_{ix}(s)$; (c) $G_{vx}(s) \& G_{ix}(s)$.

PARM.	Value	PARM.	Value	PARM.	Value	PARM.	Value
V_{in}	48 V	C_{f}	50 μF	L_2	1.5 <i>m</i> H	L_4	1.9 <i>m</i> H
V_{bus}	48 V	R_1	6.5 Ω	R_3	1.7 Ω	C_4	27 μF
P_o	100 W	C_1	60 <i>µ</i> F	L_3	1 <i>m</i> H		
L_{f}	1 <i>m</i> H	R_2	6.5 Ω	<i>R</i> ₄	11.5 Ω		

Table 8.2: Parameters of the example *LC* input filter with different dampers

dampers are depicted in Fig. 8.3(a). It can be seen that the peak of $|Z_{ox}(s)|$ can be damped by all the dampers to ensure $6dB\Omega$ GM stability of the cascaded system. However, the *RL* series damper increases $|Z_{ox}(s)|$ at the lower frequency range. Since a higher $|Z_{ox}(s)|$ means a poorer suppression of v_{bus} to the disturbance of i_{bus} , the *RL* series damper has a worse impact on $Z_{ox}(s)$.

According to (8.12), Table 8.1 and Table 8.2, the Bode plots of $1/Z_{ix}(s)$ with different dampers are presented in Fig. 8.3(b). It is shown that, though the peak of $|1/Z_{ix}(s)|$ can be damped by all the dampers, the *RC* parallel and *RL* parallel dampers increase $|1/Z_{ix}(s)|$ at the lower and higher frequency ranges, respectively. Since a higher $|1/Z_{ix}(s)|$ means a worse suppression ability of i_{in} to the disturbance of v_{in} , both *RC* parallel and *RL* parallel dampers have a worse impact on $1/Z_{ix}(s)$.

By (8.13) and (8.14), since $G_{vx}(s)$ and $G_{ix}(s)$ have the same expression, they are depicted by the same Bode plots in Fig. 8.3(c).

When $G_{vx}(s)$ has a higher cut-off frequency, v_{bus} has a quick dynamic response to v_{in} . Hence, the higher the cut-off frequency of $G_{vx}(s)$, the better. According to Fig. 8.3(c), the cut-off frequency of $G_{vx}(s)$ is increased when adopting the *RL* parallel damper, but reduced when adopting the *RC* parallel or *RL* series dampers. Therefore, the *RL* parallel damper has better impact, but the *RC* parallel and *RL* series dampers have a worse impact on $G_{vx}(s)$.

For $G_{ix}(s)$, since it reflects the suppression capability of i_{in} to the harmonic of i_{bus} , it determines the resistance to electromagnetic interference (EMI) of the *LC* input filter (Erickson and Maksimov, 2001). From this perspective, the lower the cut-off frequency of $G_{ix}(s)$ and the smaller the $|G_{ix}(s)|$, the better. According to Fig. 8.3(c), the resonant peak of $|G_{ix}(s)|$ can be damped by all the dampers, the cut-off frequency and amplitude of $G_{ix}(s)$ is reduced by both *RC* parallel and *RL* series dampers. However, the *RL* parallel damper increases the cut-off frequency and amplitude of $G_{ix}(s)$. Therefore, both *RC* parallel and *RL* series dampers have a better impact, though the *RL* parallel damper has worse impact

Damper Impact	$Z_{ox}(s)$	$1/Z_{ix}(s)$	$G_{vx}(s)$	$G_{ix}(s)$	System stability
Without damper	Normal	Normal	Normal	Normal	Unstable
<i>RC</i> PAR.	Better	Worse	Worse	Better	Stable
<i>RL</i> PAR.	Better	Worse	Better	Worse	Stable
RL SER.	Worse	Better	Worse	Better	Stable
Proposed damper	Better	Better	Better	Better	Stable

Table 8.3: Impact on *LC* input filter with different dampers

on $G_{ix}(s)$.

The impact of the existing dampers on the LC input filter is concluded in Table 8.3. Although all the dampers are able to realise their stabilisation functions, they degrade the performance of the LC input filter to some extent. Note that, the impact of the damper to be proposed later on the LC input filter is also included in Table 8.3 for comparison.

8.2 The Proposed *RLC* Damper

8.2.1 Design Principle of the Proposed *RLC* Damper

To overcome the drawbacks of the existing dampers, a *RLC* damper is proposed, and this is added in parallel with the output of the *LC* input filter and composed by a resistor R_4 , an inductor L_4 and a capacitor C_4 (See Fig. 8.4(a)). The impedance of this *RLC* damper can be expressed as $Z_{RLC}(s) = R_4 + sL_4 + (1/sC_4)$, whose Bode plot is depicted in Fig. 8.4(b). The characteristics of $Z_{RLC}(s)$ are similar to those of a band-stop filter, whose stop-band is (f_1, f_2) : If $f < f_1$, $Z_{RLC}(s) = 1/sC_4$; If $f \in [f_1, f_2]$, $Z_{RLC}(s) = R_4$; If $f > f_1$, $Z_{RLC}(s) =$ sL_4 . Thus, the *RLC* damper only plays its damper function during $[f_1, f_2]$, in which R_4 is equivalently added in parallel with Z_{o0} . Here, $f_1 = 1/(2\pi R_4 C_4)$ and $f_2 = R_4/(2\pi L_4)$.

For the CPL and its input *LC* filter, if their system does not have enough stability GM, $|Z_{o0}(s)|$ will be intersected with $|Z_{iL}/10^{(GM/20)}|$. As $Z_{o0}(s) = sL_f/(s^2L_fC_f+1)$ and $Z_{iL}(s) = -V_{bus}^2/P_o$, the Bode plots of $Z_{o0}(s)$ and $Z_{iL}/10^{(GM/20)}$ are depicted in Fig. 8.4(c). As seen, $|Z_{o0}(s)|$ is intersected with $|Z_{iL}/10^{(GM/20)}|$ at f_L and f_H , where

$$f_L = \frac{1}{2\pi} \cdot \frac{P_o 10^{(GM/20)}}{2C_f V_{bus}^2} \left[\sqrt{1 + 4V_{bus}^4 \frac{C_f}{\left(P_o 10^{(GM/20)}\right)^2 L_f}} - 1 \right]$$
(8.15)

$$f_H = \frac{1}{2\pi} \cdot \frac{P_o 10^{(GM/20)}}{2C_f V_{bus}^2} \left[\sqrt{1 + 4V_{bus}^4 \frac{C_f}{\left(P_o 10^{(GM/20)}\right)^2 L_f}} + 1 \right]$$
(8.16)



Figure 8.4: The *RLC* damper: (a) structure; (b) characteristics; (c) design principle.

Since $Z_{o4}(s) = Z_{o0}(s) || Z_{RLC}(s)$ and according to the Figs. 8.4(b) and (c), if the system GM is required to be realised with a minimum impact of the *RLC* damper, three basic design principles of the *RLC* damper should be satisfied: (1) $R_4 = \left| \left(V_{bus}^2 \right) / \left[P_o 10^{(GM/20)} \right] \right|$; (2) $f_1 = f_{Lmin}$; (3) $f_2 = f_{Hmax}$. Here, f_{Lmin} is the minimum value of f_L , f_{Hmax} is the maximum value of f_{Hmax} . Following these design principles, the Bode plot of $Z_{o4}(s)$ is plotted with dashed lines in Fig. 8.4(c). $|Z_{o4}(s)|$ is only changed during $[f_1, f_2]$ and always lower than $\left| Z_{iL}/10^{(GM/20)} \right|$ to ensure the system GM.

8.2.2 Parameters Selection of the Proposed *RLC* Damper to Achieve High Robustness against the Variations of L_f and C_f

According to Section 8.2.1, the impedance characteristics of the *RLC* damper behave like the band-stop filter shown in Fig. 8.4(b), whose stop-band is (f_1, f_2) . As discussed above, if the *RLC* damper wants to stabilise the whole system with a minimum impact on the *LC* input filter, f_1 and f_2 should be equal to f_{Lmin} and f_{Hmax} , respectively. However, according to (8.15) and (8.16), both f_L and f_H are affected by L_f and C_f . In Figs. 8.5 (a) and (b), the curves of $\frac{f_L}{f_{LR}}$ vs $\frac{L_f}{L_{fR}}$ or $\frac{C_f}{C_{fR}}$ are presented respectively. L_{fR} and C_{fR} are the rated value of L_f and C_f respectively, f_{LR} is the value of f_L under L_{fR} and C_{fR} . f_L is monotonically



Figure 8.5: The relationship between f_L , f_H and L_f , C_f : (a) $\frac{f_L}{f_{LR}}$ vs $\frac{L_f}{L_{fR}}$; (b) $\frac{f_L}{f_{LR}}$ vs $\frac{C_f}{C_{fR}}$; (c) $\frac{f_H}{f_{HR}}$ vs $\frac{L_f}{L_{fR}}$; (d) $\frac{f_H}{f_{HR}}$ vs $\frac{C_f}{C_{fR}}$.

decreased when L_f or C_f increases. Similarly, the curves of $\frac{f_H}{f_{HR}}$ vs $\frac{L_f}{L_{fR}}$ or $\frac{C_f}{C_{fR}}$ are also presented in Figs. 8.5 (c) and (d) respectively, where f_{HR} is the value of f_H under L_{fR} and C_{fR} . It is shown that, f_H is also monotonically decreased when L_f or C_f increases.

In practice, the value of L_f and C_f are not constant, but vary according to different operation conditions, such as working frequency and temperature. It is assumed that the value of L_f or C_f varies between ($\alpha_L L_{fR} \sim \beta_L L_{fR}$) and ($\alpha_C C_{fR} \sim \beta_C C_{fR}$) respectively, then by Fig. 8.5, the expression of f_{Lmin} and f_{Hmax} can be derived as

$$f_{Lmin} = \frac{1}{2\pi} \cdot \frac{P_o 10^{(GM/20)}}{2\beta_C C_{fR} V_{bus}^2} \left[\sqrt{1 + 4V_{bus}^4 \frac{\beta_C C_{fR}}{\left(P_o 10^{(GM/20)}\right)^2 \beta_L L_{fR}}} - 1 \right]$$
(8.17)

$$f_{Hmax} = \frac{1}{2\pi} \cdot \frac{P_o 10^{(GM/20)}}{2\alpha_C C_{fR} V_{bus}^2} \left[\sqrt{1 + 4V_{bus}^4 \frac{\alpha_C C_{fR}}{\left(P_o 10^{(GM/20)}\right)^2 \alpha_L L_{fR}}} + 1 \right]$$
(8.18)

Since $f_1 = f_{Lmin}$ and $f_2 = f_{Hmax}$, by Fig. 8.4(b), (8.17) and (8.18), the parameters of the *RLC* damper can be selected as

$$L_{4} = R_{4} / \left\{ \frac{P_{o} 10^{(GM/20)}}{2\alpha_{C} C_{fR} V_{bus}^{2}} \left[\sqrt{1 + 4V_{bus}^{4} \frac{\alpha_{C} C_{fR}}{\left(P_{o} 10^{(GM/20)}\right)^{2} \alpha_{L} L_{fR}}} + 1 \right] \right\}$$
(8.19)

$$C_{4} = 1 / \left\{ \frac{R_{4}P_{o}10^{(GM/20)}}{2\beta_{C}C_{fR}V_{bus}^{2}} \left[\sqrt{1 + 4V_{bus}^{4} \frac{\beta_{C}C_{fR}}{\left(P_{o}10^{(GM/20)}\right)^{2} \beta_{L}L_{fR}}} - 1 \right] \right\}$$
(8.20)



Figure 8.6: Bode plots of $Z_{o4}(s)$ when L_f or C_f changes: (a) $L_f : 0.9mH \rightarrow 1.1mH$, $C_f = 50\mu F$; (b) $L_f = 1mH$, $C_f : 45\mu F \rightarrow 55\mu F$.

where $R_4 = \left| V_{bus}^2 / \left[P_o 10^{(GM/20)} \right] \right|.$

In order to further verify the correctness of the robust design of the *RLC* damper, (8.19) and (8.20) are substituted in the example *LC* input filter in Table 8.2 to calculate R_4 , L_4 and C_4 under the following assumptions: $\alpha_L = \alpha_C = 0.9$; $\beta_L = \beta_C = 1.1$. It can then be obtained that, $R_4 = 11.5 \Omega$, $L_4 = 1.9 \, mH$ and $C_4 = 27 \mu F$. Based on the calculated parameters, the Bode plots of $Z_{o4}(s)$ with the variations of L_f or C_f are depicted in Figs. 8.6(a) and (b), respectively. It is shown that, whether L_f or C_f vary within $0.9 \sim 1.1$ of its rated value, $|Z_{o4}(s)|$ is always lower than $|Z_{iL}/10^{(GM/20)}|$ (GM = 6dB) to realise its damping function. Therefore, the designed *RLC* damper indeed achieves high robustness against the variations of L_f and C_f .

8.2.3 The Impact of the Proposed *RLC* damper on the *LC* Input Filter

According to (8.11)~(8.14), Table 8.1 and Table 8.2, the Bode plots of $Z_{ox}(s)$, $1/Z_{ix}(s)$, $G_{vx}(s)$ and $G_{ix}(s)$ of the *LC* input filter with the proposed *RLC* damper can be depicted and presented in Fig. 8.3 as well. In contrast to the existing dampers, though the *RLC* damper has the same stabilisation function, it has better impact on all the transfer functions (e.g., $Z_{ox}(s)$, $1/Z_{ix}(s)$, $G_{vx}(s)$ and $G_{ix}(s)$) of the *LC* input filter. The impact of the proposed *RLC* damper on the *LC* input filter is recorded in Table 8.3 as ' proposed damper'. Here, it is shown that, the proposed *RLC* damper is superior to the existing dampers.

It is worth pointing out that although the proposed RLC damper can stabilise the cas-

caded system with a better performance of the *LC* input filter than the existing dampers, it has two potential risks as a high order damper: 1) the *RLC* damper increases the system order, thus, if the *RLC* damper is not designed very well and fails to stabilise the cascaded system, the system instability problem may became more serious for the increased system order; 2) the *RLC* damper needs more passive components, which may cause unacceptable power loss and volume to the cascaded system. To avoid the first potential risk, the *RLC* damper should strictly abide by the proposed design principle to ensure the cascaded system is stable. To avoid the second potential risk, the *RLC* damper can be virtually realised by the control method proposed in the following section.

8.3 Virtual Implementation of the Proposed RLC Damper

As is well known, the adoption of passive damper can lead to significant power loss. In order to avoid this problem, a control strategy is proposed in this chapter to realise the proposed *RLC* damper as a virtual *RLC* (*VRLC*) damper.

8.3.1 Concept and Implementation of the VRLC Damper

As shown in Fig. 8.4(a), as the proposed *RLC* damper must be added in parallel with output port of the *LC* input filter, it can be realised by putting a virtual *RLC* damper in parallel with the input port of the CPL via control method to avoid additional power loss. This is the initial inspiration for the *VRLC* damper.

As shown in Fig. 8.7(a), the small-signal control block of a typical CPL is presented. Its variables and transfer functions are described in Table 8.4. Obviously, one intuitive way of realising the *VRLC* damper is to introduce its admittance $1/Z_{RLC}(s)$ to the control block of the CPL between its input voltage v_{bus} and input current i_{bus} , as shown in the dark dot-dashed lines in Fig. 8.7(a). However, this method cannot be achieved by control directly, and as a result the output of $1/Z_{RLC}(s)$ is actually moved to the output voltage reference and equivalently adjusts the transfer function to $G_{RLC}(s)$, as represented by the dashed lines in Fig. 8.7(a).

According to Fig. 8.7(a), if $G_{RLC}(s)$ is required to realise the virtual RLC damper



Figure 8.7: Control strategy of the *VRLC* damper: (a) concept; (b) digital implementation; (c) different discretization effect of $G_{RLC}(s)$.

1. Variables and transfer functions of the of E
Perturbation of the bus voltage
Perturbation of the load current
Perturbation of the bus current
Perturbation of the duty cycle
Perturbation of the output voltage
Perturbation of the output voltage reference
Transfer function of the voltage regulator
Transfer function of the modulator
Open-loop input impedance
Control to output voltage transfer function
Control to input current transfer function
Open-loop output impedance
Sampling coefficient of the output voltage
Open-loop load to input current transfer function
Open-loop input to output voltage transfer function

Table 8.4: Variables and transfer functions of the CPL

 $1/Z_{RLC}(s)$, the following relationship can be obtained:

$$\hat{v}_{bus} \cdot \frac{1}{Z_{RLC}(s)} = \hat{v}_{bus} \cdot G_{RLC}(s) \cdot G_c(s) \cdot G_M(s) \cdot G_{id}(s) \cdot \frac{1}{1 + T_v(s)}$$
(8.21)

where $T_v(s) = H_s(s)G_c(s)G_M(s)G_{vd}(s)$ is the loop gain of the voltage closed loop of the CPL.

According to (8.21), the expression of $G_{RLC}(s)$ can be derived as:

$$G_{RLC}(s) = \frac{1}{Z_{RLC}(s)} \cdot \frac{1 + T_{\nu}(s)}{G_{c}(s) \cdot G_{id}(s) \cdot G_{M}(s)}$$
(8.22)

Up to this point, Fig. 8.7(a) fully presents the concept of the *VRLC* damper and shows the whole derivation process of $G_{RLC}(s)$.

In practice, the *VRLC* damper can be implemented by a digital control chip, such as a Digital Signal Processor (DSP) or a Micro-controller Unit (MCU). In Fig. 8.7(b), the digital control based small signal model of the CPL with the *VRLC* damper is presented. The digital control introduces two types of delay to the control system: the computation delay and the pulse width modulation (PWM) delay (Holmes et al., 2009). The computation delay is one sampling period in the commonly used synchronous sampling scheme, which can be modelled as $e^{-T_s s}$. The PWM delay is caused by the zero-order hold (ZOH) effect, which can be expressed as $G_h(s) = (1 - e^{-T_s s})/s \approx T_s e^{-0.5T_s s}$. Here, T_s is the sampling time of the digital control system. According to Fig. 8.7(b), the *VRLC* damper actually only adds $G_{RLC}(s)$ to the original control system of the CPL. Hence, realising $G_{RLC}(s)$ is the key to realise the *VRLC* damper. Since Backward Euler, Forward Euler and Tustin are three typical discretization methods in digital control, they are all utilised to convert $G_{RLC}(s)$ to $G_{RLC}(z)$. In order to compare their discretization effect clearly, the Bode plots of $G_{RLC}(s)$ and $G_{RLC}(z)$ with different discretization methods of a specific CPL are depicted in Fig. 8.7(c), where the example CPL corresponds to the experimental system in Section 8.4. According to Fig. 8.7(c), $G_{RLC}(z)$ with the Tustin discretization method is the closest to $G_{RLC}(s)$. Therefore, the Tustin discretization method is selected to convert $G_{RLC}(s)$ to $G_{RLC}(z)$ in the final digital control system of this chapter.

8.3.2 Effectiveness Evaluation of the *VRLC* Damper

Since the purpose of the *VRLC* damper is to mimic the *RLC* damper by changing the control block of the *CPL*, the actual effectiveness of the *VRLC* damper is evaluated carefully with the comparison of the *RLC* damper in this section.

According to Fig. 8.7(b), the input impedance of the CPL with the *VRLC* damper $Z_{iL}^V(s)$ can be derived as

$$Z_{iL}^{V}(s) = \frac{\hat{v}_{bus}(s)}{\hat{i}_{bus}(s)}\Big|_{\hat{i}_{o}(s)=0} = \left(\frac{1}{Z_{iL}(s)} + \frac{1}{Z_{VRLC}(s)}\right)^{-1}$$
(8.23)

where

$$Z_{iL}(s) = \left[\frac{1}{Z_{iOL}(s)} \frac{1}{1 + T_{vd}(s)} - \frac{T_{vd}(s)}{1 + T_{vd}(s)} \left(\frac{1}{Z_{iOL}(s)} - \frac{G_{id}(s)G_{vOL}(s)}{G_{vd}(s)}\right)\right]^{-1}$$
(8.24)

$$Z_{VRLC}(s) = (1 + T_{vd}(s)) / T_{id}(s)$$
(8.25)

$$T_{vd}(s) = (1/T_s) e^{-sT_s} G_h(s) H_s(s) G_c(s) G_M(s) G_{vd}(s)$$
(8.26)

$$T_{id}(s) = (1/T_s) e^{-sT_s} G_h(s) G_{RLC}(s) G_{id}(s) G_M(s) G_C(s)$$
(8.27)

By (8.23), the *VRLC* damper adds a virtual impedance $Z_{VRLC}(s)$ in parallel with $Z_{iL}(s)$ to mimic $Z_{RLC}(s)$ as shown in Fig. 8.8(a). Thus, if the effectiveness of the *VRLC* damper is to be evaluated, only one question need be answered: whether $Z_{VRLC}(s)$ is equal to $Z_{RLC}(s)$



Figure 8.8: Comparison of the *RLC* and *VRLC* dampers: (a) function comparison; (b) Bode plots of $Z_{RLC}(s)$ and $Z_{VRLC}(s)$.



Figure 8.9: Impact of *RLC* and *VRLC* dampers on the *LC* input filter: (a) $Z_{ox}(s)$; (b) $1/Z_{ix}(s)$; (c) $G_{vx}(s) \& G_{ix}(s)$.

or not? As a result, (8.22) is substituted by (8.25) and $Z_{VRLC}(s)$ is re-written as

$$Z_{VRLC}(s) \approx Z_{RLC}(s) \cdot \left(\frac{e^{1.5sT_s} + T_v(s)}{1 + T_v(s)}\right)$$
(8.28)

According to (8.28), if $T_s = 0$, $Z_{VRLC}(s)$ is equal to $Z_{RLC}(s)$; if $T_s \neq 0$, $Z_{RLC}(s)$ and $Z_{VRLC}(s)$ will become different near the sampling frequency, and the larger T_s , the bigger the difference. In order to compare $Z_{RLC}(s)$ and $Z_{VRLC}(s)$ clearly, the Bode plots of $Z_{RLC}(s)$ and $Z_{VRLC}(s)$ of a specific CPL are further depicted in Fig. 8.8(b), where the example CPL corresponds to the experimental system in Section 8.4. As seen, the curves of $Z_{RLC}(s)$ and $Z_{VRLC}(s)$ are indeed completely coincident when $T_s = 0$. And if $T_s \neq 0$, $Z_{RLC}(s)$ and $Z_{VRLC}(s)$ only become different near sampling frequency and their difference is bigger with the increase of T_s . The Bode plots in Fig. 8.8(b) support the accuracy of (8.28).

Therefore, the effectiveness of the *VRLC* damper is only affected by the time delay (or sampling time) of the digital control system and the smaller time delay the better. In this chapter, considering T_s cannot be smaller than the execution time of the program, $10 \mu s$ is selected as the sampling time for the experimental system. As shown in Fig. 8.8(b), when $T_s = 10 \mu s$, though $Z_{VRLC}(s)$ becomes difference with $Z_{RLC}(s)$ at high frequencies, it is the same with $Z_{RLC}(s)$ at other frequency range, especially in the damping frequency area of the *RLC* damper. Therefore, for the experimental system, the *VRLC* damper can mimic the *RLC* damper very well.

In addition, in order to compare the impact of the *RLC* and *VRLC* dampers on the *LC* input filter, the Bode plots of $Z_{ox}(s) \sim G_{ix}(s)$ of the *LC* input filter with both *RLC* and *VRLC* dampers are also depicted in Fig. 8.9. It can be seen that both *RLC* and *VRLC* dampers can improve the performance of the *LC* input filter, and that their impacts are almost the same. It is worth pointing out that, in order to compare the impact of the proposed dampers and the existing dampers on the *LC* input filter, the Bode plots in Fig. 8.9 are also presented in Fig. 8.3, which clearly shows that both *RLC* and *VRLC* dampers have a better impact than the existing dampers on the *LC* input filter. In addition, both *RLC* and *VRLC* dampers are referred to as 'proposed damper', and their impact on the *LC* input filter are also recorded as 'proposed damper' in Table 8.3.

8.3.3 The Impact of the *VRLC* Damper on the CPL

The performance of the CPL can be fully evaluated by a two-port network with four typical transfer functions (Arnedo, 2008): the closed-loop input impedance $Z_{iL}(s)$, the closed-loop output impedance $Z_{oL}(s)$, the closed-loop load to input current transfer function $G_{iL}(s)$ and the closed-loop input to output voltage transfer function $G_{vL}(s)$. Therefore, in order to evaluate the impact of the *VRLC* damper on the CPL in a clear way, the Bode plots of $Z_{iL}(s) \sim Z_{oL}(s)$ of a specific CPL with/without the *VRLC* damper are depicted in Fig. 8.10.

In Fig. 8.10, the example CPL corresponds to the experimental system in Section 8.4. Other than adding $Z_{VRLC}(s)$ in parallel with $Z_{iL}(s)$ to mimic the function of the $Z_{RLC}(s)$, the *VRLC* damper keeps most of the other performances of the original CPL. Therefore, the *VRLC* damper is also acceptable in practice.

8.4 Experimental Verification

In this chapter, experimental results from a 100 W cascaded system that consists of a *LC* input filter and a 48 V / 24 V digital controlled (sampling frequency is 100 kHz) Buck converter, are shown to validate the *VRLC* damper. The main circuit and parameters of the cascaded system are presented in Fig. 8.11(a), which also shows the parameters of the *VRLC* damper: $R_4 = 11.5\Omega$, $L_4 = 1.9 mH$, $C_4 = 27 \mu F$. When there is no dampers utilised in this system, the steady-state experimental waveforms at rated load and input voltage are given in Fig. 8.11(b). This system is unstable without any dampers.

In Fig. 8.12, the full load experimental results are given of the cascaded system with different dampers when the input voltage changing from 80% rated voltage (38.4 V) to 100% rated voltage (48 V), where i_{in} , v_{bus} , i_{L1} and v_o are defined in Fig. 8.11, v_{bus} and v_o are their ac components. Three conclusions can be obtained from Fig. 8.12. Firstly, though the cascaded system is unstable without dampers, it can be stabilised by the existing dampers or the *VRLC* damper. Thus, both the existing dampers and the *VRLC* damper have the same stabilisation function. Secondly, with these dampers, the cascaded system works well during input voltage changing. Thirdly, a distinctly better dynamic performance from the *LC* input filter is obtained by the *VRLC* damper than by the existing stabilisation dampers. The dynamic performances of the *LC* input filter with different dampers are presented and



Figure 8.10: Impact of the *VRLC* damper on the CPL: (a) $Z_{iL}(s)$; (b) $Z_{oL}(s)$; (c) $G_{iL}(s)$; (d) $G_{vL}(s)$.


Figure 8.11: The experimental system: (a) main circuit; (b) steady state experimental waveforms at rated conditions without any dampers.

further compared in Fig. 8.13, where $\sigma_{I_{in}}$ and $t_{I_{in}}$ are the overshoot and regulation time of i_{in} , respectively, and $\sigma_{V_{bus}}$ and $t_{V_{bus}}$ are the overshoot and regulation time of v_{bus} , respectively. It is further shown that when the input voltage changes, the dynamic indicators of the *LC* input filter with *VRLC* damper are superior to those of the existing stabilisation dampers. It is also worth mentioning that, for a fair comparison, in this experiment the existing dampers are all realised by passive components as reported in the original literature (Erickson and Maksimov, 2001; Cespedes et al., 2011).

Similarly, when the output power is changed from 10 W to 100 W under rated input voltage, the experimental results of the cascaded system and the dynamic indicators of the *LC* input filter with different dampers are given in Figs. 8.14 and 8.15, respectively. Again, it has been verified that when the load changes, the *VRLC* damper has a better dynamic impact on the *LC* input filter than the existing dampers.

The efficiency curves of the experimental system with different dampers vs output power and input voltage are measured and plotted in Figs. 8.16(a) and (b), respectively. It is shown that, with the *VRLC* damper, the system efficiency is clearly improved. This



Figure 8.12: Experimental waveforms with different dampers when system input voltage changed from 80% to 100% rated voltage at full load: (a) with *VRLC* damper; (b) with *RC* parallel damper; (c) with *RL* parallel damper; (d) with *RL* series damper.



Figure 8.13: Comparison of $\sigma_{I_{in}}$, $t_{I_{in}}$, $\sigma_{V_{bus}}$ and $t_{V_{bus}}$ with different dampers when system input voltage changed from 80% to 100% rated voltage at full load: (a) $\sigma_{I_{in}}$; (b) $t_{I_{in}}$; (c) $\sigma_{V_{bus}}$; (d) $t_{V_{bus}}$.



Figure 8.14: Experimental waveforms with different dampers when system output power changed from 10 W to 100 W at rated input voltage: (a) with *VRLC* damper; (b) with *RC* parallel damper; (c) with *RL* parallel damper; (d) with *RL* series damper.



Figure 8.15: Comparison of $\sigma_{I_{in}}$, $t_{I_{in}}$, $\sigma_{V_{bus}}$ and $t_{V_{bus}}$ with different dampers when system output power changed from 10 W to 100 W at rated input voltage: (a) $\sigma_{I_{in}}$; (b) $t_{I_{in}}$; (c) $\sigma_{V_{bus}}$; (d) $t_{V_{bus}}$.



Figure 8.16: System efficiency curves with different dampers: (a) efficiency vs p_o at rated v_{in} ; (b) efficiency vs v_{in} at rated p_o .

phenomenon can be explained as follows: the *VRLC* damper is realised by the control method, but the existing dampers depend on passive components, thus the *VRLC* damper avoids the power loss of the damper.

Therefore, according to Fig. $8.12 \sim$ Fig. 8.16, the *VRLC* damper can not only stabilise the whole system, but also have a better impact on the performance of the *LC* input filter and improve the system efficiency compared with existing dampers.

8.5 Summary

The impact of the existing dampers (e.g., *RC* parallel damper, *RL* parallel damper and *RL* series damper) on the *LC* input filter has been, firstly, analysed using the two-port network model described in this chapter. It was found that although the above dampers could solve the instability problem of the CPL with its *LC* input filter, they degraded the performance of the original *LC* input filter to some extent. In order to overcome this, a *RLC* damper is proposed here. With this damper, the whole system can be stabilised whilst improving the performance of the *LC* input filter. In addition, the *RLC* damper is robustly resists the parameter variations of the *LC* input filter. In order to avoid the power loss

caused by passive components, the proposed *RLC* damper is further virtually implemented by control. This indicates that the *VRLC* damper not only has the same effectiveness as the *RLC* damper, but also improves the efficiency of the whole system compared to the existing dampers. Finally, the *VRLC* damper has been experimentally verified in a 100 W cascaded system.

Chapter 9

Conclusions and Future Work

This chapter summarises the work of the whole thesis and concludes with some valuable recommendations for future work.

9.1 Conclusions

In this thesis, existing stabilisation methods for the cascaded system have been reviewed. It is highlighted that although these stabilisation methods can stabilise the cascaded system, they have unavoidable drawbacks, e.g., they do not have enough load stabilisation methods, they require all the information on the cascaded system, they degrade the original performance of the cascaded system, and so on. In order to overcome these drawbacks, a family of software-based stabilisation methods has been proposed by this thesis.

Firstly, a set of virtual-impedance-based control strategies, named PVI and SVI control strategies, are proposed. The PVI or SVI control strategy connects a virtual impedance in parallel or series with the input impedance of the load converter so that the magnitude or phase of the load input impedance is modified in a very small frequency range. As a result, with the PVI or SVI control strategy, the cascaded system can be stabilised with minimal load compromise. Therefore, both the PVI and SVI control strategies can not only be treated as developments of, and complements to the load stabilisation methods, but can also be treated as good solutions to overcome the contradiction between the stability and performance of the cascaded system.

Based on the PVI and SVI control strategies, adaptive-PVI (APVI) and adaptive-SVI (ASVI) control strategies are further proposed to incorporate the adaptive function. With

the APVI or ASVI control strategy, the load converter can be stably connected to different source converters without changing their internal structure. In this thesis, the APVI and ASVI control strategies are also carefully compared. It is found that the ASVI control strategy can transform the cascaded system into an unconditional stable system without any limitations, but the APVI control strategy cannot. As a result, the ASVI control strategy is superior to the APVI control strategy.

Though the ASVI control strategy is thus far the most competitive load stabilisation method, it is not perfect. As the ASVI control strategy utilises a PI controller to find the centre frequency of the ASVI, this PI controller may miss the right frequency with excessive proportional or integral coefficients and cause the failure of the ASVI control strategy. Although this problem may not arise with small proportional and integral coefficients, it is a hidden problem for the ASVI control strategy. To make the ASVI control strategy more reliable, a MRPT controller is proposed to replace the original PI controller. With the MRPT controller, the ASVI control strategy can find the centre frequency of the ASVI without encountering the problem. Furthermore, in order to find the centre frequency of ASVI quickly, the STA is further introduced into the MRPT controller to improve its processing speed. Therefore, the improved MRPT controller makes the improved ASVI control strategy more competitive.

In addition, stabilising the cascaded system is always in conflict with maintaining the original performance of the cascaded system. Though the proposed PVI/SVI and APVI/ASVI control strategies have already minimised their impact on the cascaded system in a very small frequency range, the rest of their impact is negative. In order to resolve this, a SSVI control strategy and a *VRLC* damper are proposed in this thesis. With the SSVI control strategy, the cascaded system can be stabilised with a better dynamic performance of the source converter. Similarly, the *VRLC* damper can stabilise the cascaded system while improving the performance of the input filter. Therefore, both the SSVI control strategy and the *VRLC* damper overcome the contradiction between the stability and performance of the cascaded system perfectly. In addition, since the SSVI control strategy stabilises the system by changing the control block of the source converter, it is not only suitable for the cascaded system with one upstream converter and one downstream converter, but also for the distributed power system with multiple upstream converters and one

downstream converter.

Therefore, the stabilisation methods proposed in this thesis overcome all of the drawbacks of the existing solutions. As a consequence, it is believed that these stabilisation methods will become increasingly popular in the near future.

9.2 Future Work

In order to further stabilise the cascaded system, there is specific research that should be conducted in the future.

- Applying the PVI/SVI/APVI/ASVI control strategies into the AC/DC rectifier and DC/AC inverter: in this thesis, the PVI/SVI/APVI/ASVI control strategies are only proposed to change the input impedance of the DC/DC converter. However, the cascaded system still has other types of load converters, such as AC/DC rectifier and DC/AC inverter. Therefore, they also need the above impedance control methods if the system is unstable. Considering the essence of the PVI/SVI/APVI/ASVI control strategies is adding a special virtual impedance in parallel or series with the input port of the load converter to shape the load input impedance, future research efforts should focus on how realising these virtual impedances on the AC/DC rectifier and DC/AC inverter.
- 2. Applying the SSVI control strategy to the AC/DC rectifier: in this thesis, the SSVI control strategy is proposed to shape the output impedance of the DC/DC converter. However, the cascaded system may sometimes use the AC/DC rectifier as the source converter. Therefore, it is preferable to realise the SSVI control strategy on the AC/DC rectifier. Considering the essence of the SSVI control strategy is adding a virtual impedance in the series with the output port of the source converter to shape the source output impedance, methods of realising this virtual impedance on the AC/DC rectifier can be considered by future research work.
- 3. Needing to reduce the program execution time of the proposed stabilisation methods. With the development of the new semiconductor devices, such as Gallium Nitride (GaN) and Silicon Carbide (SiC) switches, their switching frequency can be

greatly increased, which is the benefit of the new semiconductor devices. However, for power converters, switching frequency is also limited by the program execution time of their control methods. Therefore, if the program execution time of the proposed stabilisation methods is too great, it will limit the new semiconductor devices application in power converters.

4. Developing stabilisation methods for the three-phase AC cascaded system: in this thesis, the cascaded system refers to the cascaded DC/DC converter system. However, if the cascaded system is a three-phase AC cascaded system (i.e., its intermediate bus voltage is a three-phase AC voltage), the stability criteria and the impedance requirement of the source or load converters will be entirely different. Therefore, it is better to develop the stabilisation methods for the three-phase AC cascaded system which can also be included in the future work program.

Bibliography

- Abe, S., M. Hirokawa, M. Shoyama, and T. Ninomiya: 2008, 'Optimal intermediate bus capacitance for system stability on distributed power architecture'. In: *Proc. IEEE Power Electronics Specialists Conference, PESC*. pp. 611–616.
- Abe, S., M. Hirokawa, T. Zaitsu, and T. Ninomiya: 2006, 'Study of Stabilization Design for On-Board Distributed Power Architecture'. In: *Power Electronics and Motion Control Conference*, 2006. EPE-PEMC 2006. 12th International. pp. 636–640.
- An, L. and D. Lu: 2016, 'Analysis of DC Bus Capacitor Current Ripple Reduction in Basic DC/DC Cascaded Two-Stage Power Converters'. *IEEE Trans. Ind. Electron.* PP(99), 1–1.
- Arai, J., K. Iba, T. Funabashi, Y. Nakanishi, K. Koyanagi, and R. Yokoyama: 2008, 'Power electronics and its applications to renewable energy in Japan'. *IEEE Circuits and Systems Magazine* 8(3), 52–66.
- Arnedo, L.: 2008, 'System level black-box models for dc-dc converters'. Ph.D. thesis, Virginia Polytech. Inst. and State Univ., Blacksburg, VA.
- Blaabjerg, F., A. Consoli, J. Ferreira, and J. Van Wyk: 2005, 'The future of electronic power Processing and conversion'. *IEEE Trans. Power Electron.* 20(3), 715–720.
- Bo, Y.: 2003, 'Topology investigation for front end dc-dc power conversion for distributed power system'. Ph.D. thesis, Virginia Polytechnic Institute and State University.
- Bodi, F.: 1988, 'Distributed power systems [telecommunication power supply]'. In: Telecommunications Energy Conference, 1988. INTELEC '88., 10th International. pp. 143– 150.

- Boroyevich, D., I. Cvetkovic, D. Dong, R. Burgos, F. Wang, and F. Lee: 2010, 'Future electronic power distribution systems a contemplative view'. In: *Optimization of Electrical and Electronic Equipment (OPTIM), 2010 12th International Conference on.* pp. 1369–1380.
- Cao, X., Q. C. Zhong, and W. L. Ming: 2015, 'Ripple Eliminator to Smooth DC-Bus Voltage and Reduce the Total Capacitance Required'. *IEEE Trans. Ind. Electron.* 62(4), 2224–2235.
- Cespedes, M., L. Xing, and J. Sun: 2011, 'Constant-Power Load System Stabilization by Passive Damping'. *IEEE Trans. Power Electron.* **26**(7), 1832–1836.
- Cuk, S.: 1976, 'Modeling, analysis, and design of switching converters'. Ph.D. thesis, California Institute of Technology, Pasadena.
- Du, W., J. Zhang, Y. Zhang, and Z. Qian: 2013, 'Stability Criterion for Cascaded System With Constant Power Load'. *IEEE Trans. Power Electron.* 28(4), 1843–1851.
- Emadi, A., A. Khaligh, C. Rivetta, and G. Williamson: 2006, 'Constant power loads and negative impedance instability in automotive systems: definition, modeling, stability, and control of power electronic converters and motor drives'. *Vehicular Technology, IEEE Transactions on* 55(4), 1112–1125.
- Erickson, R. W. and D. Maksimov: 2001, *Fundamentals of power electronics*. Norwell, Mass.: Kluwer Academic.
- Feng, X., J. Liu, and F. Lee: 2002, 'Impedance specifications for stable DC distributed power systems'. *IEEE Trans. Power Electron.* **17**(2), 157–162.
- Hankaniemi, M., T. Suntio, M. Sippola, and E. Oyj: 2005, 'Characterization of Regulated Converters to Ensure Stability and Performance in Distributed Power Supply Systems'.
 In: *INTELEC 05 - Twenty-Seventh International Telecommunications Conference*. pp. 533–538.
- Heath, C. C.: 1991, 'The market for distributed power systems'. In: Applied Power Electronics Conference and Exposition, 1991. APEC '91. Conference Proceedings, 1991., Sixth Annual. pp. 225–229.

- Holmes, D., T. Lipo, B. McGrath, and W. Kong: 2009, 'Optimized Design of Stationary Frame Three Phase AC Current Regulators'. *IEEE Trans. Power Electron.* 24(11), 2417– 2426.
- Hua, G. C., W. A. Tabisz, C. S. Leu, N. Dai, R. Watson, and F. C. Lee: 1994, 'Development of a DC distributed power system'. In: *Applied Power Electronics Conference and Exposition, 1994. APEC '94. Conference Proceedings 1994., Ninth Annual.* pp. 763–769 vol.2.
- Karlsson, P. and J. Svensson: 2003, 'DC bus voltage control for a distributed power system'. *IEEE Trans. Power Electron.* **18**(6), 1405–1412.
- Karppanen, M., M. Hankaniemi, T. Suntio, and M. Sippola: 2007, 'Dynamical Characterization of Peak-Current-Mode-Controlled Buck Converter With Output-Current Feedforward'. *IEEE Trans. Power Electron.* 22(2), 444–451.
- Kassakian, J. G.: 2000, 'Automotive electrical systems-the power electronics market of the future'. In: *Applied Power Electronics Conference and Exposition*, 2000. APEC 2000. *Fifteenth Annual IEEE*, Vol. 1. pp. 3–9 vol.1.
- Kazimierczuk, M. and L. Starman: 1999, 'Dynamic performance of PWM DC-DC boost converter with input voltage feedforward control'. *Circuits and Systems I: Fundamental Theory and Applications, IEEE Transactions on* **46**(12), 1473–1481.
- Konig, O., C. Hametner, G. Prochart, and S. Jakubek: 2014, 'Battery Emulation for Power-HIL Using Local Model Networks and Robust Impedance Control'. *IEEE Trans. Ind. Electron.* 61(2), 943–955.
- Kwasinski, A. and C. N. Onwuchekwa: 2006, 'DC Micro-grid for Super High Quality Distribution System Configuration and Control of Distributed Generations and Energy Storage Devices'. In: *Power Electronics Specialists Conference*, 2006. PESC '06. 37th IEEE. pp. 1–7.
- Kwasinski, A. and C. N. Onwuchekwa: 2011, 'Dynamic Behavior and Stabilization of DC Microgrids With Instantaneous Constant-Power Loads'. *IEEE Trans. Power Electron.* 26(3), 822–834.

- Lee, J. R., B. H. Cho, S. J. Kim, and F. C. Lee: 1988, 'Modeling and simulation of spacecraft power systems'. *IEEE Transactions on Aerospace and Electronic Systems* 24(3), 295–304.
- Lee, S. S., S. H. Ahn, J. H. Park, J. H. Heo, D. H. Kim, J. K. Park, M. U. Yang, K. J. Kim, and Y. T. Yoon: 2012, 'South Korean power distribution system-based operation, market structure and regulation strategies under distributed generation and smart grid'. In: 2012 IEEE Power and Energy Society General Meeting. pp. 1–7.
- Li, Y., P. Jia, and T. Q. Zheng: 2015, 'Active damping method to reduce the output impedance of the DC-DC converters'. *IET Power Electronics* **8**(1), 88–95.
- Lingjie, K., Y. Shanshui, and W. Li: 2013, 'Analysis on power supply and distribution system for spacecraft based on reliability'. In: *Power Electronics and Applications (EPE)*, 2013 15th European Conference on. pp. 1–9.
- Liu, F., S. Duan, F. Liu, B. Liu, and Y. Kang: 2008, 'A Variable Step Size INC MPPT Method for PV Systems'. *IEEE Trans. Ind. Electron.* **55**(7), 2622–2628.
- Liu, X., A. Forsyth, and A. Cross: 2007a, 'Negative Input-Resistance Compensator for a Constant Power Load'. *IEEE Trans. Ind. Electron.* 54(6), 3188–3196.
- Liu, Y., A. Pratt, P. Kumar, M. Xu, and F. C. Lee: 2007b, '390V Input VRM for High Efficiency Server Power Architecture'. In: *APEC 07 Twenty-Second Annual IEEE Applied Power Electronics Conference and Exposition*. pp. 1619–1624.
- Luo, S.: 2001, 'Front-end converter design and system integration techniques in distributed power systems'. Ph.D. thesis, School of Electrical Engineering and Computer Science, University of Central Florida.
- Luo, S.: 2005, 'A review of distributed power systems part I: dc distributed power system'. *Aerospace and Electronic Systems Magazine, IEEE* **20**(8), 5–16.
- Ma, H., C. Zheng, W. Yu, B. Gu, J. D. Lai, and Q. Feng: 2013, 'Bridgeless electrolytic capacitor-less valley-fill AC/DC converter for offline twin-bus light-emitting diode light-ing application'. *IET Power Electronics* **6**(6), 1132–1141.

- Marx, D., P. Magne, B. Nahid-Mobarakeh, S. Pierfederici, and B. Davat: 2012, 'Large Signal Stability Analysis Tools in DC Power Systems With Constant Power Loads and Variable Power Loads; A Review'. *IEEE Trans. Power Electron.* 27(4), 1773–1787.
- McCoy, T. J. and J. V. Amy: 2009, 'The state-of-the-art of integrated electric power and propulsion systems and technologies on ships'. In: 2009 IEEE Electric Ship Technologies Symposium. pp. 340–344.
- Middlebrook, R. D.: 1979, 'Input filter considerations in design and application of switching regulators'. In: *IEEE IAS*. pp. 366–382.
- Middlebrook, R. D. and S. Cuk: 1976, 'A general unified approach to modelling switchingconverter power stages'. In: Proc. IEEE Power Electronics Specialists Conference, PESC. pp. 18–34.
- Ovalle, A., G. Ramos, S. Bacha, A. Hably, and A. Rumeau: 2015, 'Decentralized Control of Voltage Source Converters in Microgrids Based on the Application of Instantaneous Power Theory'. *IEEE Trans. Ind. Electron.* 62(2), 1152–1162.
- Ozbay, H.: 1999, Introduction to Feedback Control Theory. CRC Press.
- Rahimi, A. and A. Emadi: 2009, 'Active Damping in DC/DC Power Electronic Converters: A Novel Method to Overcome the Problems of Constant Power Loads'. *IEEE Trans. Ind. Electron.* 56(5), 1428–1439.
- Ren, Y., M. Xu, J. Sun, and F. C. Lee: 2005, 'A family of high power density unregulated bus converters'. *IEEE Transactions on Power Electronics* 20(5), 1045–1054.
- S. Abe, M. Hirokawa, T. Z. and T. Ninomiya: 2006, 'Stability Design Consideration for On-Board Distributed Power System Consisting of Full-Regulated Bus Converter and POLs'. In: *Power Electronics Specialists Conference, 2006. PESC '06. 37th IEEE*. pp. 1–5.
- Sable, D., B. Cho, and R. Ridley: 1991, 'Use of leading-edge modulation to transform boost and flyback converters into minimum-phase-zero systems'. *IEEE Trans. Power Electron.* 6(4), 704–711.

- Sarlioglu, B. and C. T. Morris: 2015, 'More Electric Aircraft: Review, Challenges, and Opportunities for Commercial Transport Aircraft'. *IEEE Transactions on Transportation Electrification* 1(1), 54–64.
- Schulz, S., B. H. Cho, and F. C. Lee: 1990, 'Design considerations for a distributed power system'. In: *Power Electronics Specialists Conference*, 1990. PESC '90 Record., 21st Annual IEEE. pp. 611–617.
- Shamsi, P. and B. Fahimi: 2013, 'Dynamic Behavior of Multiport Power Electronic Interface Under Source/Load Disturbances'. *IEEE Trans. Ind. Electron.* **60**(10), 4500–4511.
- Shi, H., F. Zhuo, D. Zhang, Z. Geng, and F. Wang: 2014, 'Modeling, analysis, and measurement of impedance for three-phase AC distributed power system'. In: 2014 IEEE Energy Conversion Congress and Exposition (ECCE). pp. 4635–4639.
- Sun, J.: 2009, 'Small-Signal Methods for AC Distributed Power Systems; A Review'. *IEEE Transactions on Power Electronics* **24**(11), 2545–2554.
- Thandi, G. S., R. Zhang, K. Xing, F. C. Lee, and D. Boroyevich: 1999, 'Modeling, control and stability analysis of a PEBB based DC DPS'. *IEEE Transactions on Power Delivery* 14(2), 497–505.
- Thomas, R. L. and G. J. Hallinan: 1990, 'Design of the Space Station Freedom power system'. *IEEE Aerospace and Electronic Systems Magazine* **5**(1), 19–24.
- Thorsell, L.: 1990, 'Will distributed on-board DC/DC converters become economically beneficial in telecom switching equipment?'. In: *Telecommunications Energy Conference*, 1990. INTELEC '90., 12th International. pp. 63–69.
- Turner, R., S. Walton, and R. Duke: 2013, 'A Case Study on the Application of the Nyquist Stability Criterion as Applied to Interconnected Loads and Sources on Grids'. *IEEE Trans. Ind. Electron.* 60(7), 2740–2749.
- Wang, R., F. Wang, D. Boroyevich, R. Burgos, R. Lai, P. Ning, and K. Rajashekara: 2011,
 'A High Power Density Single-Phase PWM Rectifier With Active Ripple Energy Storage'. *IEEE Trans. Power Electron.* 26(5), 1430–1443.

- Wang, X., R. Yao, and F. Rao: 2005, 'Subsystem-interaction restraint in the two-stage DC distributed power systems with decoupling-controlled-integration structure'. *IEEE Trans. Ind. Electron.* 52(6), 1555–1563.
- Wen, H., W. Xiao, X. Wen, and P. Armstrong: 2012, 'Analysis and Evaluation of DC-Link Capacitors for High-Power-Density Electric Vehicle Drive Systems'. *IEEE Trans. Veh. Technol.* 61(7), 2950–2964.
- Wildrick, C. M.: 1993, 'Stability of Distributed Power Supply System'. Ph.D. thesis, Virginia Polytech. Inst. and State Univ., Blacksburg, VA.
- Wu, M. and D.-C. Lu: 2015, 'A Novel Stabilization Method of LC Input Filter With Constant Power Loads Without Load Performance Compromise in DC Microgrids'. *IEEE Trans. Ind. Electron.* 62(7), 4552–4562.
- Xu, C. D. and K. W. E. Cheng: 2011, 'A survey of distributed power system AC versus DC distributed power system'. In: *Power Electronics Systems and Applications (PESA)*, 2011 4th International Conference on. pp. 1–12.
- Ye, M., P. Xu, B. Yang, and F. C. Lee: 2002, 'Investigation of topology candidates for 48
 V VRM'. In: *Applied Power Electronics Conference and Exposition*, 2002. APEC 2002.
 Seventeenth Annual IEEE, Vol. 2. pp. 699–705 vol.2.
- Zhang, X., X. Ruan, H. Kim, and C. K. Tse: 2013, 'Adaptive Active Capacitor Converter for Improving Stability of Cascaded DC Power Supply System'. *IEEE Trans. Power Electron.* 28(4), 1807–1816.
- Zhang, X., X. Ruan, and C. Tse: 2015, 'Impedance-Based Local Stability Criterion for DC Distributed Power Systems'. *Circuits and Systems I: Regular Papers, IEEE Transactions* on 62(3), 916–925.
- Zhong, Q.-C. and T. Hornik: 2013, *Control of power inverters in renewable energy and smart grid integration*. Wiley-IEEE Press.

Appendix A

Fundamental Concepts of Constant Power Load (CPL) — Negative Resistor Characteristic

First, with regard to any closed-loop converter, as its loop strives to maintain a constant output voltage or current, it also tries to maintain a constant output power P_{out} to a particular load. From this point of view, any closed-loop converter can be treated as a constant power load (CPL).

If an ammeter is then inserted into a series with the CPL, and the input line modulated, one will find that as its input voltage V_{in} grows, its input current i_{in} diminishes, which strives to keep P_{out} constant. Therefore, from this input side, the CPL looks like a negative resistor, which can also be explained as follows.

Where the CPL is assumed to have 100% efficiency, its behaviour can be described via a few lines of algebra:

$$P_{in} = P_{out} = I_{in}V_{in} = I_{out}V_{out} \tag{A.1}$$

where, P_{in} , I_{in} and V_{in} are the input power, input current and input voltage of the CPL respectively. P_{out} , I_{out} and V_{out} are the output power, output current and output voltage of the CPL respectively.

Then, the inverse of the DC transfer ratio μ , can be obtained by rearranging (A.1):

$$\frac{V_{in}}{V_{out}} = \frac{I_{out}}{I_{in}} = \mu \tag{A.2}$$

The static input resistance of the CPL can simply be calculated from

$$R_{in} = \frac{V_{in}}{I_{in}} \tag{A.3}$$

The incremental resistance can be expressed as

$$R_{inc} = \frac{dV_{in}}{dI_{in}} \tag{A.4}$$

This expression represents the way this resistance moves with perturbations. Via simple manipulations, the incremental resistance can be formulated:

$$V_{in} = \frac{P_{in}}{I_{in}} \tag{A.5}$$

$$P_{in} = P_{out} = R_L I_{out}^2 \tag{A.6}$$

where, R_L can represent another switching regulator, a simple resistor, or a linear regulator. Plugging (A.6) into (A.5) and deriving as (A.4) suggests yield

$$\frac{dV_{in}}{dI_{in}} = \frac{d}{dI_{in}} \frac{R_L I_{out}^2}{I_{in}}$$
(A.7)

$$R_{inc} = -R_L \frac{I_{out}^2}{I_{in}^2} = -R_L \mu^2$$
(A.8)

(A.8) can then be reformulated as

$$R_{inc} = -\frac{V_{out}}{I_{out}}\mu^2 = -\frac{V_{out}^2}{P_{out}}\mu^2 = -\frac{V_{in}^2}{P_{out}}$$
(A.9)

Therefore, if a converter operates as a CPL, its input impedance behaves as a negative resistor. Please keep in mind that, the DC/DC converter only acts as a CPL and has a negative input impedance within the cut-off frequency of its closed loop. In other words, if the same converter operated in open-loop or at higher frequency band, it would lose its negative input impedance.